# **Rethinking incumbent firms in sustainability-oriented industry transitions:**

Conceptual advancements and empirical insights from the German meat industry

Dissertation accepted by the Faculty of Sustainability at Leuphana University Lüneburg

for the degree of Doctor of Economics and Social Sciences - Dr. rer. pol. -

by Charlott Hübel born on the 1<sup>st</sup> of June, 1992 in Bochum Submission date: 30 March 2023

First reviewer: Prof. Dr. Dr. h.c. Stefan Schaltegger Second reviewer: Prof. Dr. Jacob Hörisch Third reviewer: Jun.-Prof. Dr. Samanthi Dijkstra-Silva

The articles included in this cumulative dissertation have been or will be published as follows:

[1] Hübel, C. & Schaltegger, S. (2022): Barriers to a sustainability transformation of meat production practices - An industry actor perspective. Sustainable Production and Consumption 29, 128 – 140. https://doi.org/10.1016/j.spc.2021.10.004 *Journal article, peer reviewed, published* 

[2] Hübel, C. (2022): Entrepreneurship-driven organizational transformation for sustainability: A sensemaking lens, Journal of Organizational Change Management, 35(1), 240-256. https://doi.org/10.1108/JOCM-03-2021-0067 *Journal article, peer reviewed, published*

[3] Hübel, C., Weissbrod, I. & Schaltegger, S. (2022): Strategic alliances for corporate sustainability innovation: The 'how' and 'when' of learning processes. Long Range Planning, 102200. https://doi.org/10.1016/j.lrp.2022.102200 *Journal article, peer reviewed, published* 

[4] Dijkstra-Silva, S., Hübel, C. & Schaltegger, S (2023): Company contributions to sustainability transitions of markets: An innovation management perspective. *Status: To be submitted* 

### Acknowledgements

This PhD project has been the most profound learning journey. Back when I wrote the first words of my first paper, little did I know that in the course of the following years, I would not only develop the required knowledge and skills to navigate through academic life, but that I would fundamentally change as a person. I am deeply grateful for all the knowledge gained, all the lessons learnt and all the experiences made. First and foremost, I would like to express my gratitude to my supervisor Prof. Dr. Dr. Stefan Schaltegger without whom I would have not been able to embark on this journey. Thank you for inspiring, supporting and challenging me over these past years. I am also very grateful to Prof. Dr. Jacob Hörisch for always having provided encouraging feedback and for dedicating his time and energy towards reviewing this dissertation as a second reviewer. My deep appreciation also goes to Jun.-Prof. Dr. Samanthi Dijkstra-Silva, with whom I had the pleasure to collaborate on one of the articles of this dissertation and who agreed to become my third reviewer. Further, I want to express my gratitude to the Robert Bosch Foundation for providing me with the resources and support necessary to conduct my research as well as to my case study firm 'BIGMEAT' for allowing me to accompany their reorientation process, to look behind the scenes of a large-scale meat operation and to collect insightful data that today represents the core contribution of this dissertation.

Throughout my journey, I received tremendous support and inspiration from my colleagues of the CSM and the PoST project. I would like to thank (in alphabetical order) Anne, Clara, Cornelia, Gustavo, Josi, Felix, Ferdinand, Frederic, Hanna, Isabell, Ioana, Josi, Julia, Julius R., Julius W., Karoline, Lisa, Lydia, Naemi, Sadhbh, Sebastian, Simon, Stefan H. and Steffi for the fruitful discussions, valuable words of advice and shared moments. I would also like to acknowledge the invaluable contributions of Dr. Ilka Weissbrod, one of my co-authors, who made the paper development process such an enjoyable and straightforward endeavor. My deepest gratitude goes to Maike, who has been my rock from the very beginning and who supported me every step along the way.

Last but not least I would like to thank my family, Sabine, Michael, Tessa, Annabel, Karlotta and Jarne, for enduring the ups and downs of this process and for always having my back. Special thanks goes to my dear friend Issam for being in the right place at the right time, for helping me rediscover my determination and not lose track of what's important in life.

This dissertation is dedicated to my grandfather Klaus Hübel

# **Dissertation content**

# A. Dissertation framework (conceptual)

Integrated incumbent firm behavior in sustainability-oriented industry transitions: Developing a Multi Embeddedness Framework

# **B. Rigidity and lock-in** (empirical)

Hübel, C. & Schaltegger, S. (2022): Barriers to a sustainability transformation of meat production practices - An industry actor perspective. Sustainable Production and Consumption 29, 128-140. https://doi.org/10.1016/j.spc.2021.10.004 *Journal article, peer reviewed, published* 

### C. Intent for sustainability and contribution potential (conceptual)

Dijkstra-Silva, S., Hübel, C. & Schaltegger, S (2023): Company contributions to sustainability transitions of markets: An innovation management perspective. *Status: Submission planned for Research Policy* 

### **D. Processes of exploration and reorientation** (empirical)

Hübel, C., Weissbrod, I. & Schaltegger, S. (2022): Strategic alliances for corporate sustainability innovation: The 'how' and 'when' of learning processes. Long Range Planning, 102200. https://doi.org/10.1016/j.lrp.2022.102200 *Journal article, double blind peer reviewed, published* 

Hübel, C. (2022): Entrepreneurship-driven organizational transformation for sustainability: A sensemaking lens, Journal of Organizational Change Management, 35(1), 240-256. https://doi.org/10.1108/JOCM-03-2021-0067 *Journal article, double blind peer reviewed, published* 

# Annex

- I. Author contributions to dissertation papers and publication status
- II. Presentations related to this dissertation
- III. Declaration

# A. Dissertation framework

Integrated incumbent firm behavior in sustainability-oriented industry transitions: Developing a Multi Embeddedness Framework

# Integrated incumbent firm behavior in sustainability-oriented industry transitions: Developing a Multi Embeddedness Framework

Charlott Hübel Centre for Sustainability Management (CSM) Leuphana Universität Lüneburg Universitätsallee 1 D-21335 Lüneburg, Germany Tel. +49 176 82474738 Charlott.Huebel@leuphana.de

#### Abstract

While transition research recognizes incumbent firms' crucial role in sustainability-oriented industry transitions, descriptions of transition-related firm behavior remain dispersed. Many studies characterize incumbent firms as passive and innovation-averse, whereas others point to reactive behaviors that are triggered by external pressures and result in incremental innovations. Again other studies highlight incumbent firms' proactive exploration of radical sustainability innovations. Recently, transition scholars have called for an integrated view that acknowledges the diversity in incumbent firm behaviors within industries as well as within firms. However, research that explores processes of such integrated behavior, particularly at firm and individual level, is currently lacking. This framework paper addresses the research gap by adopting an integrated view of incumbent firm behavior, analyzing passive, reactive and proactive behaviors at firm and individual level. It does so with the aim of completing the currently disparate picture of incumbent firm behavior in sustainability-oriented industry transitions. The four articles of this dissertation are viewed through the lens of Geels' Triple Embeddedness Framework (TEF), which currently considers passive and reactive behaviors. By adding the consideration of incumbent firm proactivity as well as multi-level interactions between industry, firms and individual managers, the main elements of the TEF are extended. A new Multi Embeddedness Framework (MEF) is developed that details processes and outcomes of integrated incumbent firm behavior including passivity, reactivity and proactivity. A discussion of the developed framework reveals crucial properties of integrated incumbent firm behavior and details a new understanding of incumbent firms in sustainability-oriented industry transition. The insights put forward in this framework paper provide valuable contributions to the transition literature as well as important management implications with regard to the stimulation and promotion of proactive behaviors.

**Keywords:** Incumbent firm, firm behaviour, triple embeddedness framework, industry transition, sustainability

# **Table of Contents**

1. Introduction	1
2. Literature review and theoretical foundation	
2.1 Incumbent firm behavior in industry transitions	3
2.2 Triple Embeddedness Framework (TEF)	5
2.2.1 Industry regime lock-in	6
2.2.2 Strategic response	7
2.2.3 Gradual reorientation	
2.3 Research Objectives	
3. Methodology	
3.1 Research context	10
3.2 Basis of analysis	11
4. Framework extension: From triple to multi embeddedness	
4.1 Multi-level lock-in	14
4.2 Strategic response and deliberate intent	15
4.3 Gradual and accelerated reorientation	17
5. Discussion and contribution	
5.1 Properties of integrated incumbent firm behavior	19
5.2 Understanding of incumbent firms in industry transition	
6. Conclusion	
6.1 Management implications	
<ul><li>6.1 Management implications</li><li>6.2 Limitations and future research</li></ul>	

#### **1. Introduction**

The important role of incumbent firms in advancing sustainability-oriented industry transitions is increasingly recognized in the sustainability transition literature (Turnheim and Sovacool, 2020; Loorbach and Wijsman, 2013). Incumbent firms are referred to as firms-in-industries (Geels, 2014a), which are embedded in the established socio-technical structures of their industry and thereby contrast new entrant firms (Turnheim and Geels, 2019; Berggren et al., 2015). It is argued that incumbent firms, due to their accumulated knowledge and capital, can profoundly influence production and consumption patterns by reorienting "in the directionality of innovations" (Geels, 2014a, p. 263). However, incumbent firms' embeddedness in different market environments often encourages them to adhere to the standards, norms and routines from which they wish to depart (Wells and Nieuwenhuis, 2012; Geels, 2014b). This 'paradox of embedded agency' (Holm, 1995) makes it difficult for incumbent firms to introduce innovations that challenge established system configurations. Some incumbent firms, therefore, remain passive in transitions and are driven out of the market by competing new entrant firms (Geels, 2002). Others, however, find ways to break with established routines, engage in innovation and advance transition efforts - in a more or less radical manner and with more or less success (Bergek et al., 2013; Berggren et al., 2015). This suggests that in one and the same industry, different incumbent behaviors on a spectrum from passive to proactive can be observed (Magnusson and Werner, 2022; Ampe et al., 2021).

Despite recent calls for an integrated view of incumbent firm behavior that includes passivity, reactivity and proactivity (Van Mossel et al., 2018; Magnusson and Werner, 2022; Turnheim and Sovacool, 2020; Steen and Weaver, 2017), empirical transition research has so far neglected to explore associated processes and implications for sustainability transitions. In the past, salient transition research has adopted a passive or reactive view of incumbent firms (e.g. Geels, 2002; Geels and Schot, 2007). At most, the postulated views acknowledge that incumbent firms can reorient towards sustainability, but often only after continuous external pressures and performance decline. For instance, Geels (2014a) combined the passive and reactive view of incumbent firm behavior by developing a so called Triple Embeddedness Framework (TEF). The TEF conceptualizes incumbent firms as embedded actors locked-in to industry structures, who, upon increasing pressures from different environments, can formulate response strategies and gradually reorient towards sustainability. An integrated view of passive, reactive and proactive incumbent firm behavior and respective empirical insights is, however, lacking.

An exploration of integrated incumbent firm behavior in transitions necessitates not only strategic and operational considerations at the level of the firm, but also cognitive and emotional processes at the level of individual managers. Particularly in the context of proactive behaviors, cognitive aspects such as managerial intent, motivation, sensing and evaluating, become important (Kaffka et al., 2020; Ardichvili et al., 2003; Cornelissen and Clarke, 2010). This is due to the fact that the common outside-in perspective is now complemented by an inside-out perspective. Even though transition research acknowledges the importance of cognition (Geels, 2014a; Geels, 2020), processes thereof at the level of individual managers are not explored. Most importantly, it remains unclear how such internal processes can be linked back to industry transitions, including relevant interactions with different firm environments.

Addressing the outlined research gap, this paper adopts an integrated view of incumbent firms and attends to the research question: "*How can an integrated view of incumbent firm behavior advance our understanding of incumbent firms in sustainability-oriented industry transitions?*" Exploring this research question can account for diverging firm behaviors, including proactivity, shine light on hitherto neglected firm- and individual-level processes and reveal their implications for wider industry transition. While the exploration of integrated behavior leads to an increase in complexity, it promises a more complete picture of incumbent firm behavior, including the havior in sustainability transitions. Already combining the passive and reactive view of incumbent firm behavior, Geels' (2014a) Triple Embeddedness Framework (TEF) serves as analytical lens for this framework paper. The four articles of this dissertation are viewed through the main elements of the TEF, resulting in the framework's extension. A new Multi Embeddedness Framework (MEF) is developed that details processes at firm and individual level and reveals insights into integrated incumbent firm behavior including passivity, reactivity and proactivity.

The paper is structured as follows: Section 2 reviews the literature on incumbent firm behavior in industry transitions and presents the main elements of the TEF. Thereupon, Section 3 describes the methodology in terms of research context and basis of analysis. Section 4 details the extension of each of the TEF's elements to develop a new framework, i.e. the MEF. The following Section 5 discusses the MEF with regard to properties of integrated incumbent firm behavior and related understandings of incumbent firms in industry transition. Finally, Section 6 summarizes the main insights and includes recommendations for management and research.

#### 2. Literature review and theoretical foundation

#### 2.1 Incumbent firm behavior in industry transitions

Transition studies propose different views on incumbent firm behavior, ranging from passive and reactive to proactive (van Mossel et al., 2018; see Table 1). This section presents the different views and, in doing so, highlights recent calls for an integrated view combining passive, reactive and proactive behaviors (Magnusson and Werner, 2022).

For a long time, prevalent sustainability transition studies have described incumbent firm behavior as rather *passive* and innovation-adverse, starkly contrasting the innovation-oriented behavior of smaller start-ups in market niches (e.g. Geels, 2002; Geels and Schot, 2007; Kemp et al., 1998; Schot and Geels, 2008; Raven et al., 2016). According to these studies, market niches serve as protected spaces for the development of radical innovations, from which they are disseminated into the mainstream (Kemp et al., 1998; Geels, 2002). Incumbent firms as embedded actors in rigid industry regime structures are bounded to the status quo and find it difficult to engage in disruptive activities. The literature around the passive view on incumbent firm behavior focuses on prevailing lock-ins, resistance to reorientation and resulting competitive disadvantage (Unruh, 2000; Wells and Nieuwenhuis, 2012; Lee and Hess, 2019). Other transition studies acknowledge incumbent firms' capacity for adaptation, postulating the possibility of *reactive* behavior. According to this view, incumbent firms have the ability to partly escape lock-in and reorient towards sustainability once they face increasing external pressures and performance problems (Geels, 2014a; Penna and Geels, 2015; Karltorp and Sandén, 2012). By gradually reorienting their strategies, they remain competitive and thus contribute to industry transition (Geels, 2014a). Strategies can include institutional strategies towards the external environment of the firm (Smink et al., 2015a) as well as innovation strategies through which incumbents develop innovations themselves (Kishna et al., 2017; Onufrey and Bergek, 2021).

More than 20 years ago, Chandy and Tellis (2000) challenged the perception of incumbents as non-innovative laggards doomed to be driven out of the market by newcomers. In the past decade, the view that incumbent firm can also shape industry transitions *proactively* has received growing attention in the transition literature (Bergek et al., 2013; Berggren et al., 2015; Werner et al., 2022). Here, incumbent firms perceive opportunities for radical innovations early on and use them to their advantage to become first-movers in their industry (Steen and Weaver, 2017). Research finds that incumbent firms' innovative capacity manifests in their accumulated

capital, competences and experience, all of which allow them to develop new technologies, integrate them into existing ones and thereby potentially outperform experience- and resource-weak newcomers (Bergek et al., 2013). At the same time, incumbent firms can use their position of power to influence other industry actors, diffuse sustainable business models and thereby support industry-scale transitions (Bidmon and Knab, 2018; Rovanto and Bask, 2021). This proactive view is closely related to research streams such as sustainable entrepreneurship which has acknowledged the innovative capacity of incumbent firms (Schaltegger et al., 2016; Schaltegger and Wagner, 2011).

Incumbent	Explanation	Literature	
firm behavior	•		
Passive	Their embeddedness in dominant industry structures makes incumbent firms resist reorientation towards sustainability. Radical innovations are mainly developed and disseminated by new entrants in protected market niches.	Geels, 2002; Geels and Schot, 2007; Geels, 2010, Geels 2014b; Smith and Raven, 2012; Wells and Nieuwenhuis, 2012; Lee and Hess, 2019; Kemp et al., 1998; Unruh, 2000; Raven et al., 2016;	Geels,
Reactive	In response to increasing external pressures, incumbent firms strategically reorient towards sustainability and gradually contribute to the	Penna and Geels, 2015; Karltorp and Sandén, 2012; Kishna et al., 2017; Onufrey and Bergek, 2021; Smink et al.,	
	destabilization of the prevailing industry regime.	2015a	Stalmokaite
Proactive	Incumbent firms have the capacity to perceive market opportunities for radical sustainability innovations, integrate them with existing capabilities and substantially drive sustainability transition.	Chandy and Tellis, 2000; Bergek et al., 2013; Berggren et al., 2015; Werner et al., 2022	2020
Integrated	Incumbent firms display internally diverse behaviors ranging from passive and reactive to proactive behavior.	Van Mossel et al., 2018; Magnuss Werner, 2022; Turnheim & Sovae Steen & Weaver, 2017; Ampe et a	son and cool, 2020; al., 2021

Table 1: Literature review of different incumbent firm behaviors

More recently, transition scholars have begun to criticize the conceptual separation between passive, reactive and proactive incumbent firm behaviors. Instead, they call for a more "nuanced" (Magnusson and Werner, 2022), "heterogeneous" (Steen and Weaver, 2017; Eggers and Park, 2018) and "pluralized" (Turnheim and Sovacool, 2020; Ampe et al., 2021) understanding. Such an *integrated* view of incumbent behavior will make it "necessary to go beyond the description of the firm as a coherent unit" and instead, view incumbents as multifaceted actors capable of simultaneous, potentially contrasting behaviors (Magnusson and

Werner, 2022, p. 10). A recent empirical study by Stalmokaite and Hassler (2020) has found evidence of a mixed portfolio of integrated reactive and proactive innovation strategies in shipping firms facing challenges of decarbonization. While the study provides insights into pressure- and opportunity-driven reorientation processes over time, it did not explain fully integrated firm behavior, including firm passivity. Studies have also fallen short in linking co-existing behaviors with external environments, including their impact on industry transition. The Triple Embeddedness Framework developed by Geels (2014a) lays the conceptual groundwork for combining passive and reactive incumbent firm behavior, accounting for both resistance and reorientation. While it also does not describe fully integrated behavior, the framework begins to detail mutual interactions between incumbent firms and different external environments that mediate firm behavior. The following section presents the TEF's main elements.

#### 2.2 Triple Embeddedness Framework (TEF)

According to the TEF, incumbent firm behavior is influenced by two external environments, the socio-political and economic environment, and mediated by a so called industry regime (Geels, 2014a). The industry regime encompasses industry-specific norms, standards, knowledge and mindsets and forms incumbent firms' perceptions and activities towards industry transitions. This triple embeddedness of incumbent firms can account for firm rigidity, but can also generate the necessary conditions for firm reorientation. Incumbent firm behavior is described along three main elements: While (1) incumbent firms experience *lock-in* from the prevalent industry regime and therefore rather resist fundamental change, they still have the possibility to (2) *strategically respond* to increasing environmental pressures and (3) *gradually reorient* towards sustainability. While accounting for passivity and reactivity, the TEF, to date, does not provide insights into proactive incumbent firm behavior, which would require an indepth exploration of processes at firm and individual level. The following sections briefly summarize the three main elements of the TEF, outline the framework's shortcomings with regard to firm- and individual-level processes and identify relevant research streams.



Figure 1: Triple Embeddedness Framework (TEF) (adapted from Geels, 2014a)

#### 2.2.1 Industry regime lock-in

The TEF postulates that, due to isomorphic industry dynamics, many regime elements have become locked-in to certain patterns (Element 1 in Figure 1). The prevailing lock-in is due to different reinforcing mechanisms related to industry-specific capabilities, culture, norms and regulations. These reinforcements encourage incumbent firms to continue business as usual and thus create barriers for them to reorient towards sustainability. As a result, incumbent firm reorientation tends to be incremental and related innovations largely stay on existing trajectories.

Geels bases his ideas of rigidity, isomorphism and lock-in on social science theories such as neo-institutionalism (DiMaggio and Powell, 1983), evolutionary economics (Dosi, 2000) and economic sociology (Polanyi, 1944; Krippner, 2001). As the author aimed at explaining the complexity of the co-evolution of industries, including explanations for industry rigidity and incumbent firm passivity, he takes an industry-level perspective. This is why the TEF focuses on the explanatory power of the industry regime, which accommodates for the targeted broad understanding of institutional influences. Even though the framework does mention

organizational theory, lock-ins are exclusively explained as regime-level phenomena. Geels himself notes that, to date, the TEF "does not address what goes on inside firms, which would introduce another layer of complexity" (2014a, p. 275). More importantly, it does not address what goes on inside individuals. This is despite the fact that previous research has found lock-ins related to mindsets, mission and identity as particularly strong and hardest to change (Turnheim and Geels, 2019). Still, firm-level theories, including organizational path dependence (Sydow et al., 2009), or individual-level theories, including behavioral theories (Kollmuss and Agyeman, 2002; Ajzen, 1991) are not considered by the TEF. An application thereof, however, would provide more nuanced understandings of restrained incumbent behaviors in sustainability transitions and may help to identify levers for cognition-driven reorientation and, eventually, contributions for sustainability-oriented industry transition.

#### 2.2.2 Strategic response

The TEF suggests that incumbent firms can, to a certain degree, overcome lock-in once they face increasing external pressures such as new market competition or regulatory changes. When pressures become so high that they negatively affect firm performance, they trigger different strategic responses by incumbent firms towards their environments (Element 2 in Figure 1). The pursued strategies help firms position and defend themselves against pressures. For instance, to respond to market pressures, firms can select externally-oriented innovation strategies to develop improved products. They can also pursue internal strategies of rethinking routines, capabilities and belief systems to adapt to the changing competitive landscape. The main aim of both complementary strategy approaches is to re-establish a fit between the firm and its environments.

As Geels' (2014a) aim was to conceptualize industry reorientation in relation to grand challenges, focusing on the intersection of passive and reactive incumbent firm behavior was his deliberate choice. He did, therefore, exclusively consider strategic behavior as a response to external pressures rather than as the result of opportunity perception. In light of recent calls for more heterogeneous understandings of incumbent firms (Magnusson and Werner, 2022; Steen and Weaver, 2017; Turnheim and Sovacool, 2020), however, the additional consideration of proactive incumbent firm behavior might be a valuable research avenue. In this regard, recent transition research that considers crucial firm-market interactions concludes that "incumbent firms can respond proactively with the ambition to shape markets" (Werner et al., 2022, p. 22). The implications of this view would be that the main trigger for reorientation is not external

pressure, but instead a firm's deliberate intent to contribute to industry transition. In this regard, the sustainable entrepreneurship and sustainability innovation literature (Vuorio et al., 2018; Schaltegger and Wagner, 2011; Klewitz and Hansen, 2014) has highlighted the importance of sustainability motivations, goals and intentions of individuals for proactive innovation endeavors. Such individual-level aspects have not been considered by the TEF, so far. Geels (2014a, p. 275) himself acknowledges the potential to delve deeper into cognitive approaches "with an individualistic orientation". Another implication of a proactive view would be that the overall aim is no longer achieving a fit with external environments, but rather to actively shape and transform them. To date, the impact of different approaches for industry transitions are beyond the scope of the TEF. Once you consider proactive behavior, however, the choice of measures and their potential effectiveness in terms of industry transition becomes an essential field of analysis.

#### 2.2.3 Gradual reorientation

The TEF merges the two presented elements of lock-in and strategic response in a temporal manner by determining a consecutive process of incumbent reorientation (Element 3 in Figure 1). Initially, incumbent firms tend to resist change due to prevailing regime lock-ins. With increasing pressures and performance problems, they respond by first exploiting existing resources for incremental innovations and thereupon exploring new knowledge and technology for more radical innovations. Here, collaborations with start-ups or peripheral firms are mentioned as valuable opportunities to explore new technologies. Eventually, culminating pressures encourage firms to fundamentally rethink their purpose, beliefs and identity. This gradual reorientation process is accompanied and facilitated by cognitive processes of sensemaking and learning: External signals are identified and interpreted (sensemaking) and environmental feedback and experiences are evaluated (learning) to inform strategic choice and actions.

While the TEF argues for incumbent firm's capacity to reorient, questions remain regarding firm- and individual-level processes, i.e. regarding *how* reorientation plays out in detail and *by whom* exactly the change is brought about. While learning and sensemaking are assessed as crucial facilitators of incumbent reorientation, processes thereof, particularly at the level of individual managers, are not detailed. To give an example, Geels (2014a) mentions collaborations between incumbent firms and start-ups as facilitators of reorientation but does not delve into associated processes of learning and knowledge internalization. This is in line

with other transition studies describing the importance of interorganizational collaboration without detailing related processes at individual level (e.g. Engwall et al., 2021). In the context of the TEF, learning is triggered by environmental developments including unfamiliar events or cultural pressures. Collaborative learning environments between individuals are not further explored. This is despite the fact that seminal alliance research determines learning in alliances to be crucial for advancing corporate innovation and transformation (Weiblen and Chesbrough, 2015; Inkpen and Currall, 2004). Insights into such firm- and individual-level learning processes could form a valuable addition to the TEF, as they may detail the driving forces of reorientation. Further, the TEF relates cognitive interpretative processes, i.e. sensemaking, to the interpretation of environmental signals. However, individual processes of interpretation are not further detailed, particularly with regard to detecting and reflecting about not only market threats, but also market opportunities. This is despite the fact that research has emphasized the role of cognitive models and processes for coping with the uncertainties and ambiguities that come with the proactive initiation of innovative activities (Hill and Levenhagen, 1995). While Geels (2014a) did not delve into opportunity contexts, he did suggest that the perception and interpretation of opportunities might accelerate the process of reorientation. In this regard, research has shown how sensemaking can drive proactive innovation behavior (Kaffka et al., 2020; Ardichvili et al., 2003; Cornelissen and Clarke, 2010). Delving deeper into the literature on organizational sensemaking (Weick, 1979; Maitlis, 2005) could reveal how managers make sense of organizational developments and arrive at decisions for firm reorientation over time.

TEF Element	Addressed aspect	Shortcomings	Relevant research stream(s) to address shortcomings
Industry regime lock- in	Incumbent firms are locked-in to prevelant industry regime elements, which renders reorientation difficult	Firm- and individual-level lock-in and its implication for reorientation efforts	Behavior theories (Kollmuss and Agyeman, 2002; Ajzen, 1991) Organizational path dependence (Sydow et al. 2009)
Strategic response	Incumbent firms respond strategically to increasing external pressures with the aim of adaptation and environmental fit	Managerial intent for sustainability and proactive choice of innovation measures with the aim of contributing to industry transition	Sustainable entrepreneuship and sustainability innovation (Vuorio et al., 2018; Schaltegger and Wagner, 2011; Klewitz and Hansen, 2014)
Gradual reorientation	Incumbent firms gradually reorient towards sustainability along a process that is accompanied by externally oriented learning and sensemaking	The nature and impact of cognitive processes related to learning in alliances and firm-internal sensemaking with regard to both threats and opportunities	Alliance learning (Weiblen and Chesbrough, 2015; Inkpen and Currall, 2004) Organizational sensemaking (Weick, 1979; Maitlis, 2005)

Table 2: Main elements of the TEF, their shortcomings and relevant research streams

#### 2.3 Research Objectives

This framework paper addresses the above outlined shortcomings of the TEF by advancing it respectively. It does so by adopting an integrated view of incumbent firm behavior, combining passive, reactive and proactive behavior. While the TEF has addressed passive and reactive behavior through its focus on lock-in and strategic response, this paper adds the exploration of proactive incumbent firm contributions to sustainability-oriented industry transitions, thereby accounting for fully integrated incumbent firm behavior. Adopting this view necessitates both firm-level and individual-level analyses, thus adding, in Geels' (2014a, p. 275) words, the missing "layer of complexity" to the TEF. Analysis is based on extensive empirical data, which provides new insights into firm- and individual-level reorientation processes. Concretely, this research looks at what is going on inside and in between firms and individual managers, including cognitive processes of learning and sensemaking. It does so while still considering incumbent firms' embeddedness in and interaction with other environments. The overall research objective and sub-aspects can be summarized as follows:

This research extends the TEF by firm- and individual-level insights with the aim of arriving at an integrated view of incumbent firm behavior encompassing passive, reactive and proactive behaviors. The framework extension aims to

- determine the significance of firm- and individual-level lock-in for firm reorientation
- consider proactive behavior through managers' deliberate intent for industry transition
- detail the role of learning and sensemaking processes for firm reorientation over time

# 3. Methodology

#### 3.1 Research context

The empirical data that forms the basis for this dissertation has been obtained from 23 incumbent operations in the German meat industry. On the one hand, data was collected from 22 actors along the meat production chain, including farmers, butchers, traders, producers and retailers. On the other hand, data was collected in the context of a longitudinal in-depth analysis of one particular incumbent firm, a large European meat producer with a turnover of  $\notin$ 2.8bn in the fiscal year 2020/21. The firm launched a fundamental reorientation process by engaging in extensive exploration activities in the field of sustainable protein solutions. To ensure anonymity, the following elaborations will refer to the case firm as BIGMEAT.

The rationale for selecting the meat industry as research context results from several reasons. First, food systems are linked to major global sustainability challenges such as resource scarcity, ecosystem degradation, and climate change (Crippa et al., 2021; Springmann et al., 2018). In this regard, (industrial) meat production has been assessed to cause major negative impacts on the environment (Steinfeld et al., 2006; Fiala, 2008), animal welfare (Gregory and Grandin, 2007) and public health (Walker et al., 2005). These producer-induced impacts render analyses on transition processes in the food and meat industry important. Second, individual-level considerations, including behavioral and emotional factors, play a big part in food or meat consumption and production (Stoll-Kleemann and Schmidt, 2017; Hoek et al., 2021). For instance, in Germany, the production and consumption of meat is deeply embedded in the national culture (Mellinger, 2000), which might make it difficult for individuals to engage in transitions towards more sustainable behavioral patterns. The food context thus offers valuable opportunities to delve deeper into individual-level processes influencing transition-related behavior.

Lastly, besides the more traditional topics of mobility and energy transitions, sustainabilityoriented food system transitions have become a growing topic of interest in the transition literature (e.g. Cohen and Ilieva, 2015; El Bilali, 2019). In this regard, a recent literature review concludes that there is a research gap in analyzing the role of firms in transition processes in the agro-food system (El Bilali, 2019). Indeed, much research dealing with sustainability transitions in general (e.g. Vittersø and Tangeland, 2015) and in the meat industry in particular (e.g. Stoll-Kleemann and Schmidt, 2017) has focused on the perspective of consumers. This is despite the fact that transitions in the agro-food sector have been found to be dependent on the engagement of regime actors such as incumbent firms (Vermunt et al., 2020). There is thus high potential to address the producer perspective, analyzing incumbent meat firms' barriers and success factors for reorienting towards sustainable meat production systems.

#### **3.2 Basis of analysis**

The basis of analysis encompasses four scientific articles that all address firm- and individuallevel processes of incumbent firm reorientation. Each article corresponds with a different element of the TEF and provides relevant insights that address the identified shortcomings. Table 3 provides an overview of the different articles and the elements of the TEF that they cover.

		Articles				
		No.	Author(s)	Short title	Focus	Theoretical lens / Literature
	Regime lock-in	1	Hübel and Schaltegger, 2022	Industry Actor Barriers	External and internal reorientation barriers as perceived by individual industry actors	Model of pro-environmental behavior (Kollmuss and Agyeman, 2002)
t of the TEF	Strategic response	2	Dijkstra- Silva et al., 2023	Innovation Management	Deliberate selection of management measures for sustainability innovations and potential contributions to industry transition	Sustainability transition (Loorbach and Wijsman, 2013), sustainability innovation (Klewitz and Hansen, 2014) & innovation management (Van de Ven, 1995)
Element	Gradual reorient- tation	3	Hübel et al., 2022	Alliance learning	Processes of alliance learning and outcomes for sustainability innovation and industry transition	Alliance learning theory (Doz, 1996; Inkpen, 1998; Inkpen and Currall, 2004)
		4	Hübel, 2022	Managerial Sensemaking	Top and middle manager sensemaking and its influence on a firm's exploration and transformation	Organizational sensemaking (Weick, 1979; Gioia and Chittipeddi, 1991)

Table 3: Scientific articles considered for this dissertation

The first article contains the results of a qualitative research study conducted by Hübel and Schaltegger (2022) and is titled "*Barriers to a sustainability transformation of meat production practices - An industry actor perspective*". The analysis is based on 22 semi-structured interviews with incumbent actors, i.e. firms and individuals, along the meat production chain as well as one interview with the chief editor of a leading meat industry journal. Applying Kollmus and Agyeman's (2002) model of pro-environmental behavior, the interview analysis reveals multi-level barriers for reorienting meat production practices towards sustainability. The article provides the missing individual- and firm-level analysis of reorientation barriers.

The second article is titled "*Company contributions to the sustainability transition of markets: An innovation management perspective*" and contains conceptual work conducted by Dijkstra-Silva et al. (2023). The article links sustainability transition literature with insights from sustainability innovation and innovation management literature to propose a framework for managing company contributions to sustainability transitions of markets. The article addresses managerial choice and deliberate intent for industry transition as well as potential resulting contribution pathways, thus covering the second element of the TEF and adding intention and visioning.

The two remaining articles are both based on the in-depth case study of BIGMEAT. The empirical database includes a total of 40 interviews, observations at 11 events and meetings as well as 71 documents. The two articles represent the heart of this dissertation and reveal detailed insights into individual- and firm-level processes of incumbent reorientation at the intersection of passive, reactive and proactive behavior. Article No. 3 by Hübel et al. (2022) is titled *"Strategic alliances for corporate sustainability innovation: The 'how' and 'when' of learning processes"*. Applying seminal alliance learning theory, the article examines, how learning about and from startup firms can help advance and accelerate an incumbent firm's reorientation for *sustainability: a sensemaking lens"* analyzes sensemaking processes of BIGMEAT's top and middle managers and shows how sensemaking mediated firm-internal transformation and innovation activities.

#### 4. Framework extension: From triple to multi embeddedness

Based on the presented empirical and conceptual contributions, this chapter extends the TEF with regard to its three main elements of industry regime lock-in, strategic response and gradual reorientation. New firm- and individual-level insights address the identified shortcoming of the TEF and result in the development of a Multi Embeddedness Framework (MEF) which complements the insight provided by the original framework. The MEF offers an integrated view of incumbent firm behavior in industry transitions, meaning that it combines passive and reactive with proactive incumbent firm behavior and details processes thereof. The framework's elements are highlighted in Figure 2 and explained in the following sections.



Figure 2: Multi Embeddedness Framework (MEF)

#### 4.1 Multi-level lock-in

According to the TEF, lock-ins arising from the industry regime account for the tendency of incumbent firms to (initially) resist reorienting towards sustainability. The findings of this dissertation, however, point to a multi-level lock-in of incumbent actors. Besides the industry regime, incumbent firms seem to be additionally influenced by an 'organizational regime' at firm level as well as a 'personal regime' at individual level (Element 1 in Figure 2). The MEF accounts for the complexity of transition efforts by describing reorientation barriers resulting from the three regimes as well as their interacting elements. In the following, multi-level regime influences are described on the basis of the dissertation's empirical insights.

The analysis of meat industry actor barriers by Hübel and Schaltegger (2022) reveals that reorientation barriers not only arise from lock-ins at industry level through industry-specific infrastructure, standards and mindsets, but also at firm level through firm-specific resource allocation and operational procedures and at the level of individual managers through personal knowledge, emotional and cognitive responses, values and attitudes. Firm-level lock-in as perceived by meat industry actors, for example, can derive from long-term investments into agricultural property like pig fattening stables, which create sunk costs and render reorientation endeavors neither desirable nor feasible. The organizational regime, in this regard, encompasses

resource factors which mediate operations at firm-level. The personal regime, on the other hand, influences individual behavior. According to the results of Hübel and Schaltegger's (2022) meat industry study, personal regime lock-in, for instance, can derive from emotional commitment to long-standing family traditions in animal husbandry or butchery. Here, the authors identify particularly strong individual-level lock-ins for operation owners. Hübel's (2022) sensemaking analysis later shows that emotional commitment to conventional meat operations can become an equally impeding factor for a firm's middle managers. In the case of BIGMEAT, managers' commitment rendered it difficult for them to consider the exploration of alternative protein innovations and accept the new organizational identity as protein provider.

Hübel and Schaltegger's (2022) analysis further proposes that the difficulty of reorientation for incumbent firms is increased by the constant interaction of multi-level factors contributing to lock-in (connecting arrows between regimes in Figure 2). For instance, emotional commitment to long-standing operations (emotional lock-in) can cause a status quo bias (cognitive lock-in), which, in turn, encourages incumbent actors to defend and reinforce industry-oriented standards (industry lock-in). The differentiation of and interaction between different regime levels takes into consideration the different roles of incumbent actors as individuals, managers, organizational actors and industry members and acknowledges the complexity of reorientation efforts from an actor perspective.

#### 4.2 Strategic response and deliberate intent

The TEF accounts for reactive incumbent firm behavior by determining external pressures as triggers for escaping lock-in and gradually reorienting towards sustainability. In contrast, the conceptual and empirical contributions of this dissertation suggest the possibility of proactive incumbent firm behavior that is based on resources, values and mindsets at the level of the firm and managers. In this regard, the MEF includes the importance of individual managers' deliberate intent for sustainability (Element 2 in Figure 2). As the empirical insights show, however, deliberate intent does not replace lock-in and/or strategic responses, but instead co-exists with both.

This dissertation's analyses suggest that reorientation can be facilitated by specific firm- and individual-level factors rooted in the proactive view of incumbent firm behavior. In this regard, Dijkstra-Silva et al. (2023) propose that the management of carefully selected innovations and resources at firm level as well as managers' deliberate intent for sustainability at individual level are crucial antecedents for innovation-driven reorientation and transition contribution.

BIGMEAT, for instance, selected those innovations that could be integrated into their existing production and sales structure (Hübel et al., 2022). For maximum industry impact, they combined their market experience and distribution network with the innovativeness of novel protein solutions of international start-up firms. Exploration decision were fundamentally facilitated by the availability of the required financial and material resource base. In this regard, Dijkstra-Silva et al. (2023) suggest that the adequacy and contribution potential of innovation management measures differs with each company, depending on its size, resource base and market position. BIGMEAT, due to its large resource base and dominant position in the European market, had the potential to fundamentally influence various industry environments, including suppliers, politicians, consumers and competitors. By forming distribution partnerships which helped making sustainable protein products available to the mainstream market, BIGMEAT used its potential as major contributor to transition.

With regard to the individual level, Dijkstra-Silva et al. (2023) suggest that if managers display a deliberate intent to contribute to industry transition, they will select those management measures which promise the furthest-reaching impact across industry environments. In this regard, the authors propose so called "contribution pathways" of individual management measures: A measure targeted at improving sustainability in one industry environment can trigger changes in others, including the economic and socio-political environment, and ultimately, the industry regime. In the case of BIGMEAT, managers' intent was influenced by the firm's long term orientation as a family firm and respective openness towards sustainability topics (Hübel et al., 2022). This orientation towards long-term survival and growth made BIGMEAT top managers not only make sense of market pressures to reorient towards sustainability in general (reactive view), but also made them eager to detect and use opportunities to drive reorientation towards promising new markets (proactive view) (Hübel, 2022). Deliberate managerial choice was thus based on the firm's long-term orientation and facilitated by continuous managerial sensemaking. Hübel's (2022) sensemaking analysis shows how two particularly passionate members of the top management team had detected opportunities in the alternative protein field years before meat alternatives became a mainstream topic. Sensing the chance for additional business growth, they initiated first investments and distribution partnerships. While the initial launch of exploration activities was done without the explicit intent to contribute to sustainability transition, learning processes in the context of alliances impacted individual top managers' perceptions and made them change their intentions over time (Hübel et al., 2022). However, as the analyses also showed, deliberate intent and

proactivity by few managers of the top management team did not spill over to middle managers who were largely engaged in the conventional meat business. Here, strategic response behaviors aimed at (re-)establishing environmental fit dominated, resulting in a firm-level co-existence of strategic response and deliberate intent.

#### 4.3 Gradual and accelerated reorientation

The TEF describes reorientation as a gradual process, over the course of which incumbent firms move from denial to fundamental recreation. In contrast to the suggestions of the TEF, the analyses of BIGMEAT's alliance learning and sensemaking processes reveal that reorientation processes at firm- and individual level have to be neither consecutive nor gradual. Instead, they suggest that reorientation towards radical innovation can be vastly accelerated, implying the simultaneous existence of different stages of reorientation within a firm (Element 3 in Figure 2). The MEF therefore incorporates the possibility of co-existing gradual and accelerated processes of reorientation. The split arrow in the inner circle in Figure 2 visualizes this co-existence. In the following, potential acceleration as well as co-existence of different reorientation as well as co-existence of different reorientation as well as co-existence.

As the analyses of Hübel et al. (2022) and Hübel (2022) show, BIGMEAT's exploration activities were accelerated by (1) alliance learning and (2) alternating sensemaking (fine-lined arrows going through the inner circle in Figure 2). Regarding the first aspect, continuous alliance learning made BIGMEAT progressively reorient towards different protein markets, from meat alternatives to egg and cheese alternatives. The analysis of Hübel et al. (2022) reveals that learning at BIGMEAT was vastly increased and accelerated through the firm's creation of a broad portfolio of alliances with different start-ups for alternative protein solutions. Here, the paper distinguishes the important roles of learning *about* and learning *from* start-ups. Learnings about the characteristics, values and goals of the start-up partners lead to the formation of further alliances with start-ups for increasingly radical sustainability-oriented solutions (e.g. 3D-printed meat). Learning from partner firms prompted BIGMEAT managers to adopt new perspectives and integrate global sustainability concerns into decision-making. Successful reorientation towards sustainability thus not only depends on the internalization of technical knowledge, but especially on personal value-based learnings. For instance, alliance partners taught BIGMEAT top managers the importance of global food security, which made them more eager to explore further alternative protein solutions. Interestingly, the analysis also found that negative learning experiences in alliances in terms of process delays and broken promises were

crucial for the acceleration of BIGMEAT's in-house innovation activities in terms of plantbased protein solutions. The case of BIGMEAT not only showed the link between individuallevel learning and firm-level innovation, but also the link between learning, innovation and industry transition. For instance, the learning alliances contributed to the transition of the economic environment through the increased availability of alternative protein solutions in the mainstream market as well as the transition of industry regime elements through the increased acceptance of and knowledge about vegan alternatives.

Regarding the second factor, Hübel (2022) suggests that sensemaking activities by top and middle managers can help firms navigate and accelerate internal transformation processes. The analysis of BIGMEAT's sensemaking processes finds that this navigation and acceleration is achieved through a continuous alternation between sensemaking for innovation activities (interpretation of environmental or firm-internal changes) and sensemaking of innovation activities (interpretation of exploration impacts). If pursued at both top and middle management level, sensemaking processes can profoundly advance and accelerate a firm's reorientation towards sustainability, including changes to structure, culture and identity. BIGMEAT only unleashed the full potential of sensemaking rather late in its exploration process by including middle managers into sensemaking and creating an open space for ideas. While the inclusion of middle managers helped overcoming hierarchical tensions, it also lay bare internal diversity in terms of different managerial intentions and preferences. Eventually, those processes lead to a compromised solution, as BIGMEAT created an overarching identity of "protein provider" which allowed different meanings to co-exist. Instead of the replacement of established meanings, meanings were accumulated over time, which reflected the parallel pursuit of reactive and proactive strategies. While BIGMEAT's top managers reacted to environmental pressures to increase eco-efficiency and animal welfare in the conventional meat business, they proactively engaged in exploration for radical alternative protein solutions. The acceleration of exploration through learning and sensemaking existed independently from gradual reorientation in the conventional business.

#### 5. Discussion and contribution

While the TEF conceptualizes passive and reactive incumbent firm behavior, the newly developed MEF provides processual firm- and individual-level insights into *integrated* incumbent firm behavior that encompasses passivity, reactivity and proactivity. This section discusses the MEF with regard to specific properties of integrated incumbent firm behavior and their implications for a potentially new understanding of incumbent firms in industry transition. In each subsection, contributions to the sustainability transition literature are highlighted.

#### 5.1 Properties of integrated incumbent firm behavior

The elements of the MEF point to several properties of integrated incumbent firm behavior. These properties can be subsumed under the three terms (1) co-existence (2) accumulation and (3) interaction, each of which is discussed in the following.

#### (1) Co-existence

Integrated firm behavior is most fundamentally based on the co-existence of different processes and meanings. While the TEF suggests one path for incumbent firm reorientation (lock-in -> strategic response -> gradual orientation), the MEF allows for several paths to overlap. This overlap manifests in the co-existence of multi-level regime lock-in and firm reorientation, the co-existence of strategic response and deliberate intent as well as the co-existence of gradual reorientation and accelerated transformation. Previous transition research has already pointed to heterogeneous incumbent firm behavior, ranging from innovation-adverse to innovationoriented and from competitive to collaborative behavior (Magnusson and Werner, 2022; Steen and Weaver, 2017). However, respective studies merely point to behavioral diversity *among* different firms of the same industry rather than behavioral diversity within one and the same firm. Stalmokaite and Hassler's (2020) recent study approached internal diversity by revealing the co-existence of reactive and proactive strategies. This finding may relate to innovation management research arguing for the simultaneous pursuit of exploitation and exploration activities (O'Reilly and Tushman, 2011). The MEF, however, argues for co-existent passivity, reactivity and proactivity and thus completes the picture of fully integrated firm behavior. It therefore contributes to sustainability transition research that currently lacks such a comprehensive behavioral view. The MEF further adds the perspective of the embeddedness of incumbent firms in different regimes and industry environments. Co-existing developments always result from or relate to developments in other environments.

#### (2) Accumulation

Integrated incumbent firm behavior results in an accumulation of meaning, not in the replacement thereof. According to the TEF, increasing pressures allow incumbent firms to escape lock-in and pursue strategies that lead to firm reorientation. The ultimate outcome of this process is strategic recreation, meaning comprehensive changes in strategy, technology, mission, beliefs and identity. Despite not formulating it explicitly, the TEF suggests that previous firm activities and meanings cease to exist. The integrated view of the MEF argues for the integration of the old into the new, which manifests differently among hierarchical management levels. Alternating sensemaking of old and new meanings help create new meanings that incorporate established ones. This not only includes the integration of exploitation and exploration activities, but also the integration of different exploration paths. In the case of BIGMEAT, alternating sensemaking of innovation activities helped the firm to create a diverse alliance portfolio with increasingly radical value propositions. The overall outcome of these processes, however, was not the continued replacement of previous activities, but an accumulation that created an overarching, integral identity. This processual perspective of the integrated view of incumbent firm behavior represents a new contribution to transition research. In the transition literature, the aspect of accumulation has been addressed with regard to industry developments, where new knowledge is progressively combined with established technologies (Bergek et al., 2013). However, firm-level insights have been lacking, so far. At firm level, diversity in strategy pursuit has been discussed (Stalmokaite and Hassler, 2020), but without considering implications of accumulation.

#### (3) Interaction

The interaction between industry environments, firms of the same industry and between individuals inside a firm is cause and outcome of integrated firm behavior. According to the TEF, the relationship between the firm and its environments is bi-directional in the sense that incumbent firms receive information form the environment and formulate strategies to respond to that information. Complementing these insights, the MEF accounts for more complex relationships by revealing a multi-level interaction between different regimes and environments. On the one hand, regime interactions increase complex lock-in accounting for passive behavior. On the other hand, environmental interactions can cause positive chain reactions of taken sustainability measures, accounting for reactive and proactive behaviors.

In addition, the MEF not only shows how incumbent firms react to information from the environment (reactive behavior), but also how they deliberately seek new knowledge through collaborations with other industry members (proactive behavior). The TEF along with many conceptualizations of incumbent firms in sustainability transitions do not consider collaborative relationships between incumbent and new entrant firms (e.g. Geels and Schot, 2007; Kemp et al., 1998; Schot and Geels, 2008; Raven et al., 2016). These frameworks regard incumbent firms as separate from innovation spaces, rather incapable of collaborating with niche firms due to prevailing institutional logics (Smink et al., 2015b). Associated discussions with regard to niche regime interactions have remained on system level (Costa et al., 2022; Mylan et al., 2019). The MEF, in contrast, shows how individual-level learning in alliances with start-ups can be cause and outcome of proactive behavior of incumbent firm managers. Additional interactions between top and middle managers in the form of shared sensemaking processes help reflect on diverse internal behavioral patterns and can lead to shifts from passivity to reactivity, or reactivity to proactivity respectively. These cognitive individual-level insights in terms of learning and sensemaking processes present a fundamentally new perspective to transition research, which today, has focused on system-level (e.g. Geels, 2002; Geels and Schot, 2007) and, to some extent, firm-level analyses (e.g. Geels, 2014a; Stalmokaite and Hassler, 2020).

The three presented properties of integrated incumbent firm behavior describe the complexity of sustainability-oriented reorientation and innovation. The co-existence of passivity, reactivity and proactivity, the accumulation of meanings over time and the interactions between individuals, firms and firm environments render reorientation a multi-facetted and contested endeavor. At this point, the question remains how these new insights into integrated incumbent behavior influence the current understanding of incumbent firms in industry transition. Integrated incumbent firm behavior considers an incumbent firm's capacity for proactivity, including the stimulation and acceleration of sustainability-oriented exploration. In this regard, transition studies have stated that the early detection, development and dissemination of radical innovations bears the potential of incumbent firms to contribute to industry transitions (Bergek et al., 2013; Berggren et al., 2015; Chandy and Tellis, 2000). However, as proactive behaviors tend to co-exist with passive and reactive behaviors, an incumbent firms' eventual contribution to industry transitions remains unclear. In the end, does integrated behavior make incumbent firms drivers, laggards or losers of industry transition? The following section will discuss the implications of the three presented properties with regard to a potentially new understanding of incumbent firms in industry transition.

#### 5.2 Understanding of incumbent firms in industry transition

When analyzing the three properties of integrated firm behavior in terms of their implications for industry transition, different conclusions can be drawn. The two properties of co-existence and accumulation seem to provide firms with the potential to moderately contribute to industry transitions (see first two rows in Table 4). This conclusion is largely in line with transition research ascribing a rather reduced role to incumbent firms (Geels, 2002; Geels and Schot, 2007; Kemp et al., 1998). As proactive behavior always co-exists with some degree of passivity and reactivity inside the firm, different preferences have to be considered and balanced. The continuous balancing over time is likely to result in an accumulation of meanings, which makes it difficult, if not impossible, to fully replace established, potentially unsustainable firm practices by those oriented towards sustainability. It is true that bi-directional sensemaking between established and new meanings may help accelerate internal exploration and reorientation. However, as the case of BIGMEAT shows, the outcome of such processes in terms of firm identity and sustainability-oriented in-house innovations tend to be of rather moderate nature. While the developed in-house innovations might increase industry sustainability by providing consumers with sustainable alternatives, their moderateness may impede more fundamental transition. The conclusion can be drawn that if an incumbent firm seeks to proactively drive transitions, the co-existence and accumulation of different behaviors and meanings inside the firm may have a hampering effect on such endeavors. Considering the urgency of profound industry transition regarding pressing sustainability challenges (van der Leeuw et al., 2012), internal incumbent firm processes alone seem to be insufficient to drive industry transition.

Properties of integrated incumbent	Description	Understanding of incumbent firms in indu	stry transition
firm behavior		Per property	Overall
Co-existence	Parallel development of different behaviors and processes inside the firm related to passivity, reactivity and proactivity.	Incumbent firms as actors with the potential to <i>moderately</i> contribute to industry transitions through - Continuous balancing of different behaviors inside the firm, including the exploration of sustainable alternatives	Incumbent firms as actors with the potential to <i>moderately</i> or <i>profoundly</i> contribute to industry transition, depending on the amphasized
Accumulation	Progressive increase of meanings, combining old and	Incumbent firms as actors with the potential to <i>moderately</i> contribute to industry transitions through	property

Table 4: New understanding of incumbent firms in industry transitions

InteractionMutual interlinkages and processes between firm environments as well as between and within firms.Incumbent firms as actors with the potential to <i>profoundly</i> contribute to industry transitions through- Strategic use of causal chains in and between industry environments with innovation of distribution partnerships with innovation-focused (new entrant) firms and exploitation of related alliance learnings - Initiation of shared sensemaking processes for sustainability-oriented exploration		new activities (or business units) within the firm.	- Compromised firm-internal solutions with regard to firm identity and sustainability innovations
•	Interaction	Mutual interlinkages and processes between firm environments as well as between and within firms.	<ul> <li>Incumbent firms as actors with the potential to <i>profoundly</i> contribute to industry transitions through <ul> <li>Strategic use of causal chains in and between industry environments</li> <li>Formation of distribution partnerships with innovation-focused (new entrant) firms and exploitation of related alliance learnings</li> <li>Initiation of shared sensemaking processes for sustainability-oriented exploration</li> </ul></li></ul>

While the properties of co-existence and accumulation might allow for rather moderate incumbent firm contributions to industry transition, the property of interaction holds the potential for more profound firm contributions (see third row in Table 4). First, as the paper by Dijkstra-Silva et al. (2023) suggests, proactive behavior can become very impactful, if it is deliberately oriented towards influencing industry environments. This may be the case even if such behavior is only limited to few, but powerful individuals inside the firm. Proactive top managers can purposefully select measures while considering causal chains within industry environments and, thus, potentially trigger far-reaching industry change. Second, the BIGMEAT case showed how engaging with sustainability-oriented start-ups in the form of distribution partnerships can actively increase the availability of sustainable solutions at the market. Respective alliance learning processes can further develop managers' deliberate intent to contribute to industry transition, resulting in the formation of more alliances with progressively radical value propositions. The outcome for industry transition can be increased. Third, involving middle managers into shared sensemaking early on can accelerate internal reorientation and exploration activities, which, in turn, can shorten the time to market of inhouse innovations. These implications show the crucial importance of multi-level interactions between individuals, between firms and between firm environments for driving industry transitions. Previous research, including the TEF, has addressed firm-environment relationships (e.g. Smink et al., 2015a) as well as interfirm collaborations (e.g. Engwall et al., 2021), focusing on strategic response and adaptation or externally-oriented learning. This study is the first to provide in-depth insights into the complexity, but also crucial role of multi-level interactions involved in industry transitions, particularly key learning and sensemaking processes at the level of individual managers.

Combining the previous elaborations creates a rather ambiguous understanding of incumbent firms in industry transition. Incumbent firms can be both, moderate and profound contributors to industry transition. As all three properties always apply simultaneously, different potentials for industry transition contribution coincide. For instance, an exclusive focus on the property of interaction would not be possible, as all properties are inherently interrelated and thereby complete the complex picture of integrated incumbent firm behavior. This interrelation, however, also opens up opportunities for an increased contribution potential of incumbent firms. In this regard, this paper argues that incumbent firms possess the ability to emphasize specific properties, which can, in turn, increase or decrease another property's influence. For instance, an incumbent firm that initiates a particularly large number of external interactions to internalize new knowledge and drive innovation, will automatically experience a change in coexisting behaviors. Proactive behaviors become more prominent, while passive and reactive behaviors become more subtle. Leveraging multiple interactions can become outcome and cause of proactive behaviors and thereby increase a firm's contribution to sustainability transition. This is how, eventually, an incumbent firm can actively influence its role in sustainability-oriented industry transition.

### 6. Conclusion

This paper offers an integrated view of incumbent firm behavior which goes beyond the understanding of incumbents as coherent actors that engage in passive or reactive behavior in industry transitions. In line with recent conceptual research proposing a heterogeneous understanding of incumbent firm behavior (Magnusson and Werner, 2022; Turnheim and Sovacool, 2020), the works of this dissertation find evidence of simultaneous passive, reactive and proactive behaviors inside incumbent firms, varying according to business units and management levels. Extending Geels' (2014a) *Triple* Embeddedness Framework, the developed *Multi* Embeddedness Framework proposes that integrated incumbent firm behavior results from incumbent firms' embeddedness in multiple environments and regime influences. New firm- and individual-level insights detail processes and outcomes of integrated firm behavior and reveal three crucial behavioral properties: Co-existence, accumulation and interaction. Discussing these properties in the context of industry transition suggests that incumbent firm managers may actively increase their firm's contribution of sustainability-oriented industry transitions by emphasizing multi-level interactions between firm environments, firms and individuals. By making possible interactions explicit, the MEF could
be an adequate framework to guide firms in their endeavor to contribute to industry transition in a meaningful way. The insights put forward by this paper have several implications for management and research.

# **6.1 Management implications**

The discussion section showed that, if multi-level interactions are deliberately emphasized and promoted, incumbent firms might be able to increase their contribution to sustainability-oriented industry transition. The property of interaction, however, relies on the prevalence of proactive behaviors amongst individual managers. To date, a vast number research studies has found largely passive and reactive incumbent firm behavior (e.g. Wells and Nieuwenhuis, 2012; Smink et al., 2015a), with cases like BIGMEAT being the exception rather than the rule. Thus, identifying measures for the stimulation and promotion of proactive behaviors inside the firm could be a valuable undertaking.

Based on the insights of this dissertation, proactive behavior is likely to be displayed by few individual managers, possibly at top management level. As suggested by the BIGMEAT studies, top managers can stimulate and accelerate the progressive dissemination of proactive behavior within the firm through two related measures: Fostering a culture of sensemaking and fostering a culture of learning. First, fostering a culture of sensemaking across different managerial levels requires formalization and active promotion. In this regard, the sensemaking literature suggests several measures and tools to promote sensemaking, including the initiation of change workshops (Ala-Laurinaho et al., 2017), software-based concept mapping (Faily et al., 2012), strategic linguistic terms (Jalonen et al., 2018) and storytelling (Bietti et al., 2018). The aim of these measures is to encourage organizational members to make sense of uncertain or novel situations and to become more reflective about personal knowledge and assumptions, firm-internal practices and external developments. As the BIGMEAT case shows, involving middle management into sensemaking practices is crucial for the operationalization of proactive activities. If managers of different hierarchical levels continuously alternate between making sense of environmental developments and making sense of internal developments, including innovation activities and changes to firm identity, they might be able to detect market opportunities early on and translate them effectively into innovation activities on firm level.

Second, to increase the extent and radicality of exploration and innovation activities, this research's findings advise top managers to promote a culture of learning. At BIGMEAT, learning in the context of alliances profoundly increased the extent and radicality of exploration

activities and ultimately contributed to industry transition. These processes were fundamentally facilitated by managers' openness towards new knowledge, which has been deemed crucial for the pursuit of proactive innovation activities (Calantone et al., 2002). Organization and management research identified key antecedents of learning orientation. For instance, top managers can promote learning by establishing an organic organizational structure based on shared responsibilities, flexibility and open communication (Fiol and Lyles, 1985; Rebelo and Duarte Gomes, 2011). Particularly, if top managers encourage participative decision-making, employees are more likely to freely voice their opinions, share information and exchange ideas (Flores et al., 2012). To foster participation and open communication, top managers can adopt a transformational leadership approach through which they encourage creativity, communicate inspirational visions and act as role models for positive change (Coad and Berry, 1998).

As these elaborations show, a sensemaking culture is crucial for the initiation of exploration activities. Managers have to become aware of potential external opportunities and threats, the firms' potential capabilities and inefficiencies as well as their personal abilities and flaws. Thereupon, a strong learning orientation can facilitate the exchange of internal knowledge, the internalization of external knowledge and the respective continuous adaptation of exploration activities. At BIGMEAT, the exploration of the alternative protein field was additionally facilitated by the long-term thinking of top managers, which, in turn, was based on the family firm's general sustainability orientation. In this regard, previous research confirms that sustainability orientation can facilitate innovations (Jin et al., 2018). The same research (Jin et al., 2018) also found that innovation culture can facilitate sustainability orientation. Indeed, the alliances that were formed as part of BIGMEAT's exploration created crucial learnings that further increased sustainability awareness amongst managers. These insights suggest that proactivity can be source and outcome of sustainability orientation. While the mentioned aspects, sensemaking, learning and sustainability orientation, are crucial antecedents for sustainability-oriented proactivity in incumbent firms, this research also clearly suggests that exclusively proactive behavior and/or the complete overhaul of an incumbent firm's established meanings seems highly unlikely. Incumbent firm managers will always have to face the challenge of balancing different internal behaviors, exploitation and exploration, reluctance and ambition, tradition and progress. This study merely offers a way for them to navigate through this balancing act while actively pushing for the maximum outcome for sustainability.

# 6.2 Limitations and future research

This research comes with several limitations which offer opportunities for future research. Most of the empirical insights that resulted in the extension of the TEF were based on the case of BIGMEAT, which constitutes a unique example of reorientation in the meat industry. To increase external validity of the new framework, future research could test the framework's main propositions in different firms and industries. At BIGMEAT, proactive behavior was limited to few individuals of the top management team, while middle managers exerted largely passive behavior in the early phases of reorientation. The question remains whether this study's proposals with regard to properties of integrated firm behavior and implications for industry transition would still hold true if the internal occurrence of individual behaviors was different. In this regard, for instance, some studies have pointed to a much higher involvement of middle managers in organizational change efforts (Sharma and Good, 2013), while others have found diversified motivations and behaviors among managers of the same hierarchical level (Visser and Crane, 2010).

Further, the case study data did not reveal distinguishable insights into the relationship between individual-level processes and developments in socio-political environments. While the data does indicate impact on the economic market environments and different regime structures, implications with regard to civil society and politics are not further discussed. It is left to future research to investigate the implications of integrated firm behavior on such environments and to test whether the suggested properties and their transition impact potential remain the same. Due to the MEF's multi-level approach, such testing would require in-depth data about individual-level processes in relation to industry developments, ideally with empirical insights into transition impacts. While corresponding with Geels' (2014a, p.275) suggestion to add "another layer of complexity" through the analysis of individual-level processes, it should not be the aim of future empirical studies to cover all aspects of the MEF simultaneously. Instead, future studies could focus on particular sub-aspects. For instance, due to the nature of the BIGMEAT case, this study analyzed interactions in the form of alliances with start-ups. Future research could transfer the MEF's propositions regarding interactions to other alliance contexts, including alliances with universities (Orecchini et al., 2012) or suppliers (Berardi and de Brito, 2021). Based on the analyzed interactions with actors in other industry environments, further extensions of the framework might formulate new propositions and thereby progressively extend the complex picture of incumbent-driven industry transition towards sustainability.

# References

Ajzen, I. (1991). The theory of planned behavior. *Organizational Behavior and Human Decision Processes*, 50(2), 179-211. https://doi.org/10.1016/0749-5978(91)90020-T

Ala-Laurinaho, A., Kurki, A.L., & Abildgaard, J.S. (2017). Supporting sensemaking to promote a systemic view of organizational change–contributions from activity theory, *Journal of Change Management*, Vol. 17 No. 4, pp. 367-387. https://doi.org/10.1080/14697017.2017.1309566.

Ampe, K., Paredis, E., Asveld, L., Osseweijer, P., & Block, T. (2021). Incumbents' enabling role in niche-innovation: Power dynamics in a wastewater project. *Environmental Innovation and Societal Transitions*, 39, 73-85. https://doi.org/10.1016/j.eist.2021.03.004

Ardichvili, A., Cardozo, R., & Ray, S. (2003). A theory of entrepreneurial opportunity identification and development. *Journal of Business Venturing*, 18(1), 105-123. https://doi.org/10.1016/S0883-9026(01)00068-4

Augenstein, K., & Palzkill, A. (2015). The dilemma of incumbents in sustainability transitions: a narrative approach. *Administrative Sciences*, 6(1), 1. https://doi.org/10.3390/admsci6010001

Berardi, P. C., & de Brito, R. P. (2021). Supply chain collaboration for a circular economy-From transition to continuous improvement. *Journal of Cleaner Production*, *328*, 129511. https://doi.org/10.1016/j.jclepro.2021.129511

Bergek, A., Berggren, C., Magnusson, T., & Hobday, M. (2013). Technological discontinuities and the challenge for incumbent firms: Destruction, disruption or creative accumulation?. *Research Policy*, *42*(6-7), 1210-1224. https://doi.org/10.1016/j.respol.2013.02.009

Berggren, C., Magnusson, T., & Sushandoyo, D. (2015). Transition pathways revisited: Established firms as multi-level actors in the heavy vehicle industry. *Research Policy*, 44(5), 1017-1028. https://doi.org/10.1016/j.respol.2014.11.009

Bidmon, C. M., & Knab, S. F. (2018). The three roles of business models in societal transitions: New linkages between business model and transition research. *Journal of Cleaner Production*, *178*, 903-916. https://doi.org/10.1016/j.jclepro.2017.12.198

Bietti, L. M., Tilston, O., & Bangerter, A. (2019). Storytelling as adaptive collective sensemaking. *Topics in Cognitive Science*, *11*(4), 710-732. https://doi.org/10.1111/tops.12358

Calantone, R. J., Cavusgil, S. T., & Zhao, Y. (2002). Learning orientation, firm innovation capability, and firm performance. *Industrial Marketing Management*, *31*(6), 515-524. https://doi.org/10.1016/S0019-8501(01)00203-6

Chandy, R. K., & Tellis, G. J. (2000). The incumbent's curse? Incumbency, size, and radical product innovation. *Journal of Marketing*, 64(3), 1-17. https://doi.org/10.1509/jmkg.64.3.1.18033 Coad, A. F., & Berry, A. J. (1998). Transformational leadership and learning orientation. *Leadership & Organization Development Journal*, *19*(3), 164-172. https://doi.org/10.1108/01437739810210211

Cohen, N., & Ilieva, R. T. (2015). Transitioning the food system: A strategic practice management approach for cities. *Environmental Innovation and Societal Transitions*, *17*, 199-217. https://doi.org/10.1016/j.eist.2015.01.003

Cornelissen, J.P. & Clarke, J.S. (2010), Imagining and rationalizing opportunities: inductive reasoning and the creation and justification of new ventures, *Academy of Management Review*, Vol. 35 No. 4, pp. 539-557. https://doi.org/10.5465/amr.35.4.zok539

Costa, I., Bui, S., De Schutter, O., & Dedeurwaerdere, T. (2022). A network perspective to niche-regime interactions and learning at the regime level. *Environmental Innovation and Societal Transitions*, *43*, 62-79. https://doi.org/10.1016/j.eist.2022.03.001

Crippa, M., Solazzo, E., Guizzardi, D., Monforti-Ferrario, F., Tubiello, F. N., & Leip, A. J. N. F. (2021). Food systems are responsible for a third of global anthropogenic GHG emissions. *Nature Food*, 2(3), 198-209. https://doi.org/10.1038/s43016-021-00225-9

DiMaggio, P. J., & Powell, W. W. (1983). The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review*, 147-160. https://doi.org/10.2307/2095101

Dijkstra-Silva, S., Hübel, C. & Schaltegger, S (2023): Company contributions to sustainability transitions of markets: An innovation management perspective. *To be submitted* 

Dosi, G., 2000. Innovation, Organization and Economic Dynamics. Edward Elgar, Cheltenham, UK.

Doz, Y.L. (1996). The evolution of cooperation in strategic alliances: initial conditions or learning processes? *Strategic Management Journal*, 17, 55–83. https://doi.org/10.1002/smj.4250171006

El Bilali, H. (2019). Research on agro-food sustainability transitions: A systematic review of research themes and an analysis of research gaps. *Journal of Cleaner Production*, 221, 353-364. https://doi.org/10.1016/j.jclepro.2019.02.232

Engwall, M., Kaulio, M., Karakaya, E., Miterev, M., & Berlin, D. (2021). Experimental networks for business model innovation: A way for incumbents to navigate sustainability transitions?. *Technovation*, *108*, 102330. https://doi.org/10.1016/j.technovation.2021.102330

Faily, S., Lyle, J., Paul, A., Atzeni, A., Blomme, D., Desruelle, H., & Bangalore, K. (2012).
Requirements sensemaking using concept maps. In *Human-Centered Software Engineering:* 4th International Conference, HCSE 2012, Toulouse, France, October 29-31, 2012.
Proceedings 4 (pp. 217-232). Springer Berlin Heidelberg.

Fiala, N. (2008). Meeting the demand: an estimation of potential future greenhouse gas emissions from meat production. *Ecological Economics*, 67(3), 412-419. https://doi.org/10.1016/j.ecolecon.2007.12.021 Fiol, C. M., & Lyles, M. A. (1985). Organizational learning. *Academy of management review*, *10*(4), 803-813. https://doi.org/10.5465/amr.1985.4279103

Flores, L. G., Zheng, W., Rau, D., & Thomas, C. H. (2012). Organizational learning: Subprocess identification, construct validation, and an empirical test of cultural antecedents. *Journal of Management*, *38*(2), 640-667. https://doi.org/10.1177/0149206310384631

Geels, F. W. (2002). Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. *Research Policy*, *31*(8-9), 1257-1274. https://doi.org/10.1016/S0048-7333(02)00062-8

Geels, F. W. (2014a). Reconceptualising the co-evolution of firms-in-industries and their environments: Developing an inter-disciplinary Triple Embeddedness Framework. *Research Policy*, *43*(2), 261-277. https://doi.org/10.1016/j.respol.2013.10.006

Geels, F. W. (2014b). Regime resistance against low-carbon transitions: introducing politics and power into the multi-level perspective. *Theory, Culture & Society, 31*(5), 21-40. https://doi.org/10.1177/0263276414531627

Geels, F. W. (2020). Micro-foundations of the multi-level perspective on socio-technical transitions: Developing a multi-dimensional model of agency through crossovers between social constructivism, evolutionary economics and neo-institutional theory. *Technological Forecasting and Social Change*, *152*, 119894. https://doi.org/10.1016/j.techfore.2019.119894

Geels, F. W., & Schot, J. (2007). Typology of sociotechnical transition pathways. *Research Policy*, *36*(3), 399-417. https://doi.org/10.1016/j.respol.2007.01.003

Gioia, D. A., & Chittipeddi, K. (1991). Sensemaking and sensegiving in strategic change initiation. Strategic management journal, 12(6), 433-448. https://doi.org/10.1002/smj.4250120604

Gregory, N. G., & Grandin, T. (Eds.). (2007). Animal welfare and meat production. CABI.

Hill, R.C. and Levenhagen, M. (1995). Metaphors and mental models: sensemaking and sensegiving in innovative and entrepreneurial activities, *Journal of Management*, 21(6), 1057-1074, https://doi.org/10.1016/0149-2063(95)90022-5

Hoek, A. C., Malekpour, S., Raven, R., Court, E., & Byrne, E. (2021). Towards environmentally sustainable food systems: decision-making factors in sustainable food production and consumption. *Sustainable Production and Consumption*, 26, 610-626. https://doi.org/10.1016/j.spc.2020.12.009

Holm, P. (1995). The dynamics of institutionalization: Transformation processes in Norwegian fisheries. *Administrative Science Quarterly*, 398-422. https://doi.org/10.2307/2393791

Hübel, C. (2022). Entrepreneurship-driven organizational transformation for sustainability: a sensemaking lens. *Journal of Organizational Change Management*, *35*(1), 240-256. https://doi.org/10.1108/JOCM-03-2021-0067 Hübel, C., & Schaltegger, S. (2022). Barriers to a sustainability transformation of meat production practices-An industry actor perspective. *Sustainable Production and Consumption*, *29*, 128-140. https://doi.org/10.1016/j.spc.2021.10.004

Hübel, C., Weissbrod, I., & Schaltegger, S. (2022). Strategic alliances for corporate sustainability innovation: The 'how'and 'when'of learning processes. *Long Range Planning*, *55*(6), 102200. https://doi.org/10.1016/j.lrp.2022.102200

Inkpen, A.C., 1998. Learning and knowledge acquisition through international strategic alliances. Academy of Management Perspectives, 12, 69–80. https://doi.org/10.5465/ame.1998.1333953.

Inkpen, A. C., & Currall, S. C. (2004). The coevolution of trust, control, and learning in joint ventures. *Organization Science*, *15*(5), 586-599. https://doi.org/10.1287/orsc.1040.0079

Jalonen, K., Schildt, H., & Vaara, E. (2018). Strategic concepts as micro-level tools in strategic sensemaking. *Strategic Management Journal*, *39*(10), 2794-2826. https://doi.org/10.1002/smj.2924

Kaffka, G.A., Singaram, R., & Kraaijenbrink, J. (2020). New venture creation and the development of entrepreneurial cognition. *Academy of Management Proceedings*, 2020(1), 20704, https://doi.org/10.5465/AMBPP.2020.37

Karltorp, K., & Sandén, B. A. (2012). Explaining regime destabilisation in the pulp and paper industry. *Environmental Innovation and Societal Transitions*, *2*, 66-81. https://doi.org/10.1016/j.eist.2011.12.001

Kemp, R., Schot, J., & Hoogma, R. (1998). Regime shifts to sustainability through processes of niche formation: the approach of strategic niche management. *Technology Analysis & Strategic Management*, *10*(2), 175-198. https://doi.org/10.1080/09537329808524310

Kishna, M., Negro, S., Alkemade, F., & Hekkert, M. (2017). Innovation at the end of the life cycle: discontinuous innovation strategies by incumbents. *Industry and Innovation*, 24(3), 263-279. https://doi.org/10.1080/13662716.2016.1226163

Klewitz, J., & Hansen, E. G. (2014). Sustainability-oriented innovation of SMEs: a systematic review. *Journal of cleaner production*, 65, 57-75. https://doi.org/10.1016/j.jclepro.2013.07.017

Kollmuss, A., & Agyeman, J. (2002). Mind the gap: why do people act environmentally and what are the barriers to pro-environmental behavior?. *Environmental Education Research*, 8(3), 239-260. https://doi.org/10.1080/13504620220145401

Krippner, G.R., 2001. The elusive market: embeddedness and the paradigm of eco-nomic sociology. *Theory and Society*, 30 (6), 775–810. https://doi.org/10.1023/A:1013330324198

Lee, D., & Hess, D. J. (2019). Incumbent resistance and the solar transition: Changing opportunity structures and framing strategies. *Environmental Innovation and Societal Transitions*, *33*, 183-195. https://doi.org/10.1016/j.eist.2019.05.005

Loorbach, D., & Wijsman, K. (2013). Business transition management: exploring a new role for business in sustainability transitions. *Journal of Cleaner Production*, 45, 20-28. https://doi.org/10.1016/j.jclepro.2012.11.002

Magnusson, T., & Werner, V. (2022). Conceptualisations of incumbent firms in sustainability transitions: Insights from organisation theory and a systematic literature review. *Business Strategy and the Environment*, 32(2), 903-919. https://doi.org/10.1002/bse.3081

Maitlis, S. (2005). The social processes of organizational sensemaking. *Academy of Management Journal*, 48(1), 21-49. https://doi.org/10.5465/amj.2005.15993111

Mellinger, N. (2000). *Fleisch: Ursprung und Wandel einer Lust: eine kulturanthropologische Studie*. Campus Verlag.

Mylan, J., Morris, C., Beech, E., & Geels, F. W. (2019). Rage against the regime: Niche-regime interactions in the societal embedding of plant-based milk. *Environmental Innovation and Societal Transitions*, 31, 233-247. https://doi.org/10.1016/j.eist.2018.11.001

Onufrey, K., & Bergek, A. (2021). Transformation in a mature industry: The role of business and innovation strategies. *Technovation*, *105*, 102190. https://doi.org/10.1016/j.technovation.2020.102190

Orecchini, F., Valitutti, V., & Vitali, G. (2012). Industry and academia for a transition towards sustainability: advancing sustainability science through university–business collaborations. *Sustainability Science*, 7, 57-73. https://doi.org/10.1007/s11625-011-0151-3

O'Reilly III, C. A., & Tushman, M. L. (2011). Organizational ambidexterity in action: How managers explore and exploit. *California Management Review*, *53*(4), 5-22. https://doi.org/10.1525/cmr.2011.53.4.5

Penna, C. C., & Geels, F. W. (2015). Climate change and the slow reorientation of the American car industry (1979–2012): An application and extension of the Dialectic Issue LifeCycle (DILC) model. *Research Policy*, *44*(5), 1029-1048. https://doi.org/10.1016/j.respol.2014.11.010

Polanyi, K., 1944. *The Great Transformation: The Political and Economic Origins of Our Time*. Beacon Press, Boston.

Raven, R., Kern, F., Verhees, B., & Smith, A. (2016). Niche construction and empowerment through socio-political work. A meta-analysis of six low-carbon technology cases. *Environmental Innovation and Societal Transitions*, *18*, 164-180. https://doi.org/10.1016/j.eist.2015.02.002

Rebelo, T. M., & Duarte Gomes, A. (2011). Conditioning factors of an organizational learning culture. *Journal of workplace learning*, *23*(3), 173-194. https://doi.org/10.1108/13665621111117215

Rovanto, I. K., & Bask, A. (2021). Systemic circular business model application at the company, supply chain and society levels—A view into circular economy native and adopter

companies. *Business Strategy and the Environment*, *30*(2), 1153-1173. https://doi.org/10.1002/bse.2677

Schaltegger, S., & Wagner, M. (2011). Sustainable entrepreneurship and sustainability innovation: categories and interactions. *Business Strategy and the Environment*, 20(4), 222-237. https://doi.org/10.1002/bse.682

Schaltegger, S., Lüdeke-Freund, F., & Hansen, E. G. (2016). Business models for sustainability: A co-evolutionary analysis of sustainable entrepreneurship, innovation, and transformation. *Organization & Environment*, 29(3), 264-289. https://doi.org/10.1177/1086026616633272

Schot, J., & Geels, F. W. (2008). Strategic niche management and sustainable innovation journeys: theory, findings, research agenda, and policy. *Technology Analysis & Strategic Management*, 20(5), 537-554. https://doi.org/10.1080/09537320802292651

Sharma, G., & Good, D. (2013). The work of middle managers: Sensemaking and sensegiving for creating positive social change. *The Journal of Applied Behavioral Science*, *49*(1), 95-122. https://doi.org/10.1177/0021886312471375

Smink, M. M., Hekkert, M. P., & Negro, S. O. (2015a). Keeping sustainable innovation on a leash? Exploring incumbents' institutional strategies. *Business Strategy and the Environment*, 24(2), 86-101. https://doi.org/10.1002/bse.1808

Smink, M., Negro, S. O., Niesten, E., & Hekkert, M. P. (2015b). How mismatching institutional logics hinder niche–regime interaction and how boundary spanners intervene. *Technological Forecasting and Social Change*, *100*, 225-237. https://doi.org/10.1016/j.techfore.2015.07.004

Smith, A., & Raven, R. (2012). What is protective space? Reconsidering niches in transitions to sustainability. *Research Policy*, *41*(6), 1025-1036. https://doi.org/10.1016/j.respol.2011.12.012

Stalmokaitė, I., & Hassler, B. (2020). Dynamic capabilities and strategic reorientation towards decarbonisation in Baltic Sea shipping. *Environmental Innovation and Societal Transitions*, *37*, 187-202. https://doi.org/10.1016/j.eist.2020.09.002

Steen, M., & Weaver, T. (2017). Incumbents' diversification and cross-sectorial energy industry dynamics. *Research Policy*, *46*(6), 1071-1086. https://doi.org/10.1016/j.respol.2017.04.001

Steinfeld, H., Gerber, P., Wassenaar, T. D., Castel, V., Rosales, M., Rosales, M., & de Haan, C. (2006). *Livestock's long shadow: environmental issues and options*. Food & Agriculture Organization.

Stoll-Kleemann, S., & Schmidt, U. J. (2017). Reducing meat consumption in developed and transition countries to counter climate change and biodiversity loss: a review of influence factors. *Regional Environmental Change*, *17*, 1261-1277. https://doi.org/10.1007/s10113-016-1057-5

Springmann, M., Clark, M., Mason-D'Croz, D., Wiebe, K., Bodirsky, B. L., Lassaletta, L., ... & Willett, W. (2018). Options for keeping the food system within environmental limits. *Nature*, 562(7728), 519-525. https://doi.org/10.1038/s41586-018-0594-0

Sydow, J., Schreyögg, G., & Koch, J. (2009). Organizational path dependence: Opening the black box. *Academy of Management Review*, 34(4), 689-709. https://doi.org/10.5465/amr.34.4.zok689

Turnheim, B., & Sovacool, B. K. (2020). Forever stuck in old ways? Pluralising incumbencies in sustainability transitions. *Environmental Innovation and Societal Transitions*, 35, 180-184. https://doi.org/10.1016/j.eist.2019.10.012

Turnheim, B., & Geels, F. W. (2019). Incumbent actors, guided search paths, and landmark projects in infra-system transitions: Re-thinking Strategic Niche Management with a case study of French tramway diffusion (1971–2016). *Research Policy*, *48*(6), 1412-1428. https://doi.org/10.1016/j.respol.2019.02.002

Unruh, G. C. (2000). Understanding carbon lock-in. *Energy Policy*, 28(12), 817-830. https://doi.org/10.1016/S0301-4215(00)00070-7

Van der Leeuw, S., Wiek, A., Harlow, J., & Buizer, J. (2012). How much time do we have? Urgency and rhetoric in sustainability science. *Sustainability Science*, *7*, 115-120. https://doi.org/10.1007/s11625-011-0153-1

Van de Ven, A. H. (1995). Managing the process of organizational innovation. In G.P. Huber & W.H. Glick (Eds.). *Organizational change and redesign: Ideas and insights for improving performance* (269-294). Oxford: Oxford University Press.

Van Mossel, A., van Rijnsoever, F. J., & Hekkert, M. P. (2018). Navigators through the storm: A review of organization theories and the behavior of incumbent firms during transitions. *Environmental Innovation and Societal Transitions*, 26, 44-63. https://doi.org/10.1016/j.eist.2017.07.001

Visser, W., & Crane, A. (2010). Corporate sustainability and the individual: Understanding what drives sustainability professionals as change agents. *Available at SSRN 1559087*. http://dx.doi.org/10.2139/ssrn.1559087

Vittersø, G., & Tangeland, T. (2015). The role of consumers in transitions towards sustainable food consumption. The case of organic food in Norway. *Journal of Cleaner Production*, 92, 91-99. https://doi.org/10.1016/j.jclepro.2014.12.055

Vuorio, A. M., Puumalainen, K., & Fellnhofer, K. (2018). Drivers of entrepreneurial intentions in sustainable entrepreneurship. *International Journal of Entrepreneurial Behavior & Research*, 24(2), 359-381. https://doi.org/10.1108/IJEBR-03-2016-0097

Vermunt, D. A., Negro, S. O., Van Laerhoven, F. S. J., Verweij, P. A., & Hekkert, M. P. (2020). Sustainability transitions in the agri-food sector: how ecology affects transition dynamics. *Environmental Innovation and Societal Transitions*, *36*, 236-249. https://doi.org/10.1016/j.eist.2020.06.003 Walker, P., Rhubart-Berg, P., McKenzie, S., Kelling, K., & Lawrence, R. S. (2005). Public health implications of meat production and consumption. *Public Health Nutrition*, 8(4), 348-356. https://doi.org/10.1079/PHN2005727

Weick, K.E. (1979). The Social Psychology of Organizing, 2nd ed., McGraw-Hill, New York.

Weiblen, T., & Chesbrough, H. W. (2015). Engaging with startups to enhance corporate innovation. *California Management Review*, *57*(2), 66-90. https://doi.org/10.1525/cmr.2015.57.2.66.

Wells, P., & Nieuwenhuis, P. (2012). Transition failure: Understanding continuity in the automotive industry. *Technological Forecasting and Social Change*, *79*(9), 1681-1692. https://doi.org/10.1016/j.techfore.2012.06.008

Werner, V., Flaig, A., Magnusson, T., & Ottosson, M. (2022). Using dynamic capabilities to shape markets for alternative technologies: A comparative case study of automotive incumbents. *Environmental Innovation and Societal Transitions*, *42*, 12-26. https://doi.org/10.1016/j.eist.2021.10.031

# **B.** Rigidity and lock-in

Hübel, C. & Schaltegger, S. (2022): Barriers to a sustainability transformation of meat production practices - An industry actor perspective. Sustainable Production and Consumption 29, 128 – 140. https://doi.org/10.1016/j.spc.2021.10.004

Contents lists available at ScienceDirect







journal homepage: www.elsevier.com/locate/spc

# Barriers to a sustainability transformation of meat production practices - An industry actor perspective

# Charlott Hübel, Stefan Schaltegger\*

Centre for Sustainability Management (CSM), Leuphana University Lüneburg, Universitätsallee 1, D-21335 Lüneburg, Germany

#### ARTICLE INFO

Article history: Received 3 May 2021 Revised 22 September 2021 Accepted 4 October 2021 Available online 6 October 2021

Editor: Dr. Peter Bradley

*Keywords:* Sustainability transformation Meat production Barriers Behavior change Industry actor

# ABSTRACT

The negative impacts caused by current industrial meat production challenge industry actors to transform meat production towards more sustainable practices. However, despite the necessity for change and the availability of various sustainability solutions, transformational shifts have failed to materialize, so far. This study applies and extends the Kollmuss-Agyeman model of pro-environmental behavior to the context of producers and analyzes their perceived barriers to the transformation of meat production practices. The qualitative empirical study is based on 23 interviews with actors along the meat production chain. The analysis reveals that industry actors' perceived barriers are highly complex. This complexity results from (1) multiple interactions among and between internal and external influencing factors and (2) the simultaneous existence of barriers that reinforce the status quo on the one hand and restrain the pursuit of sustainable production practices on the other hand. Based on these findings, opportunities for overcoming barriers are discussed in today's context. This study's contributions are twofold: First, the analysis complements previous research on barriers to sustainable meat production practices by detailing barriers and their interactions for actors along the entire meat production chain. Second, this investigation extends the Kollmuss-Agyeman model by specifying interactions of external influencing factors and by differentiating reinforcement and restrain barriers relevant for producers.

© 2021 Published by Elsevier B.V. on behalf of Institution of Chemical Engineers.

# 1. Introduction

The global consumption of meat products is expected to rise by 73% until 2050, with the vast majority of products being sourced and manufactured in industrial settings (FAO, 2011). Industrial meat production practices, however, cause major negative impacts on land, water, biodiversity, animal and consumer health (Steinfeld et al., 2006; Ilea, 2009; Rossi and Garner, 2014). Increasing concerns about the sustainability of such practices have led to calls for a sustainability transformation of meat production implying a *fundamental* reconsideration of how and how much meat is produced (Nierenberg, 2006; Nalau and Handmer, 2015). Transformative change of meat production practices can include, amongst others, a shift to extensive, agro-ecological forms of livestock farming (Escribano, 2016; Herren et al., 2015; Nardone et al., 2004) or to the production of alternative protein sources such as plantbased meat, cultivated meat or mycoproteins (Bhat et al., 2017; Hashempour-Baltork et al., 2020).

Established industry actors play a crucial role in driving transformational change of industries by implementing and supporting

\* Corresponding author. *E-mail address:* stefan.schaltegger@leuphana.de (S. Schaltegger). sustainable production practices (Schaltegger and Hansen, 2017; Schaltegger et al., 2016). In the meat industry, 'industry actors' refers to individuals and organizations along the production chain, who are involved in the sourcing, production, processing and distribution of meat (as e.g. used by Zokaei and Simons, 2006). Despite the availability of various sustainability approaches, however, meat production practices of industry actors have not yet amounted to fundamental change in the number of animals slaughtered, in the share of organic or agro-ecological farming or in the availability of alternative protein products (Allievi et al., 2015; Chemnitz and Wenz, 2021). This study's purpose is to identify causes for meat industry actors' reluctance to shift towards sustainable production practices. Due to actors' embeddedness in different contexts (Battilana et al., 2009), various internal factors (i.e. personal, socio-cultural) and external factors (i.e. economic, regulatory) influence decision-making and may jointly create barriers to transformation (Hoek et al., 2021). This study therefore departs from the following research question: 'What are perceived internal and external barriers of industry actors along the meat production chain to transform their production practices towards sustainability?'.

A number of publications address barriers to sustainable production and particularly sustainable food production. While the respective research distinguishes internal (i.e. cognitive and emotional) and external (i.e. economic and regulatory) barriers (Laurett et al., 2021; Stuart et al., 2014; Fleming and Vanclay, 2010; Hoek et al., 2021; Liu et al., 2021), research on transformation barriers as perceived by actors involved in the production of meat is, however, scarce and fragmented (e.g. Läpple and Kelley, 2013; Hyland et al., 2016; Hermann et al., 2016). The existing studies mostly focus on the perspective of farmers but leave a research gap with regard to a broader analysis of barriers and interactions between barriers. Sustainability impacts and influencing factors of meat production are regarded as highly complex (Stoll-Kleemann and Schmidt, 2017) and a comprehensive analysis of perceived barriers to sustainable agricultural production is considered a crucial prerequisite for achieving sustainable development (Laurett et al., 2021). To date, however, no study has systematically investigated internal and external barriers as perceived by different actors along the meat production chain, including farmers, livestock traders, butchers and meat companies.

Previous research highlights the potential of Kollmuss and Agyeman's (2002) model of pro-environmental behavior (hereinafter referred to as Kollmuss-Agyeman model), which considers internal and external influencing factors as well as their interactions (Siegel et al., 2018; Stoll-Kleemann and Schmidt, 2017; Stoll-Kleemann, 2019). This study applies and transfers the model to the context of meat production by conducting a qualitative empirical analysis along the meat production chain in Germany. The qualitative analysis aims at gaining a more comprehensive picture of barriers as perceived by meat industry actors in order to complement previous research on sustainable production practices in general and meat production practices in particular. While the model has previously been applied to analyze meat consumer behavior (Stoll-Kleemann and Schmidt, 2017) an analysis in the meat producer context is missing. The purpose of such a producer-oriented analysis is to create novel insights from a production actor perspective that may differ from previous analyses and could help to further develop the Kollmuss-Agyeman model.

This study's contributions are twofold: First, it complements previous barrier analyses to sustainable meat production by revealing and detailing the complexity of transformation barriers as perceived by meat industry actors. The complexity results from (1) interactions among and between internal and external influencing factors and (2) the simultaneous existence of barriers that reinforce the status quo on the one hand and restrain the pursuit of sustainable production practices on the other hand. Second, this study extends the Kollmuss-Agyeman model for the context of producers by specifying interactions of external influencing factors and by differentiating reinforcement and restraint barriers.

The following section, Section 2, reviews the relevant literature and introduces the Kollmuss-Agyeman model. Section 3 outlines the applied methods, Section 4 presents the results and Section 5 discusses the empirical findings in light of existing research and theory. The concluding section, Section 6, outlines potential limitations and proposes implications for research and practice.

#### 2. Literature review

#### 2.1. Barriers to sustainable production practices

There is a considerable amount of research analyzing barriers to sustainable production practices in general (e.g. Klitkou et al., 2015; De Jesus & Medonca, 2019; Gupta et al., 2020; Guenther et al., 2013) and sustainable food production practices in particular (Laurett et al., 2021; Stuart et al., 2014; Fleming and Vanclay, 2010; Hoek et al., 2021; Liu et al., 2021; Rodriguez et al., 2009). Many studies distinguish between internal and external barriers. Internal barriers address actors' lack of sustainability

knowledge and awareness (Laurett et al., 2021; Fleming and Vanclay, 2010), cognitive disconnects (Ives et al., 2018; Pachirat, 2011) as well as negative emotional responses such as apathy and delegation (Uusi-Rauva and Heikkurinen, 2013; Festinger, 1957). Some studies investigate barriers of disempowerment and dependencies amongst smaller industry actors arising from power asymmetries between big and small industry actors (Virah-Sawmy et al., 2019; Belaya et al., 2009; Talay et al., 2018).

External barriers identified in the research literature include infrastructure and technology 'lock-in' (Klitkou et al., 2015; Unruh, 2000; Gupta et al., 2020) as well as various economic and regulatory factors of influence. Economic barriers arise from market factors such as insufficient consumer demand (Guenther et al., 2013; Gupta et al., 2020) as well as organizational factors such as sunk costs, i.e. irreversible investments into established facilities, economies of scale, i.e. cost advantages gained through an increased production scale (Klitkou et al., 2015) as well as uncertainties about the return on investments (Gupta et al., 2020; David, 1985). Regulatory barriers can arise from inhibiting regulations, a lack of subsidies (Laurett et al., 2021) or conflicting policies (De Jesus and Mendonça, 2018), i.e. policies aiming for sustainability improvements for one aspect, but simultaneously causing sustainability problems for another aspect. A few studies address interactions between barriers (Hoek et al., 2021; Klitkou et al., 2015; Luthra et al., 2014). These interactions may increase the complexity of barriers to sustainability-oriented changes in production practices, but also the difficulty to identify starting points for transformation.

Some research focuses on barriers for sustainable meat production. Table 1 lists studies that deal with barriers to specific sustainable practices for different livestock types, whilst most studies adopt the perspective of farmers. The identification of barriers ranges from economic and regulatory to personal barriers. Contextual barriers regarding infrastructure, subsidies, information and training are mentioned most in the studies.

While the presented studies offer first insights into barriers to sustainability-oriented change in meat production, three important aspects have not been analyzed to date: First, by mainly focusing on livestock farmers and producers, the perceptions of further actors along the meat production chain has not been investigated so far. Second, the studies identify internal and external barriers, while interactions between them still remain to be analyzed. Third, with the exception of Läpple and Kelley (2013), no particular model or theory has informed such analyses. Läpple and Kelley (2013) applied the Theory of Planned Behavior (Ajzen, 1991) to identify the determinants of Irish cattle and sheep farmers' intentions to convert to organic farming. While the Theory of Planned Behavior can predict and explain a range of behaviors, a comprehensive account of environment, economic and regulatory factors that may influence actors' behaviors is still missing. To expand the existing research knowledge on barriers of sustainability transformation, the following analysis applies the Kollmuss-Agyeman model to empirically investigate internal and external barriers to changes in meat production practices.

#### 2.2. Model of pro-environmental behavior

The Kollmuss-Agyeman model offers a comprehensive approach to barrier analysis and aims to explain behavioral change on the basis of internal and external factors influencing transformation decisions and processes. The model is well suited for analyzing meat producers' transformation barriers for two reasons as explained in the following.

First, by considering both internal and external influencing factors as well as the interactions between them, the model not only accommodates for industry actors' embeddedness in and engage-

#### Table 1

Previous research on barriers to sustainable meat production.

Author(s)	Perspective or actor group	Study focus	Model/theory	Identified barriers to change
Läpple and Kelley (2013)	Conventional cattle and sheep farmers in Ireland	Understanding the uptake of organic farming	Theory of Planned Behavior	Insufficient subsidy payments, technical inability, lack of farm management
Gwin (2009)	Grass-fed and organic beef producer groups in the U.S.	Innovation and challenges for grass-fed beef in the U.S.	-	Conventional meat infrastructure, difficulty to balance increased production speed and sustainability goals, low availability of raw materials
Panahzadeh Parikhani et al. (2015)	Livestock farmers of Meshkinshahr	Barriers to the application of Good Agricultural Practices	-	Five barrier categories: Infrastructure barriers, informational- educational barriers, institutional-support barriers, personal barriers, economical barriers
Newton and Blaustein-Rejto (2021)	Rural farmers and livestock producers in the U.S.	Opportunities and challenges of plant-based and cultured meat production	-	Fear of loss of livelihood or income, transitioning barriers to the alternative meat sector
Hyland et al. (2016)	Beef and sheep farmers in Wales	Perceptions of climate change and willingness to implement measures	Constructs: Self-identity, behavioral capacity	Avoidance, denial and desensitization through a lack of understanding and awareness
Hermann et al. (2016)	Organic and conventional hog farmers in Germany	Investment behavior and status quo bias	-	Status quo bias
(Virah-Sawmy et al., 2019)	Major companies in the soy-meat food sector, trade between Europe & Brazil	Assessment of the effectiveness of policy instruments	- Constructs: Policy paradigms	Lack of application of instruments due to lack of coordination in market governance; different risk perception and uptake of sustainability measures along the supply chain



Fig. 1. Kollmuss and Agyeman's (2002, p. 257) model of pro-environmental behavior.

ment with different contexts (Battilana et al., 2009; Hoek et al., 2021), but also for the complexity of sustainability transformation (Abson et al., 2017). It thus offers a more comprehensive view than previous models such as the Theory of Planned Behavior (Ajzen, 1991), the Value-Belief-Norm Model (Stern, 2000), the Value Identity Personal Norm Model (van der Werff and Steg, 2016) or mergers thereof (Ates, 2020). Fig. 1 shows Kollmuss and Agyeman's (2002) original model, which indicates how pro-environmental behavior is influenced by the interaction of different factors (gray arrows) and which barriers impede positive influence on pro-environmental behavior (black boxes). Internal factors include actors' "complex of pro-environmental conscious-

#### Table 2

Previous research applying or building on the Kollmus-Agyeman model.

Author(s)	Study focus	Reference to model	Contribution to the model
Siegel et al. (2018)	(Re)Storying Pro-Environmental Behavior	Model development	Considering the complex, varied, and interconnected influences: Intertwining External and Internal Factors of Influence
Stoll-Kleemann and Schmidt (2017)	Barriers and opportunities for reducing meat consumption	Model application	Incorporation of further factors: Habits and taste
Stoll-Kleemann (2019)	Barriers and opportunities for behavior change towards ocean literacy	Model application	Incorporation of further factors: Habits and comfort
Uusi-Rauva and	Barriers to environmental	Theoretical review strongly	Development of a multilevel model of barriers to individual
Heikkurinen (2013)	advocacy campaigns in organizations	informed by the model	environment behavior in the organizational context. Addition of the organizational context to individual and societal context
Latif et al. (2013)	Role of environmental knowledge in creating pro-environmental residents	Theoretical review strongly informed by the model	Empirical confirmation of the model: No direct relationship between environmental knowledge and pro-environmental behavior. Environmental value as mediator.
Byers and Gilmer (2018)	Unified approach to sustainable consumption behavior	Synthesis of research on understanding and promoting pro-environmental behavior	Synthesis of Kollmuss and Agyeman (2002), Fuchs (2017) and Phipps et al. (2013) to a conceptual framework on how consumption behavior can be developed and changed. Pro-environmental behavior is understood as nested within societal subsystems.
(Petri and Faust, 2021)	Understanding permaculturist motivations among residents	Synthesis of behavioral research	Combination of various determinants to construct a whole picture of behavior patterns

ness", i.e. their knowledge of sustainable solutions, emotional and cognitive involvement and pro-environmental values, as well as their "locus of control", i.e. their perceived ability to bring about change through their own actions (Kollmuss and Agyeman, 2002; Newhouse, 1991). Internal factors are in interactions with each other and increase the complexity of behavioral change. External factors include socio-cultural norms and standards, the industry infrastructure, economic conditions and the regulatory environment. To answer the key research question, this study focuses on how internal and external factors influence behavior negatively, thus providing a better understanding of why meat production practices have not transformed considerably, so far.

Second, an adoption of the model to the meat production context can complement previous applications of the model and provide new insights that may confirm or extend the model itself or previous theoretical extensions. Table 2 lists studies that have applied and dealt with the model previously.

Stoll-Kleemann and Schmidt (2017) applied the model to the context of meat consumption, showing that meat consumer behavior is particularly influenced by internal factors, including emotions, cognitive dissonance and socio-cultural factors like social identity. This finding is in line with sustainability transformation research, which emphasizes the relevance of "deep" personal factors for sustainability transformation (Abson et al., 2017; Ives et al., 2020). Uusi-Rauva and Heikkurinen (2013), however, suggest that in the organizational context, external factors may become much more prevalent, as personal factors are deeply embedded in and influenced by organizational and societal factors. This corresponds with the results of a recent review by Hoek et al. (2021), who put particular emphasis on the external context. Taking the idea of embeddedness further, Siegel et al. (2018) as well as Byers and Gilmer (2018) propose a complex interconnectedness among and between internal and external factors. To date, however, this complex interconnectedness, particularly between external factors, has not been incorporated in the Kollmus-Agyeman model. This may be due to the fact that it has never before been applied to the context of industry actors. Some applications of the model (Stoll-Kleemann and Schmidt, 2017; Stoll-Kleemann, 2019) added new influencing factors, including taste and comfort, suggesting barriers based on positive emotions or connotations with the status quo. In the same vain, path dependence research (Vergne and Durand, 2011) suggests that reluctance to change not only stems from negative mechanisms rendering alternative paths less attractive, but also from positive mechanisms supporting the current path.

The review of the existing research literature reveals that the application of the Kollmus-Agyeman model to the context of meat *producers* could both provide complementary insights into factors influencing behavior change towards sustainability, as well as extend the model to the analysis of barriers as perceived by producers.

# 3. Method

## 3.1. Design and empirical setting

A qualitative research approach has been deemed suitable for research analyzing complexities of sustainability transformation (Strauss, 1987; Luederitz et al., 2017) and barriers to sustainable production practices (for the supply chain see, e.g., Sajjad et al., 2020; for the fashion industry see, e.g., Sirilertsuwan et al., 2019). This qualitative analysis investigates the meat industry of Germany as a relevant empirical case, since previous publications have pointed to slow sustainability-oriented change in German meat production, despite increasing social pressures and the existence of various different sustainability solutions (Chemnitz and Wenz, 2021; Chemnitz et al., 2018; Clausen and Mathes, 2017).

### 3.2. Sampling

Through heterogeneous and snowball sampling, 23 participants from the main meat production phases of rearing, fattening, slaughtering, processing, marketing and retailing were selected (Savin-Baden and Major, 2013). The sample was mainly collected in Northwest Germany (Fig. 2), which is the region with the highest livestock density in Germany (Fig. 3). The participants cover different phases of production and sizes of operation. The analysis focuses on meat *production* and therefore does neither consider consumers nor gastronomical facilities and private households.

#### 3.3. Data collection

Data was collected with 23 semi-structured interviews (Table 3) and from participant observation. The semi-structured interviews provide insights into the respondents' "opinions and beliefs" (Easterby-Smith et al., 2012), which was deemed particularly important for the identification of perceived internal barriers. The interview script included questions regarding work practices, internal

Table 3		
Overview	of	interviews.

Interviewee(s)		Meat production chain phase(s)
F1	Farmer (owner)	Rearing
F2 & F3	Two Farmers (owners)	Rearing –Fattening
F4	Farmer (owner)	Fattening
F5	Farmer(owner)	Fattening
F6	Farmer (owner)	Fattening
LT 1	Livestock trader (managing director)	Trade & Transport
LT 2	Livestock trader (managing director)	Trade & Transport
S1	Slaughterhouse manager	Slaughtering – processing
S2	Small slaughterhouse operator (owner)	Slaughtering – processing – marketing
P1	Processing plant manager (officer for agriculture and animal welfare)	Processing
B1	Butcher (owner)	Processing – marketing
B2	Butcher (owner)	Processing – marketing
B3	Butcher (owner)	Processing – marketing
B4 & B5	Butcher association manager (managing director) & butcher (owner)	Processing – marketing
PA1	Producer association manager (managing director)	(Fattening) – processing – marketing
MC1	Meat company manager (communication manager)	Processing – marketing
MC2	Meat company manager (supply chain manager)	Processing – marketing
MC3	Meat company manager (senior buyer)	Processing – marketing
MC4	Meat company manger (product manager)	Vertically integrated
MM1	Meat marketing platform manager (managing director)	Marketing
R1	Retail manager (managing director of a supermarket)	Retail
R2	Retail manager (CSR manager of a discounter)	Retail
E1	Meat industry (chief editor of industry journal)	Support function



Fig. 2. Geographical distribution of interviews.

as well as external factors (see Appendix A for the interview script used in the research).

All interviews were conducted with actors directly involved in the production, marketing, and sales of meat, with the exception of one interview conducted with the chief editor of a leading meat industry journal. The number of interviews was guided by data saturation (Glaser and Strauss, 1967). All interviews but one were conducted at the locations of industry actors or via telephone between July 2018 and October 2018, lasted on average about 45 min and were subsequently transcribed. For one interview, answers were provided in written form. In six cases, data was additionally



Fig. 3. Areas in Germany with high livestock density (Thünen-Institut/Statistisches Bundesamt, 2016).

collected in the form of observations during site visits (1 rearing farm, 3 fattening farms, 1 butchery and 1 livestock trader). Observation helped to gain more comprehensive insights and served to triangulate findings (Yin, 2009). Data from observation was protocolled (Babbie, 2013). To secure participants' anonymity, all data is presented here without participant or company names (Gioia et al., 2013).

## 3.4. Data analysis

Data was analyzed deductively, in light of internal and external factors as defined by the Kollmuss-Agyeman model. A combination of a-priori coding (through a code template) and pattern coding (emergent from the data) resulted in the identifi-

#### Table 4

Results of the data analysis.

Influence factors (categories)		Barriers for meat industry actors (2nd order themes)		
Internal	Consciousness complex (knowledge, emotional & cognitive reactions, values & attitudes)	A status quo bias (Hermann et al., 2016;(Engler et al., 2019), i.e. cognitive attachment to business as usual, impedes reflections about its negative impacts <i>Cognitive dissonance</i> (Stoll-Kleemann and Schmidt, 2017; Festinger, 1957), i.e. a value-behavior gap, which results in denial and delegation of responsibility		
		Disconnects (lves et al., 2018; Pachirat, 2011) impede cognitive and emotional involvement in the impacts of industrial meat production		
Apath to ina		Apathy (Uusi-Rauva and Heikkurinen, 2013) results from fear and frustration about contextual conditions; leads to inaction concerning sustainability		
Emotional commitment reinforces posi knowledge		Emotional commitment reinforces positive attitudes towards conventional meat; blocks sustainability-relevant knowledge		
Locus of control Power asymmetries (Virah-Sawmy e to drive change and perceived cont		Power asymmetries (Virah-Sawmy et al., 2019; Belaya et al., 2009; Talay et al., 2018) reduce the perceived ability to drive change and perceived control over the success of initiatives		
External Norms & standards A conventi continued		A conventional meat paradigm (Gwin, 2009; Dagevos and Voordouw, 2013) guides behavior in favor of the continued production of conventional meat		
Industry-orie		Industry-oriented standards (Klitkou et al., 2015) complicate shifts to small-scale production		
Industry infrastructure Industry specialization and synergies (Klitk and marketing conventional meat		and marketing conventional meat		
	Economic conditions	Market (dis)incentives (Guenther et al., 2013; Gupta et al., 2020) arising from the global market for conventional meat hamper shifts to ecological products		
		Economies of scale (Klitkou et al., 2015) foster the adherence to paradigms of growth, expansion, specialization		
		and concentration Sunk investments (Cupta et al. 2020: David 1985) and negative financial feedback loops hamper investmen		
		into sustainability solutions		
	Regulatory environment	Regulatory (dis-)incentives (Laurett et al., 2021;(De Jesus and Mendonça, 2018) create framework conditions to the detriment of small-scale meat production		

cation of 13 second-order themes and the allocation to respective influence factor categories (Fereday and Muir-Cochrane, 2006; Saldaña, 2015). By alternating between the emergent second-order themes and previously reviewed literature, themes were linked, where possible, to previous research on barriers to sustainable production practices (Gioia et al., 2013). For example, this analysis found cognitive dissonance to be an important barrier for meat producers, mirroring Stoll-Kleemann and Schmidt's (2017) findings on meat consumption behavior. Further, the data revealed a status quo bias among the interviewees, tying in with a previous investigation of hog farmers by Hermann et al. (2016). As another example, the data pointed to barriers attributable to economies of scale – a barrier which has previously been discussed in the context of energy production by Klitkou et al. (2015). While most second-order themes could be linked to precedents in literature, one theme emotional commitment - is a new concept derived from the data (Gioia et al., 2013). Table 4 (see Section 4) provides an overview of the 13 second-order themes and their link to previous literature on barriers to sustainable production as well as associated influence factor categories of the Kollmuss-Agyeman model. Measures employed to achieve validity include peer scrutiny, reflective commentary as well as the involvement of more than one data source and researcher (Shenton, 2004). Data analysis was validated in the context of a workshop entitled "The changing meat industry – paths and approaches of a sustainability transformation" in March 2019, to which all study participants were invited.

#### 4. Results

This section outlines key findings from the conducted interviews and observations. Table 4 provides an overview of results of the data analysis, including the identified second-order themes and influence factor categories. It shows how different internal and external factors influence industry actors' meat production behavior and create multiple barriers to a sustainability transformation of production practices. The following subsections are structured according to the results of the data analysis, including internal barriers (Section 4.1) and external barriers (Section 4.2). Furthermore, barrier interactions arising from the data are outlined (Section 4.3).

## 4.1. Internal barriers

#### 4.1.1. Consciousness complex

The data reveals that the consciousness complex, particularly cognitive and emotional reactions, influence industry actors' behavior with regard to meat production practices. It frames the thinking of meat industry actors in terms of the adequacy and importance of specialization (F1; E1), intensive livestock farming (F5) and conventional meat products (B3). Here, actors engage in "shifting baselines" (Pauly, 1995, p. 430), meaning that they use a selective, considerably recent past in the frame of the same conventional operations as a baseline for justifying the current system. Two interviewees, for instance, stated that "animals are now kept better than 30 years ago" (LT2) and that "40 years ago, the sows had still been chained" (F5). Against this background, a processing plant manager perceives a 10 percent increase in space a sufficient success for animal welfare (P1). The data further shows that actors experience cognitive dissonance, i.e. moments where their behavior does not match their values (Festinger, 1957). For instance, even though some interviewees openly question the reasonableness of industrial meat production (MC3; R1), they overcome the dissonance by denying their own responsibility and instead blaming consumers (MC2), politics (MC4) or other industry members for slow transformational change: "The barriers are not with us, but are clearly with the farmers" (B4). In sum, the interviews reveal that cognitive awareness of the negative impacts of industrial meat production is suppressed by a status quo bias that "the current state of things is locked-in as a mental reference point" (Engler et al., 2019), p. 610).

The data further shows that industrial meat production fosters various *disconnects* (Ives et al., 2018), which impede actors from becoming cognitively and emotionally involved with the negative impacts of production practices. It is particularly apparent on industrial livestock farms, where farmers perceive the architectural design and short fattening periods as impeding factors for the development of personal relations with the animals (Observation; F2; F3). Here, farmers point to the fact that they "rarely touch the animal. It is led, kept and herded into other sheds" (F2). Different interviewees further point to highly anonymized and spatially separated supply chains, which lead to disconnects between produc-

ers (S1; MM1) in the sense that different members of the meat industry "have lost the understanding for each other" (B2). The geographical distance of production and slaughter facilities (Observation S1; E1) as well as "no-trespassing"-signs on livestock farms (Observation; F5) additionally contribute to a consumer-producerdisconnect (Pachirat, 2011). As a consequence, one of the interviewed farmers experiences that "agriculture fulminates against consumers, consumers fulminate against agriculture" (F5), where nobody feels responsible and enabled to change. This circumstance, in turn, results in growing frustration amongst industry actors over their "very difficult situation" caused by increasing legislation (F6), the "unfair" treatment by the media (F5) and the lack of appreciation from the public for their (sustainability) efforts (LT2; F2; F3). These perceptions can, in the worst case, lead to apathy about the likelihood and feasibility of sustainability measures (informed by "sense of fatalism" as outlined by Uusi-Rauva and Heikkurinen, 2013). Some "don't know how long they will still have the strength" to make efforts "nobody appreciates" (F5). Here, interviewees also point to the perceived injustice of urging meat industry actors to change while other areas of production and consumption are perceived to remain unchallenged (F2; F3; F6). Others are "paralyzed" by existential fears (F1; F5; S2; B1). As one butcher comments, the situation "has gotten so deadlocked that there is just not the courage to fundamentally change" (B1).

However, not only negative, but also positive emotional reactions create barriers to changes in meat production practices. In many cases, the interviews show that meat industry actors' attitudes, values and resulting emotional commitment to their work in livestock farming and butcher trade have been shaped by longstanding family traditions (F1; F2; F3; B2; B3; B5). In turn, these practices "is exactly what [they] want to pass on to [their] children" (B2). The fact that most farmers and butchers own their operations themselves intensifies the commitment and further strengthens respective professional identities (Zellweger and Astrachan, 2008). Due to past efforts, industry actors feel that they have earned the right to secure their achievements, as one of the farmers passionately resumes: "We fight for it now [...] and we will carry on" (F5). Not only farmers and butchers, but also managers of meat companies are proud of their meat products, referring to them as "top quality" (B3) "exceptional" (B4; B5), "natural" (MC2) and "honest and clean" (MC3). Interviewees find meat to be "very important foodstuff" (B2; B3; B4), which should be available for everybody and, therefore, cheap (F1). Accordingly, they build and maintain specialized knowledge in conventional meat production and only few engage with strategies for fundamental industry change (P1; MC2). For instance, many reject alternatives such as plant-based substitutes and cell-based meat as "unnatural" (MC2), "dubious" (F6), "out of the question" (B1) and "a long way off" (P1). Self-reinforcing commitment creates a barrier to sustainability transformation that could make some professional titles and job descriptions obsolete.

#### 4.1.2. Locus of control

The data shows that *prevailing power asymmetries* along the meat production chain influence meat industry actors' perceptions of how much sustainability-oriented change is considered possible. Interviewees identify industrial slaughterhouses and retailers as actors in the meat production chain exerting power over others and upholding the status quo (F2; F3; S1; MC2; E1). On the one hand, this unequal distribution of power leads to the reduced perceived ability amongst many actors to drive transformation. While farmers feel obliged by industrial slaughterhouses to adhere to conventional specifications for animal mass and size (F2; F3), meat producers feel pressured by retailers to fulfill conventional meat product requirements in terms of shelf life, ingredients, product characteristics and trends (F1; MC2). On the other hand, the unequal dis-

tribution of power, coupled with the interlacement of production chain stages, leads to a reduced perceived control over the success of sustainability initiatives. Farmers, for instance, feel that a shift to organic or agro-ecological livestock production crucially depends on actions by retailers (F1; F6). As one farmer concludes: "If in retail nobody can communicate what we produce, then the topic is dead" (F1).

# 4.2. External barriers

#### 4.2.1. Norms and standards

Cultural norms and industry standards create further barriers for meat industry actors to change their production practices. Concerning the former, industry actors' activities are embedded in a "meat-friendly" national culture (Mellinger, 2000), defined by a conventional meat paradigm (term informed by "meatcentric paradigm" as used by Dagevos and Voordouw, 2013). Industry actors themselves also actively contribute to upholding this paradigm. For instance, the advertisement-induced romanticizing of conventional meat production through large meat producing companies continually strengthens socio-cultural preferences for (cheap) conventional meat (Gossard and York, 2003): "For a long time, we have advertised green fields, but in reality, it wasn't like that [...] For a long time, we have advertised healthy meat, but it was just conventional meat." (LT1). According to the senior buyer of a large meat company, this messaging has successfully established a positive image of conventional meat brands in society, the legitimacy of which is constantly reinforced (MC3). The conventional meat paradigm with images and beliefs of meat being inherently a healthy and natural product creates barriers to change towards extensive, agro-ecological meat production or towards the production of alternative protein sources.

Concerning the latter, industry actors are very much aware that *industry standards* in terms of animal breeds, fattening periods, shed structures, transport, slaughtering and processing are aligned to the industrialized mass production of meat (Observations F2; F3; F5; LT1; S2). They also refer to the highly regulated and standardized documentation, relating to hygiene, product sampling and payroll accounting (S2; B4; B5; E1). Adopting the established standards allows highly industrialized and automated facilities to work in a cost and time efficient way (S1). Small-scale slaughter facilities and butcheries, who can hardly meet the vast number of requirements and cost-efficient quantity of production, gradually disappear (S2; B1; B3; B5). As a consequence, the diversity of actors and approaches to meat production that could provide a basis for different paths of development and change, is reduced.

#### 4.2.2. Industry infrastructure

Industry specialization and synergies reinforce the existing industrial structures for the production and marketing of conventional meat. The highly specialized supply chains and infrastructure, with production stages fitting to each other, allow industry actors to work very efficiently (S1; B2). At the same time, interviewees note that the established network of professional intermediaries like producer associations and livestock traders increase the anonymity of business relations and makes them easily exchangeable (F2; F3; LT2; S1; B4). For the manager of an industrial slaughterhouse, "it is supply and demand and business relationships result from that." (S1). Any transformation towards sustainability that requires changing interactions in the supply chain can therefore not be realized by one actor alone but requires many actors to agree on the same or similar change and to act in a coordinated manner. As the infrastructure of each actor is highly specialized, too, investments and time would be needed, including coordination of who changes when. One meat company manager outlines the difficulty regarding the introduction of organic meat products:

"We as [Meat company X] have not even access to the farmers. This means that I always have to contact the slaughterhouse first to get in contact with the farmers. And then, the slaughterhouse and [Meat company X] have to convince the farmer: we'd like to do it like this, would you join? But the farmer will only join if he gets more money." (MC3)

#### 4.2.3. Economic conditions

The interviews reveal that barriers linked to economic factors include both market and organizational incentives and disincentives. The availability of cheap animal feed from abroad for conventional animal rearing (PA1) and attractive export opportunities (S1) create market incentives for big meat producers to continue business as usual. Interviewees agree that incentives to sell meat from conventionally raised animals are high, as there is a well-functioning global system, in which, for instance, German demand for fine cuts (e.g. filet) is complemented by Asian demand for "residual" parts (e.g. feet, ears and tails) (F5; S1; B4; B5; E1). Thus, advantageous complementary marketing opportunities in different markets create barriers to transformational change for German meat industry actors who actively participate in the global market (S1). At the same time that global demand for conventional meat products is high, local organic or agro-ecological meat remains in market niches due to constantly low consumer demand and the lack of a globalized market for such products (F1; PA1; MM1; R1). This circumstance disincentivizes meat industry actors to consider organic or agro-ecological products as viable alternatives (MC3; MC4).

Concerning organizational barriers, through economies of scale (David, 1985) meat industry actors feel encouraged to adhere to and expand conventional production (LT1; MC1; MC3). Through specialization, automation and large-scale acquisition activities, big industry actors like international meat companies and industrial slaughterhouses continue to benefit from decreasing costs (F1; LT2; S1; MC1). With increasing global competition, "the market will push [them] to grow even bigger" (LT2). Consequently, achieving a high degree of concentration, consolidating cost leadership and making continuous investments is an essential factor for business success (S1; P1; MC1; MC4), or even for survival in the market (LT2). As the interviewed industry actors are fully aware of this situation, this barrier impairs change that reduces specialization or requires different kinds of specialization leading to cost increase of less (fast) cost reduction than for those who further optimize their specialization role in the industrial system. For other meat industry actors, especially farmers and smaller meat marketers, sunk investments and high uncertainties about the return on investment for sustainable alternatives create barriers to transformation (F4; F5; F6; S2). Initial investments also captivate industry actors in negative feedback loops. Here, a pig fattening farmer points to the high investments into animal sheds, which are usually amortized over more or less 25 years and constitute long-term liabilities (F6).

#### 4.2.4. Regulatory environment

On the one hand, the data shows that powerful actors benefit from *regulatory incentives* such as subsidies for processing and transportation of meat that create a reluctance to transformational change (S1; B3). Financial incentives for incremental improvements (e.g. animal welfare initiatives) at times also impede more transformative action (F1; F6). On the other hand, the data highlights *regulatory disincentives* such as conflicting policies and excessive bureaucratic requirements that create transformation barriers. Concerning the former, the product manager of a meat company points to regulatory contradictions with regard to shed adjustments for animal welfare improvements:

"Since the building permit is accompanied by a change of emissions limits, it [the shed extension] is not approved by the authorities. So, on the one hand the animal welfare authorities say "do it" and on the other hand the building authorities prevent it." (MC4)

Further, the multitude of different sustainability standards and labels causes confusion among producers, which, at worst, leads to inaction through overload, uncertainty and irritation (F5; P1; R1; R2). Concerning excessive bureaucratic requirements, interviewees referred to documentation, quality control and conversion procedures (e.g. to organic farming), which are implied in sustainability measures and, therefore, render them unattractive (F6; S2; MC3; PA1; MM1). The influential power of some industrially operating large-scale actors and the power imbalance between large and small operators leads to a constant reinforcement of legal requirements, which restrict possible transformative change (F1; P1).

### 4.3. Barrier interactions

The data reveals multiple interactions among and between internal and external barriers. As an example for interactions among internal barriers, *emotional commitment* to their operations can cause a *status quo bias* among some of the interviewed farmers and butchers, which impedes them from engaging with alternative solutions (F1; F2; F5). As an example for interactions among external barriers, *market (dis)incentives* lead to highly specialized and largescale operations, encouraging farmers, slaughterhouse managers and meat company managers to adhere to and uphold *industryoriented standards* (F1; S1; MC1). As an example for interactions between internal and external barriers, *economies of scale* can result in internal barriers such as *cognitive disconnects* (F2; F3). Cognitive disconnects, in turn, hamper reflections about the impacts of industrial meat production leading to inaction, thereby upholding the *conventional meat paradigm* (LT2).

# 5. Discussion: the complexity of transforming meat production practices

The qualitative empirical analysis of barriers for a sustainability transformation of meat production practices reveals a high degree of complexity with multiple obstacles, which meat industry actors perceive. Identifying them is a first step to develop approaches for overcoming them and for sustainability transformation (Guenther et al., 2013).

# 5.1. Barrier complexity and model extension

Based on the Kollmuss-Agyeman model, Fig. 4 provides an overview of the findings. The complexity of barriers originates from multiple interactions among and between internal and external factors (left column), leading to the simultaneous existence of barriers characterized by reinforcement and restraint (second column). Reinforcement barriers encourage industry actors to continue pursuing established practices and make them perceive transformation as something undesirable (plus signs in second column). At the same time, restraint barriers hinder industry actors from pursuing sustainable alternatives and make them perceive transformation as something rather unfeasible (minus signs in second column). The on-going mutual interaction of different barriers results in a continuation of business as usual and, ultimately, in the lack of a sustainability transformation of meat production practices (right column). In the following, factor interactions and resulting barrier types displayed in the two columns of Fig. 4 are explained further and linked to previous research.

**Factor interactions** – The analysis reveals the existence of interactions among and between different internal and external factors, which create mutually enhancing barriers. Factor interactions



Fig. 4. Barriers to a sustainability transformation of meat production practices (based on the Kollmuss-Agyeman model).

are shown by gray arrows in left column of Fig. 4. While this finding is in line with research on barriers to sustainable production systems (Hoek et al., 2021; Klitkou et al., 2015; Luthra et al., 2014), it contributes to previous research on barriers to sustainable meat production practices, where barrier interactions have not been considered so far (e.g. Läpple and Kelley, 2013; Gwin, 2009; Hermann et al., 2016). In addition, this study provides further insights into the particularities of industry actors' external barriers and barrier interactions that result in an extension of the Kollmuss-Agyeman model. First, linking to earlier suggestions by Hoek et al. (2021) and Uusi-Rauva and Heikkurinen (2013), this study shows just how much meat industry actors' behaviors are shaped by the external context. The interviewed meat industry actors perceived strong influences from the organizational, industry, economic and regulatory context that jointly create barriers to sustainability-oriented changes in meat production practices. Seeing organizational, economic and regulatory influencing factors on par with internal factors diverts from Stoll-Kleemann and Schmidt's (2017) findings for the context of meat consumption, which emphasized the dominance of internal and socio-cultural factors. Second, this study extends the Kollmuss-Agyeman model by conceptualizing and visualizing external barrier interactions (Siegel et al., 2018; Byers and Gilmer, 2018). The results show that an application in the context of producers – as a difference to consumers, residents and individuals – asks for a new emphasis on external factors. The extension complements Kollmuss and Agyeman's (2002) conceptualization of interactions among internal factors and unveils the previous underrepresentation of external factors. At the same time, it considers the system interconnectedness, which has been suggested, but not yet specified, by Siegel et al. (2018) and Byers and Gilmer (2018).

**Reinforcement and restraint barriers** – The findings reveal that the barriers created by mutually interacting internal and external factors can be differentiated into two types of barriers: Barriers characterized by reinforcement that uphold the status quo and barriers characterized by restraint that render alternatives unattractive or unfeasible. The bracketed plus and minus signs in the second column of Fig. 4 differentiate such reinforcement and restraint barriers. This additional barrier complexity extends the Kollmuss- Agyeman model and model adoptions in other contexts (e.g. Stoll-Kleemann and Schmidt, 2017). While adding a previously unconsidered aspect to Kollmuss and Agyeman's work, this finding relates to existing research on path dependence mechanisms (Vergne and Durand, 2011). This study's findings propose

that experiences of reinforcement or restraint vary for different actors along the meat production chain. The interviews reveal that while big powerful industry actors are reluctant to pursue transformation because they are incentivized to continue business as usual (reinforcement barriers), smaller industry actors are rather trapped in negative feedback loops that render change unfeasible (restraint barriers). This finding extends Virah-Sawmy et al.'s (2019) argument that companies along the soy-meat value chain experience sustainability challenges and associated risks quite differently. The differentiation of barrier types and the specification of barrier complexity contributes to previous research on barriers to sustainable production practices (Laurett et al., 2021; Hoek et al., 2021; Gupta et al., 2020; Gwin, 2009; Rodriguez et al., 2009). The question, however, remains, where within this complex web of barriers starting points for sustainability transformation could lie.

#### 5.2. Opportunities for overcoming barriers in today's context

To create transparency about the web of interacting barriers is a first important step for identifying starting points for industry and food system transformation (Hoek et al., 2021). While Kollmuss and Agyeman (2002) suggest that transformative behavior change relies on the combined effect of internal and external factors, sustainability transformation research (Abson et al., 2017; Ives et al., 2020; (O'Brien and Sygna, 2013) ascribes particular importance to "deep" internal factors, including knowledge, mindsets and values, for achieving change towards sustainability. This corresponds with recent findings that behavior changes in general (Ates, 2020), and changes of agricultural practices in particular (Govindharaj et al., 2021) are crucially dependent on people's knowledge, self-identity, attitudes as well as subjective and moral norms. Through changes at the personal level, actors are enabled to engage in approaches such as sustainable entrepreneurship, which has been discussed in literature as an important approach for achieving sustainability transformations of markets (Johnson and Schaltegger, 2020). To foster change at the personal level, however, external triggers and nudges may be needed (Byerly et al., 2018).

In this regard, previous research has shown that external global shocks can positively influence pro-environmental behavior (Ivlevs, 2019). The scandal 'Dieselgate' for the automotive industry (e.g. Brand, 2016) or the current global Covid pandemic are examples for how external shocks with disruptive impacts across production chains can accelerate shifts in industry actors' mindsets and, consequently, practices. The pandemic has raised discussions about the fragility of our food systems and the need for resilience in food supply chains (Boyacı-Gündüz et al., 2021; Hobbs, 2021). Discussions have particularly focused on the meat industry, whose practices in slaughtering and processing have been severely disrupted. As a consequence, awareness is growing for the need to accelerate the transformation towards alternative meat production systems, particularly alternative proteins (Anomaly, 2020) and/or agroecological livestock production (Perrin and Martin, 2021). In the US, for instance, the Covid pandemic encouraged producers to slow growth rates (Marchant-Forde and Boyle, 2020). In Germany, the slaughterhouse scandals during the Covid pandemic hampered the consumers' image of the meat industry and at the same time accelerated the producers' exploration of the alternative protein sector (Chemnitz and Wenz, 2021). The question is, however, whether producers will continue to enforce changes beyond the pandemic.

# 6. Conclusion

Applying the Kollmuss-Agyeman model, this study finds that meat industry actors perceive multiple barriers for a sustainability transformation of meat production practices, and that the interactions among and between internal and external factors create a complex and reinforcing web of barriers. In doing so, the conducted analysis contributes to previous research on sustainable production practices in general and sustainable meat production practices in particular, and in addition, extends the Kollmuss-Agyeman model. Based on the insights of sustainability transformation research (Abson et al., 2017; Ives et al., 2020), this study suggests a temporal hierarchy of behavior changes by identifying change of internal factors, including knowledge, mindsets and values, as crucial starting points for transformation. These "deep" factors of change can be stimulated by, for instance, external triggers, which encourage actors to reflect upon their current practices and to pursue sustainability more fundamentally. The analysis comes with some limitations and has implications for both research and practice, which will be detailed in the following.

## 6.1. Limitations

As all interviews were conducted with actors in the German meat industry, limitations arise with regard to the geographical scope of this analysis. The study's results may not be representative for other countries or industries. While various conditional factors may be comparable in the EU, this may be different for Non-EU countries and other continents. For instance, economic barriers arising from market incentives might be even higher in countries like the U.S. or Brazil, which are the biggest exporters of meat worldwide. At the same time, socio-cultural barriers might be perceived as lower in countries like India or Bangladesh, where meat consumption is not as firmly embedded in the national culture. Indeed, a recent study found that markets in India possess particularly high potential for plant-based and cultivated meat alternatives (Bryant et al., 2019). Analysis results might also differ for producers in other industries. Since meat production operations are often embedded in family traditions, producers experience particularly high emotional barriers. In contrast, producers in the automotive, oil or gas industry, might experience less emotional barriers, but higher techno-economic barriers due to the complexity of shifts to renewable energy innovations.

#### 6.2. Implications for research

Considering the identified limitations, this study provides a basis for further applications of the Kollmuss-Agyeman model to producers and respective investigations of barriers to sustainable production practices in different national contexts and industries. In this regard, a quantitative analysis of barriers could also be dealt with. Further, this analysis indicated differences in barrier perceptions resulting from diverging positions in power. It may be a valuable future research avenue to identify strategies for harnessing existing positions of power for advancing sustainability transformation and opportunities for empowering (seemingly) marginalized industry actors. In doing so, participatory approaches could allow for the integration of practical knowledge into scholarly work. Practitioners often "develop a deep understanding of the problems and tasks that arise in particular situations" (Van de Ven, 2007, p. 4) and can therefore provide relevant contributions to problem identification and solution development (e.g. Lang et al., 2012). Finally, future research could examine, which types of barriers dominate when seeking different sustainability approaches (i.e. improvements in animal welfare, shifts to extensive production systems or the substitution of animal-based products by alternative proteins).

## 6.3. Implications for practice

In terms of practical implications, this study is the first to offer meat industry actors a comprehensive picture of their complex barrier situation, particularly with regard to relevant contextual conditions as well as interactions between internal and external barriers. Many barriers are related to the high complexity of the current meat supply chain, which leads to spatial and cognitive disconnects resulting from highly specialized operations amongst actors. Overcoming these barriers and achieving transformation beyond the boundaries of individual organizations therefore requires a high level of cooperation throughout the production chain. Industry actors are encouraged to reflect upon their personal scope of action and engage in collaboration for sustainable production practices. The analysis furthermore offers political decisionmakers a framework to better understand interactions between legislation, standards, infrastructure and socio-cultural norms. For instance, understanding the link between contradictory legislation, actors' inaction and the maintenance of norms and standards of industrial meat production might lead to multi-level initiatives for encouraging sustainable practices as well as focused, more effective legislation. Particularly the disclosure of the difficulties smallscale operations face might initiate change towards less complex regulations supporting sustainability transformation. As the interviews indicate, insufficient incentives for more sustainable production practices, targeted incentives are needed to establish advanced animal welfare programs as well as to diversify and regionalize agricultural practices. Since many industry actors seem to be unaware of more sustainable options such as the production of alternative protein products, policy initiatives that integrate knowledge from different research domains and practice are needed to provide actors with relevant information on sustainable alternatives.

# **Declaration of Competing Interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### **Appendix A. Interview Script**

## **Opening the interview**

Introduction of researcher and research context – Presentation of interview contents – Information on confidentiality and anonymity – Signing of agreement to participate in the research

# Interview Part 1: Company development and work practices

- (1) Historical development of operation or company (built expertise, family tradition ...)
- (2) Work practices and work environment of interviewee
- (3) Work routines with other industry actors along the supply chain
- (4) Changes in work environment and related challenges

#### **Interview Part 2: External factors**

- (1) External factors influencing work operations (regulations, consumer demand ...)
- (2) Current developments and trends in industry and society
- (3) Strategies to respond to new developments/trends and encountered challenges

#### **Interview Part 3: Internal factors**

- Awareness of and opinion on sustainable production practices (e.g. extensive organic meat production & production of alternative protein products)
- (2) Perceived role/agency of interviewee in advancing sustainable production practices

(3) Personal estimate of the future development of meat production

# Ending the interview

Thanking the interviewee for his/her participation – Opportunity for the interviewee to add statements – Explanation of the further procedure

#### References

- Ajzen, I., 1991. The theory of planned behavior. Organ. Behav. Hum. Decis. Process. 50 (2), 179–211. doi:10.1016/0749-5978(91)90020-T.
- Abson, D.J., Fischer, J., Leventon, J., Newig, J., Schomerus, T., Vismaier, U., von Wehrden, H., Abernethy, P., Ives, C.D., Jager, N.W., Lang, D.J., 2017. Leverage points for sustainability transformation. Ambio 46 (1), 30–39. doi:10.1007/ s13280-016-0800-y.
- Allievi, F., Vinnari, M., Luukkanen, J., 2015. Meat consumption and production: analysis of efficiency, sufficiency and consistency of global trends. J. Clean. Prod. 92, 142–151. doi:10.1016/j.jclepro.2014.12.075.
- Anomaly, J., 2020. Cultured meat would prevent the next Covid crisis. Anim. Sentien. 5 (30), 5. doi:10.51291/2377-7478.1633.
- Ateş, H., 2020. Merging theory of planned behavior and value identity personal norm model to explain pro-environmental behaviors. Sustain/ Prod. Consum. 24, 169–180. doi:10.1016/j.spc.2020.07.006.
- Babbie, E.R., 2013. The Practice of Social Research, 13th ed. Wadsworth Cengage Learning.
- Battilana, J., Leca, B., Boxenbaum, E., 2009. How actors change institutions: towards a theory of institutional entrepreneurship. Acad. Manage. Annals 3 (1), 65–107. doi:10.5465/19416520903053598.
- Belaya, V., Gagalyuk, T., Hanf, J., 2009. Measuring asymmetrical power distribution in supply chain networks: what is the appropriate method? J. Relat. Mark. 8 (2), 165–193. doi:10.1080/15332660902876927.
- Bhat, Z.F., Kumar, S., Bhat, H.F., 2017. In vitro meat. A future animal-free harvest. Crit. Rev. Food Sci. Nutr. 57 (4), 782–789. doi:10.1080/10408398.2014.924899.
- Boyacι-Gündüz, C.P., Ibrahim, S.A., Wei, O.C., Galanakis, C.M., 2021. Transformation of the Food Sector: security and Resilience during the COVID-19 Pandemic. Foods 10 (3), 497. doi:10.3390/foods10030497.
- Brand, C., 2016. Beyond 'Dieselgate': implications of unaccounted and future air pollutant emissions and energy use for cars in the United Kingdom. Energy Policy 97, 1–12. doi:10.1016/j.enpol.2016.06.036.
- Bryant, C., Szejda, K., Parekh, N., Deshpande, V., Tse, B., 2019. A survey of consumer perceptions of plant-based and clean meat in the USA, India, and China. Front. Sustain. Food Syst. 3, 11. doi:10.3389/fsufs.2019.00011.
- ... & Byerly, H., Balmford, A., Ferraro, P.J., Hammond Wagner, C., Palchak, E., Polasky, S., Fisher, B., 2018. Nudging pro-environmental behavior: evidence and opportunities. Front. Ecol. Environ. 16 (3), 159–168. doi:10.1002/fee.1777.
- Byers, V., Gilmer, A., 2018. Developing a unified approach to sustainable consumption behavior: opportunities for a new environmental paradigm. Eur. J. Sustain. Dev. 7 (1). doi:10.14207/ejsd.2018.v7n1p1, 1-1.
- Chemnitz, C., Rehmer, C., Wenz, K., 2018. Fleischatlas 2018. Daten und Fakten über Tiere als Nahrungsmittel (Meat Atlas 2018. Facts and Figures About the Animals We Eat). Heinrich Böll Stiftung, Bund für Umwelt und Naturschutz, Le Monde Diplomatique.
- Chemnitz, C., Wenz, K., 2021. Fleischatlas 2021. Daten und Fakten über Tiere als Nahrungsmittel (Meat Atlas 2021. Facts and Figures About Animals We Eat). Heinrich Böll Stiftung, Bund für Umwelt und Naturschutz, Le Monde Diplomatique.
- Clausen, J., Mathes, M., 2017. Fleischkonsum. Transformationsfeldanalyse im Rahmen des Projekts Evolution2Green. Transformationspfade zu Einer Green Economy (Meat consumption. Transformation field Analysis As Part of the Evolution2Green project - transformation paths to a Green Economy). Borderstep Institut.
- Dagevos, H., Voordouw, J., 2013. Sustainability and meat consumption: is reduction realistic? Sustainability 9 (2), 60–69. doi:10.1080/15487733.2013.11908115.
- David, P.A., 1985. Clio and the economics of QWERTY. Am. Econ. Rev. 75 (2), 332-337.
- De Jesus, A., Mendonça, S., 2018. Lost in transition? Drivers and barriers in the ecoinnovation road to the circular economy. Ecol. Econ. 145, 75–89. doi:10.1016/j. ecolecon.2017.08.001.
- Easterby-Smith, M., Thorpe, R., Jackson, P., 2012. Management Research, 4th ed. Sage Publications.
- Engler, J.O., Abson, D.J., von Wehrden, H., 2019. Navigating cognition biases in the search of sustainability. Ambio 48 (6), 605–618. doi:10.1007/s13280-018-1100-5.
- Escribano, A.J., 2016. Organic livestock farming. Challenges, perspectives, and strategies to increase its contribution to the agrifood system's sustainability. a review. In: Konvalina, P. (Ed.), Organic Farming - A Promising Way of Food Production. InTechOpen, pp. 229–260. doi:10.5772/61272.
- FAO, 2011. World Livestock 2011-livestock in Food Security. Food and Agriculture Organization of the United Nations (FAO) http://www.fao.org/3/i2373e/i2373e. pdf.
- Fereday, J., Muir-Cochrane, E., 2006. Demonstrating rigor using thematic analysis: a hybrid approach of inductive and deductive coding and theme development. Int. J. Qual. Methods 5 (1), 80–92. doi:10.1177/160940690600500107.
- Festinger, L., 1957. A Theory of Cognitive Dissonance. Stanford University Press.

- Fleming, A., Vanclay, F., 2010. Farmer responses to climate change and sustainable agriculture. A review. Agron. Sustain. Dev. 30 (1), 11–19.
- Gioia, D.A., Corley, K.G., Hamilton, A.L., 2013. Seeking qualitative rigor in inductive research: notes on the Gioia methodology. Organ. Res. Methods 16 (1), 15–31. doi:10.1177/1094428112452151.
- Glaser, B.G., Strauss, A.L., 1967. The Discovery of Grounded Theory: Strategies For Qualitative Research. Aldine Transaction.
- Gossard, M.H., York, R., 2003. Social structural influences on meat consumption. Hum. Ecol. Rev. 10, 1–9.
- Govindharaj, G., Gowda, B., Sendhil, R., Adak, T., Raghu, S., Patil, N., Mahendiran, A., ChandraRath, P., Kumar, G.A.K., Damalas, C.A., 2021. Determinants of rice farmers' intention to use pesticides in eastern India: application of an extended version of the planned behavior theory. Sustain. Prod. Consum. 26, 814–823. doi:10.1016/j.spc.2020.12.036.
- Gupta, H., Kusi-Sarpong, S., Rezaei, J., 2020. Barriers and overcoming strategies to supply chain sustainability innovation. Resour. Conserv. Recycl. 161, 104819. doi:10.1016/j.resconrec.2020.104819.
- Guenther, E., Hueske, A.K., Stechemesser, K., Buscher, L., 2013. The 'why not'perspective of green purchasing: a multilevel case study analysis. J. Change Manage. 13 (4), 407–423. doi:10.1080/14697017.2013.851950.
- Gwin, L., 2009. Scaling-up sustainable livestock production: innovation and challenges for grass-fed beef in the US. J. Sustain. Agricul. 33 (2), 189–209. doi:10. 1080/10440040802660095.
- Hashempour-Baltork, F., Khosravi-Darani, K., Hosseini, H., Farshi, P., Reihani, S.F.S., 2020. Mycoproteins as safe meat substitutes. J. Clean. Prod. 253, 119958. doi:10. 1016/j.jclepro.2020.119958.
- Hermann, D., Mußhoff, O., Agethen, K., 2016. Investment behavior and status quo bias of conventional and organic hog farmers: an experimental approach. Renew. Agric. Food Syst. 31 (4), 318–329. doi:10.1017/S1742170515000265.
- Herren, H.R., Hilbeck, A., Hoffmann, U., Home, R., Levidow, L., Müller, A., Nelson, E., Oehen, B., Pimbert, M., 2015. Feeding the People: Agroecology for Nourishing the World and Transforming the Agri-Food System. IFOAM EU Group.
- Hobbs, J.E., 2021. The Covid-19 pandemic and meat supply chains. Meat Sci., 108459 doi:10.1016/j.meatsci.2021.108459.
- Hoek, A.C., Malekpour, S., Raven, R., Court, E., Byrne, E., 2021. Towards environmentally sustainable food systems: decision-making factors in sustainable food production and consumption. Sustain. Prod. Consum. 26, 610–626. doi: 10.1016/j.spc. 2020.12.009.
- Hyland, J.J., Jones, D.L., Parkhill, K.A., Barnes, A.P., Williams, A.P., 2016. Farmers' perceptions of climate change: identifying types. Agric. Human Values 33 (2), 323– 339. doi:10.1007/s10460-015-9608-9.
- Ilea, R.C., 2009. Intensive livestock farming. global trends, increased environmental concerns, and ethical solutions. J. Agricul. Environ. Ethics 22 (2), 153–167. doi:10.1007/s10806-008-9136-3.
- Ives, C.D., Abson, D.J., von Wehrden, H., Dorninger, C., Klaniecki, K., Fischer, J., 2018. Reconnecting with nature for sustainability. Sustain. Sci. 13 (5), 1389–1397. doi:10.1007/s11625-018-0542-9.
- Ives, C.D., Freeth, R., Fischer, J., 2020. Inside-out sustainability: the neglect of inner worlds. Ambio 49 (1), 208–217. doi:10.1007/s13280-019-01187-w.
- Ivlevs, A., 2019. Adverse welfare shocks and pro-environmental behavior: evidence from the global economic crisis. Rev. Income Wealth 65 (2), 293–311. doi:10. 1111/roiw.12355.
- Johnson, M., Schaltegger, S., 2020. Entrepreneurship for sustainable development: A review and multilevel causal mechanism framework. Entrepreneurship Theory and Practice 44 (6), 1144–1173. doi:10.1177/1042258719885368.
- Klitkou, A., Bolwig, S., Hansen, T., Wessberg, N., 2015. The role of lock-in mechanisms in transition processes: the case of energy for road transport. Environ. Innov. Soc. Transit. 16, 22–37. doi:10.1016/j.eist.2015.07.005.
- Kollmuss, A., Agyeman, J., 2002. Mind the gap: why do people act environmentally and what are the barriers to pro-environmental behavior? Environ. Educ. Res. 8 (3), 239–260. doi:10.1080/13504620220145401.
- Lang, D.J., Wiek, A., Bergmann, M., Stauffacher, M., Martens, P., Moll, P., Swilling, M., Thomas, C.J., 2012. Transdisciplinary research in sustainability science: practice, principles, and challenges. Sustain. Sci. 7 (1), 25–43. doi:10.1007/ s11625-011-0149-x.
- Läpple, D., Kelley, H., 2013. Understanding the uptake of organic farming: accounting for heterogeneities among Irish farmers. Ecol. Econ. 88, 11–19. doi:10.1016/ j.ecolecon.2012.12.025.
- Latif, S.A., Omar, M.S., Bidin, Y.H., Awang, Z., 2013. Role of environmental knowledge in creating pro-environmental residents. Proc.-Soc. Behav. Sci. 105, 866– 874. doi:10.1016/j.sbspro.2013.11.088.
- Laurett, R., Paço, A., Mainardes, E.W., 2021. Sustainable development in agriculture and its antecedents, barriers and consequences-An exploratory study. Sustain. Prod. Consum. 27, 298–311. doi:10.1016/j.spc.2020.10.032.
- Liu, Y., Wood, L.C., Venkatesh, V.G., Zhang, A., Farooque, M., 2021. Barriers to sustainable food consumption and production in China: a fuzzy DEMATEL analysis from a circular economy perspective. Sustain. Prod. Consum. 28, 1114–1129. doi:10.1016/j.spc.2021.07.028.
- Luederitz, C., Abson, D.J., Audet, R., Lang, D.J., 2017. Many pathways toward sustainability. Not conflict but co-learning between transition narratives. Sustain. Sci. 12 (3), 393–407. doi:10.1007/s11625-016-0414-0.
- Luthra, S., Kumar, S., Kharb, R., Ansari, M.F., Shimmi, S.L., 2014. Adoption of smart grid technologies: an analysis of interactions among barriers. Renew. Sustain. Energy Rev. 33, 554–565. doi:10.1016/j.rser.2014.02.030.
- Marchant-Forde, J.N., Boyle, L.A., 2020. COVID-19 effects on livestock production: a One Welfare issue. Front. Vet. Sci. 7, 734. doi:10.3389/fvets.2020.585787.

- Mellinger, N., 2000. Fleisch: Ursprung und Wandel einer Lust. Eine kulturanthropologische Studie. (Meat: Origin and Change of a lust. A cultural Anthropological Study). Campus Verlag.
- Nalau, J., Handmer, J., 2015. When is transformation a viable policy alternative? Environ. Sci. Policy 54, 349–356. doi:10.1016/j.envsci.2015.07.022.
- Nardone, A., Zervas, G., Ronchi, B., 2004. Sustainability of small ruminant organic systems of production. Livest. Prod. Sci. 90 (1), 27–39. doi:10.1016/j.livprodsci. 2004.07.004.
- Newhouse, N., 1991. Implications of attitude and behavior research for environmental conservation. J. Environ. Educ. 22 (1), 26–32. doi:10.1080/00958964.1990. 9943043.
- Newton, P., Blaustein-Rejto, D., 2021. Social and economic opportunities and challenges of plant-based and cultured meat for rural producers in the US. Front. Sustain. Food Syst. 5, 10. doi:10.3389/fsufs.2021.624270.
- Nierenberg, D., 2006. Rethinking the global meat industry. In: Starke, L. (Ed.), State of the World 2006: A Worldwatch Institute Report On Progress Toward a Sustainable Society. W. W. Norton & Company, pp. 22–38. doi:10.4324/9781849775953.
- O'Brien, K., Sygna, L., 2013. Responding to climate change: the three spheres of transformation. In: *Proceedings of Transformation in a Changing Climate*, 19-21 June 2013. University of Oslo, pp. 16-23.
- Pachirat, T., 2011. Every Twelve seconds: Industrialized slaughter and the Politics of Sight. Yale University Press.
- Panahzadeh Parikhani, M., Razzaghi Borkhani, F., Shabanali Fami, H., Motiee, N., Hosseinpoor, A., 2015. Major barriers to application of Good Agricultural Practices (GAPs) technologies in sustainability of livestock units. Int. J. Agricul. Manage. Dev. 5 (3), 169–178. doi:10.5455/ijamd.161640.
- Pauly, D., 1995. Anecdotes and the shifting baseline syndrome of fisheries. Trends Ecol. Evol. (Amst.) 10 (10), 430. doi:10.1016/s0169-5347(00)89171-5.
- Perrin, A., Martin, G., 2021. Resilience of French organic dairy cattle farms and supply chains to the Covid-19 pandemic. Agric. Syst. 190, 103082. doi:10.1016/j. agsy.2021.103082.
- Petri, H., Faust, H., 2021. Understanding permaculturist motivations among residents of the "PermaKulturRaum" in Goettingen, Germany: a qualitative analysis. SN Soc. Sci. 1 (1), 1–26. doi:10.1007/s43545-020-00024-2.
- Rodriguez, J.M., Molnar, J.J., Fazio, R.A., Sydnor, E., Lowe, M.J., 2009. Barriers to adoption of sustainable agriculture practices: change agent perspectives. Renew. Agric. Food Syst. 60–71. doi:10.1017/S1742170508002421.
- Rossi, J., Garner, S.A., 2014. Industrial farm animal production. A comprehensive moral critique. J. Agricul. Environ. Ethics 27 (3), 479–522. doi:10.1007/ s10806-014-9497-8.
- Sajjad, A., Eweje, G., Tappin, D., 2020. Managerial perspectives on drivers for and barriers to sustainable supply chain management implementation. Evidence from New Zealand. Bus. Strategy Environ. 29 (2), 592–604. doi:10.1002/bse. 2389.

Saldaña, J., 2015. The Coding Manual For Qualitative Researchers. Sage.

- Savin-Baden, M., Major, C.H., 2013. Qualititaive Research: The Essential Guide to Theory and Practice. Routledge.
- Schaltegger, S., Lüdeke-Freund, F., Hansen, E.G., 2016. Business models for sustainability: a co-evolutionary analysis of sustainable entrepreneurship, innovation, and transformation. Organ. Environ. 29 (3), 264–289. doi:10.1177/ 1086026616633272.
- Schaltegger, S., Hansen, E., 2017. Industry transformation through sustainable entrepreneurship. Examples in the apparel and energy industries. In: McIntosh, M. (Ed.), The Necessary transition. The journey Towards the Sustainable Enterprise Economy. Routledge/Taylor Francis, pp. 182–197.
- Shenton, A.K., 2004. Strategies for ensuring trustworthiness in qualitative research projects. Educ. Inf. 22 (2), 63–75. doi:10.3233/EFI-2004-22201.
- Siegel, L., Cutter-Mackenzie-Knowles, A., Bellert, A., 2018. Still 'minding the gap'sixteen years later:(re) storying pro-environmental behavior. Aust. J. Environ. Educ. 34 (2), 189–203. doi:10.1017/aee.2018.32.
- Sirilertsuwan, P., Hjelmgren, D., Ekwall, D., 2019. Exploring current enablers and barriers for sustainable proximity manufacturing. J, Fash. Mark. Manage 23 (4), 551–571. doi:10.1108/JFMM-09-2018-0114.
- Steinfeld, H., Gerber, P., Wassenaar, T., Castel, V., Rosales, M., de Haan, C., 2006. Livestock's Long Shadow. Food and Agriculture Organization of the United Nations (FAO) http://www.fao.org/3/a0701e/a0701e.pdf.
- Stern, P.C., 2000. Toward a coherent theory of environmentally significant behavior. J. Soc. Issues 56, 407–424. doi:10.1111/0022-4537.00175.
- Stoll-Kleemann, S., 2019. Feasible options for behavior change toward more effective ocean literacy: a systematic review. Front. Mar. Sci. 6, 273. doi:10.3389/fmars. 2019.00273.
- Stoll-Kleemann, S., Schmidt, U.J., 2017. Reducing meat consumption in developed and transition countries to counter climate change and biodiversity loss: a review of influence factors. Reg. Environ. Change 17 (5), 1261–1277. doi:10.1007/ s10113-016-1057-5.
- Strauss, A.L., 1987. Qualitative Analysis For Social Scientists. Cambridge University Press, Cambridge.
- Stuart, D., Schewe, R.L., McDermott, M., 2014. Reducing nitrogen fertilizer application as a climate change mitigation strategy: understanding farmer decisionmaking and potential barriers to change in the US. Land Use Policy 36, 210–218. doi:10.1016/j.landusepol.2013.08.011.
- Talay, C., Oxborrow, L., Brindley, C., 2018. An exploration of power asymmetry in the apparel industry in the UK and Turkey. Ind. Mark. Manage. 74, 162–174. doi:10.1016/j.indmarman.2018.03.008.
- Unruh, G.C., 2000. Understanding carbon lock-in. Energy Policy 28 (12), 817–830. doi:10.1016/S0301-4215(00)00070-7.

- Uusi-Rauva, C., Heikkurinen, P., 2013. Overcoming barriers to successful environmental advocacy campaigns in the organizational context. Environ. Commun. 7 (4), 475-492. doi:10.1080/17524032.2013.810164.
- Van der Werff, E., Steg, L., 2016. The psychology of participation and interest in smart energy systems: comparing the value-belief-norm theory and the valueidentity-personal norm model. Energy Res. Soc. Sci. 22, 107–114. doi:10.1016/j. erss.2016.08.022.
- Van de Ven, A.H., 2007. Engaged scholarship: A guide For Organizational and Social
- Vergee, Y.-L., 2007. Engaged Scholarship: A guide For Organizational and Social Research. Oxford University Press on Demand.
   Vergne, J.-P., Durand, R., 2011. The path of most persistence. An evolutionary per-spective on path dependence and dynamic capabilities. Org. Stud. 32 (3), 365– 382. doi:10.1177/0170840610397485.
- Virah-Sawmy, M., Durán, A.P., Green, J.M., Guerrero, A.M., Biggs, D., West, C.D., 2019. Sustainability gridlock in a global agricultural commodity chain: reframing the soy-meat food system. Sustain. Prod. Consum. 18, 210-223. doi:10.1016/j.spc. 2019.01.003.
- Yin, R.K., 2009. Case Study Research: Design and Methods, 4th ed. Sage.
- Zellweger, T.M., Astrachan, J.H., 2008. On the emotional value of owning a firm. Family Bus. Rev. 21 (4), 347–363. doi:10.1177/08944865080210040106.
- Zokaei, K., Simons, D., 2006. Performance improvements through implementation of Rev. 9 (1030–2016–82410), 30–53. doi:10.22004/ag.econ.8194.

# C. Intent for sustainability and contribution potential

Dijkstra-Silva, S., Hübel, C. & Schaltegger, S (2023): Company contributions to sustainability transitions of markets: An innovation management perspective (*to be submitted*)

# Company contributions to the sustainability transition of markets: An innovation management perspective

Samanthi Dijkstra-Silva<sup>a</sup>, Charlott Hübel<sup>b</sup> and Stefan Schaltegger<sup>b</sup>

<sup>a</sup>TU Dresden, Helmholtzstr. 10, 01069 Dresden <sup>b</sup>Leuphana Universität, Universitätsallee 1, 21335 Lüneburg

# Abstract

To contribute to sustainability transitions of markets requires companies to transform within and beyond their core business. Vast research suggests that companies can do so by engaging in sustainable product, process and organizational innovations. The extant literature has argued that such sustainability innovations can lead to a creative destruction of unsustainable market structures, including supply chains, consumption patterns, the competitive landscape and regulatory conditions. To date, a comprehensive and systematic view on how sustainability innovations can be *managed* to contribute to sustainability transitions of markets is however lacking. This article contributes to the sustainability transition literature by offering an innovation management perspective on company contributions to sustainable market transitions. A conceptual framework is developed that distinguishes five scopes of market transitions that companies can target: the organization itself (core), supply chains and suppliers (input), consumers and consumer systems (the output), competitors and alternative organizations (the substitute) and the socio-political environment (the arena). Linking the five scopes to innovation processes, products and organizations results in fifteen fields of action in which companies can select targeted management measures. Based on the framework, different contribution pathways of companies' are discussed theoretically and with practical case examples for potential company contributions to the sustainability transition of markets.

Keywords: sustainability transition, innovation management, market transition, company contribution

# **1. Introduction**

Companies, whether new entrants or incumbent firms, can play a crucial role in contributing to the sustainability transition of markets (Geels and Schot, 2007; Loorbach and Wijsman, 2013; Geels, 2014). The relevance of companies for sustainability transitions is grounded in their embeddedness in and influence on different market environments, including the economic, cultural and political (Geels, 2014). Due to their constant interaction with these environments, companies can re-structure supply chains (Thorlakson et al., 2018), change consumer behavior (Pinkse and Bohnsack, 2021), shape the competitive landscape (Schaltegger et al., 2016) and generate far-reaching impact on societal infrastructures and institutions (Bolton and Hannon, 2016), as well as, ultimately, the natural environment (Whiteman et al., 2013). By effectively addressing all environments relevant to the sustainability transition of markets, companies can become game changers in their markets (Loorbach and Wijsman, 2013; Schaltegger et al., 2022). To generate such transformative impact, companies are often required to introduce and manage sustainability innovations on a product, process and organizational level (Klewitz and Hansen, 2014; Silvestre and Ţîrcă, 2019).

Much of the previous analyses on innovative company contributions to sustainability transitions of markets takes an overarching systems perspective with an emphasis on market interactions between different companies (e.g. Hockerts and Wüstenhagen, 2010; Markard et al., 2012). Other studies are located at the intersection of companies, innovation and policy (Delmas et al., 2019; Kivimaa and Rogge, 2022). A considerable body of literature, furthermore, discusses examples of companies and their innovation-driven sustainability contributions to organizational (Blum-Kusterer and Hussain, 2001; Hübel, 2022), market (Hörisch, 2018; Hansen and Schaltegger, 2013; Schaltegger et al., 2016) or societal transformations (Bolton and Hannon, 2016; Bidmon and Knab, 2018). While these analyses are instructive in illustrating transformative innovations of companies, our conceptual understanding as to how companies apply management measures aimed at the sustainability transition of markets is limited.

The existing literature reveals both, the relevance of company innovations and contributions to sustainability transitions as well as a huge variety of different kinds of corporate sustainability contributions. This important research, linking niche-level induced sustainability effects beyond organizational boundaries, is, however, disperse. A framework that supports a systematic management of innovative company contributions to different scopes of sustainability transitions of markets is missing, so far. With increasing sustainability challenges, the demand for research increases on how companies can address their contributions to the sustainability transition in a systematic manner (Schaltegger et al., 2020). Various authors have called for a management perspective, suggesting to link concepts and frameworks used in management studies with sustainability transition-related research questions (Markard et al., 2012; Bolton and Hannon, 2016; Köhler et al., 2019).

Addressing this research gap, the following conceptual research analytically combines research on sustainability transition and innovation management and develops a theoretical framework for managing company innovations aimed at sustainability transitions of markets. It proposes various management measures at the intersection of three, often distinguished types of sustainability innovations, i.e. process, product and organizational, and five scopes of market transition that can be identified in the transition management literature, i.e. the company (the core), supply chains (the input), consumption of the sold products and services (the output) and competition (the substitute) as well as regulations, governance and culture (the arena). The framework has relevant practical implications in terms of a systematic analysis of management options as well as the prioritization of management measures and adequate strategic orientations.

The framework contributes to the academic literature by taking an innovation management perspective to sustainability transitions of markets. By combining largely separate bodies of literature, the paper advances conceptual clarity by introducing scopes of transitions which can be targeted by management. Based on these scopes, contribution pathways are identified, which support associated company contributions to a sustainability transition. Implied assumptions, managerial intent and choice of measures for targeted contribution pathways are discussed critically.

The paper is structured as follows: After a review of the literature on sustainability transitions of markets, we outline the innovation management perspective and introduce three types of sustainability innovations that are often distinguished. We thereupon develop a framework by distinguishing different scopes of market transition, linking them to different types of sustainability innovations and identifying management measures. The framework is discussed with regard to managerial intent and choice of measures. We highlight how the framework can guide the development of contribution pathways, which may help systematize and increase a company's contribution to the sustainability transition of markets.

# 2. Companies and sustainability transitions of markets

In the sustainability transition literature, the sustainability transition of markets is considered a crucial sub-pillar of a system-wide societal sustainability transition (Geels and Schot, 2007). Sustainability transitions of markets entail the fundamental re-orientation of established markets towards creating social and environmental values (Loorbach and Wijsman, 2013). We briefly outline three lines of discussion in the sustainability transition literature relevant for our analysis; the relevance of companies for a sustainability transition of markets, the embeddedness of companies in the market environment as well as effective means to achieve a market transition.

Addressing the first line of discussion, the sustainability transition and entrepreneurship literature argues that companies, as key actors within a market, can initiate and shape the sustainability transition of markets (Geels and Schot, 2007; Loorbach and Wijsman, 2013; Schaltegger and Wagner 2011). The role of companies in market transitions has been mainly analyzed taking a system perspective (Markard et al., 2012; Loorbach et al., 2017). The sustainability transitions research domain applies a systems perspective to explain how dynamics at different system levels can lead to the large-scale diffusion of sustainability innovations and the replacement of unsustainable practices (Geels and Schot, 2007; Geels 2011; Chappin and Ligtvoet, 2014; Geels, 2014). Similarly, the sustainable entrepreneurship research domain regards sustainable markets as the result of individual (e.g. Schaltegger, 2002; Dixon and Clifford, 2007) and co-evolutionary market interactions between incumbent firms and sustainability start-ups in market niches (Hockerts and Wüstenhagen, 2010; Schaltegger et al., 2016). Both research domains consider sustainability start-ups in market niches as important sources of radical novelties and as initiators of transition processes, while incumbent firms of established regimes are often assigned a rather reactive role in transitions. More recently, the role of incumbents in sustainability transitions has been highlighted (Schaltegger et al., 2022). Some authors have challenged the bottom-up bias and suggested a distributed notion of agency, acknowledging proactiveness in both niche and regime actors (Geels, 2011; 2020; Berggren et al., 2015) as well as in strategic alliances (Hübel et al., 2022). Recognizing the crucial role of both incumbents and newcomers, both large and small companies, for the sustainability transition of markets, the framework developed in this paper seeks suitability for all types of companies.

With regard to the second line of discussion, market transitions are understood to not only require changes in product offerings and purchasing behavior but to go along with more profound changes within and between organizations as well as the overarching social, cultural and political environments (Farla et al., 2012; Markard et al., 2012). In this regard, research has acknowledged the embeddedness of companies in and influence on these environments (Battilana et al., 2009; Coenen et al., 2012; Rogge and Reichard, 2016). The activities of a company to advance market transitions are therefore not limited to organizational boundaries. Geels' (2014) triple embeddedness framework details how companies are embedded in and simultaneously shape the economic and socio-political environment. While the economic environment constitutes market relationships to suppliers, consumers and competitors, the socio-political environment highlights companies' links to political decision-makers, institutions and civil society groups (Turnheim and Geels, 2012). By employing strategies and measures that are not limited to the organization itself, but also influence suppliers, consumers, competitors and socio-political conditions, companies can substantially advance changes in markets and industries – and ultimately society at large (Loorbach et al., 2010; Loorbach and Wijsman, 2013; Loorbach et al., 2017; Geels, 2014).

Addressing the effective transformation of markets, the transition literature points to the introduction and diffusion of sustainability innovations by companies (Elzen et al., 2004; Geels and Schot, 2007; Geels, 2014; Smink et al., 2015). In this regard, much of the literature has taken a systems perspective. Various studies describe the diffusion of innovations from niche to regime level (Geels and Schot 2007) as well as the co-evolution of innovation-focused start-ups and 'laggard' incumbents (Schaltegger et al., 2016). Few studies have zoomed in to firm level, taking a strategic focus (Geels, 2014; Berggren et al., 2015) or conceptualizing internal transition management (Loorbach and Wijsman, 2013; Schaltegger et al., 2022). However, a detailed conceptualization of what scopes of market transition companies can address with the management of sustainability innovations is lacking so far. This is despite the fact that previous research has pointed to different impacts created by sustainability innovations (Klewitz and Hansen, 2014; Adams et al., 2016) and has explicitly put a management perspective on the 'agenda for sustainability transitions research' (Köhler et al., 2019).

To lay the groundwork for the link between managing sustainability innovations and corporate contributions to the sustainability transition of markets, the next section deals with the innovation management perspective introducing three types of innovations for sustainability, process, product and organization innovations.

# 3. An innovation management perspective

Companies are key enablers of innovation (Weber and Rohracher, 2012) and can initiate processes of creative destruction through the introduction and diffusion of innovations in the market (Schumpeter, 1943; Schaltegger et al., 2018; Johnson and Schaltegger, 2020). The success of innovations is largely reliant on a company's effective innovation management (Tushman and Nadler, 1986; Tidd, 2001). To achieve large-scale impact in the market, innovations for sustainability therefore need to be managed deliberately (Klewitz and Hansen, 2014). This section therefore distinguishes different types of innovations and associated management measures.

The sustainability-oriented and conventional innovation literature often distinguishes three types of innovations: Process, product and organizational innovations (Polder et al., 2010; Ballot et al., 2015; Klewitz and Hansen 2014; Tavassoli and Karlsson, 2015; Adams et al., 2016; Carboni and Russu, 2018; see Table 1). Process innovations are often considered to be the starting point for management to address sustainability, and can trigger subsequent changes in the product life cycle or organization (Klewitz and Hansen, 2014). In corporate practice, process innovations are often incremental (e.g. Carboni and Russu, 2018). Product innovations that change the value offered by the company fundamentally, can be both incremental and radical (Dangelico and Pujari, 2010). Organizational innovations, such as e.g. business model innovations, mostly have a rather fundamental impact on all essential domains of organizational activity and thus are often radical in nature (Boons and Lüdeke-Freund, 2013; Schaltegger et al., 2012).

	Type of sus- tainability	Definition	Examples / Concepts	Literature
_	innovation			
	Process inno- vations for sustainability	Improvement of produc- tion or supply chain pro- cesses through e.g. envi- ronmental technologies, which are facilitated by non-technological changes	<ul> <li>Cleaner production</li> <li>Zero waste production</li> <li>Eco-efficiency</li> <li>Industrial symbiosis</li> <li>Dematerialization</li> <li>Circular production</li> </ul>	Adams et al., 2016; Altham, 2007; Boer and During, 2001; Chertow, 2000, 2007; Gurbuz et al., 2004; Klewitz and Hansen, 2014; Michel- sen and Fet, 2010; Moyano- Fuentes et al., 2018
	Product inno- vations for sustainability	New products or services or a combination of both, which seek to achieve market differentiation while at the same time minimizing environmen- tal impacts and/or being socially responsible	<ul> <li>Sustainable product design</li> <li>Sustainable sourcing</li> <li>No/Low-waste packaging</li> <li>Sustainable product label- ling and certification</li> <li>Product-service systems</li> </ul>	Bhamra et al., 2008; Borin et al., 2011; Dangelico and Pujari, 2010; de Medeiros et al., 2014; Iyer and Soberman, 2016; Jaffry et al., 2004; Kennedy et al., 2017; Mont, 2002; Rennings, 2000; Tang and Bahmra, 2008

Table 1: Process, product and organizational innovations
Organiza- tional inno- vations for sustainability ganizations Fundamental changes to structures, routines and management practices, leading to new types of or- ganizations	<ul> <li>Systematic integration of sustainability into the core business</li> <li>Sustainable business model innovation</li> <li>Systems innovation</li> </ul>	Antikainen and Valkokari, 2016; Bolton and Hannon, 2016; Chesbrough, 2010; Evans et al., 2017; Gaziulusoy et al., 2013; Klewitz and Hansen, 2013, 2014; Rennings, 2000; Schaltegger et al., 2012, 2016; Schiederig et al., 2012; Schumpeter 1943; Silva 2021
--	--	---

To manage process innovations for sustainability, companies are required to access or create new knowledge, adapt equipment or procedures and integrate them into existing performance measurement and reporting systems (Robertson et al., 2012; Morioka and Carvalho, 2016). A strong focus on learning is a basis to advance a company's process innovation capability (Frishammar et al., 2012) such as business experimentation capabilities for sustainability (Weissbrod and Bocken, 2017) or life cycle thinking (Kralisch et al., 2017). To implement changes, process innovations for sustainability typically require the allocation and coordination of intraorganizational resources – and intra-organizational collaboration (Gopalakrishnan et al., 1999) - but can also involve external stakeholders along the supply chain (Adams et al., 2016). The latter may include introducing, co-developing and monitoring sustainability-oriented operational change at existing supplier companies (Walton et al., 1998). To ensure effectiveness, it is necessary to monitor the adherence to these codes of conduct and guidelines continuously and introduce fines for misconduct. Integration can be supported by formalized management tools such as environmental management systems (Rennings et al., 2006; Wagner, 2008), environmental management accounting (Burritt et al., 2002) as well as social and ethical codes of conduct for business operations (Haugh and Talwar, 2010).

Similarly, *product innovations for sustainability* can be managed successfully through internally focused innovation-oriented learning and knowledge management (Cormican and O'Sullivan, 2003; De Medeiros et al., 2014) or externally oriented partnering and collaboration (Bhamra et al., 2008; Klewitz and Hansen, 2014; Melander, 2017). Both internally and externally focused approaches usually require companies to integrate sustainability performance metrics in product development (Kennedy et al., 2017) and purchasing strategies (Murray, 2000), often aided by life-cycle assessments (Chang et al., 2014). Due to the novelty and interdisciplinary nature of sustainable product innovations, management often relies on acquisitions (e.g. Austin and Leonard, 2008) or cooperative strategies including collaborative technology scouting, open innovation and strategic learning alliances (Kennedy et al., 2017; Hübel et al., 2022). Collaborative methods include NGO-business-partnerships (Pattberg, 2004; Mousavi and Bossink, 2020) as well as consumer-integrated product development and user-led innovation (Ornetzeder and Rohracher, 2006; Hoffmann, 2007). To ensure far-reaching impact, product innovations for sustainability need to take the attractiveness for mainstream consumers into account, including few or no compromises with conventional product usage preferences (Pinkse and Bohnsack, 2021). Nudging strategies (Bodur et al., 2015; Demarque et al., 2015; Lehner et al., 2015), sustainability labelling (van Dam and Jonge, 2015) as well as transparency management (Montecci et al., 2021) can further facilitate sustainability-oriented consumer choices.

The management of organizational innovations for sustainability requires transformational leaders supporting innovation- and learning-oriented cultures (Miles et al., 2009; Chen and Chang, 2012; Senge et al., 2015; Adams et al., 2016), sustainability visions (Ferdig, 2007; Schaltegger et al., 2012; Pieroni et al., 2019) as well as boundary spanning and empowerment for sustainability-oriented change (Razavi and Attarnezhad, 2013). Learning and awareness building for sustainability can be facilitated through e.g. employee trainings and other company initiatives in education for sustainable development (Shen and Benson, 2016). To ensure comprehensive change along the entire value chain, a company can transform its own business models (Boons and Lüdecke-Freund, 2013) and further engage in collaborative business modelling (Heikkilä and Heikkilä, 2013; Rohrbeck et al., 2013) to encourage sustainable business model innovations at suppliers that align with the sustainable value proposition of the focal company (Norris et al., 2021). Due to their comprehensiveness and disruptive character, organizational innovations require further management skills. These include the ability to coordinate people with heterogeneous interests (Van de Ven, 1995), to consider both formal aspects, such as structure and tools, as well as informal aspects, such as culture and mindsets (Boer and During, 2001) and to balance sometimes contradictory demands for economic, social and environmental solutions (Ferdig, 2007; Stubbs, 2019).

Depending on the choice of innovation for sustainability, a company's contribution to a sustainability transition of markets may differ. To date, however, the contribution of a company to a sustainability transition (of markets) remains a rather fuzzy term, which needs further conceptualization. The next section combines transitions and innovation management perspectives to conceptualize possible contributions to sustainability transformations beyond organizational boundaries by introducing a theoretical framework that distinguishes scopes of market transitions and by discussing their link to contributions to a sustainability transition.

### 4. Framework for managing company innovations for market transitions

In the following, we develop a framework for managing company innovations aimed at a sustainability transition of markets. We conceptually advance our understanding of a sustainability transition of markets by distinguishing five scopes of market transitions, and linking these to the three types of sustainability innovations (Table 2): The first dimension of the framework identifies different scopes of market transitions, which a company can consider, the second dimension addresses the management of innovations for sustainability. By linking insights from these two literature streams, we identify fifteen fields of sustainability transition action illustrated with exemplary management measures across different scopes. We argue that, while companies may already apply these measures individually to reach strategic aims, our framework is profoundly based on managerial intent to contribute to sustainability transitions of markets. To substantiate the applicability of the theoretical framework, we reviewed two practice examples, the food company Unilever as an incumbent and the energy company Lichtblick as a start-up (see Appendix A). While some measures might have resulted in a sustainability transition of selected markets, e.g. the Marine Stewardship Council (MSC) certification by raising standards in the fishing industry affecting suppliers and consumers by Unilever, other activities have not resulted in a sustainability transition of markets. This has two implications. First, the need to select measures carefully when aiming at a sustainability transition of markets (see Chapter 4.3 under "choice of measures"). Second, some management approaches provide a base for further levers (partially addressed in Chapter 5 'company contribution pathways'). For example, when offering of a sustainable product the focal company needs to be sustainable itself. Additionally, a sustainability mindset of suppliers, e.g. through supplier training, might be necessary to realize sustainability solutions.

The following section goes on to (1) identify five scopes of market transition, (2) link them with innovation management approaches in an overall framework, and (3) lay out the framework's basic assumptions in terms of management intent and choice of measures.

### **4.1 Scopes of market transition**

Recent research emphasizes the necessity to analyze how companies can contribute to change within and beyond organizational boundaries, i.e. change at the level of the organization, the level of markets as well as on the level of society and the natural environment (Johnson and Schaltegger, 2020; Rijnsoever, 2022). The link between managing sustainability innovations at

the micro level of a company and creating management contributions to sustainability transitions at the market level, however, has not been conceptualized so far. By drawing both on the sustainability transitions literature (Loorbach and Wijsman, 2013; Loorbach et al. 2017; Geels, 2014) and the sustainability innovation management and entrepreneurship literature (Hockerts and Wüstenhagen, 2010; Schaltegger and Wagner, 2011), our framework distinguishes *five scopes of a market transition*: Organizations can focus on innovations that aim at a sustainability transition in the company itself (*the core*), supply chains (*the input*), consumption patterns (*the output*), competitor behaviors (*the substitute*) and the socio-political environment (*the arena*).

Fundamental for any sustainability transition of a market is managing the organization itself. The organization addresses *the core*, which needs to operate sustainably (Loorbach and Wijsman, 2013) in order to create any changes beyond its boundaries. Companies can pursue corporate sustainability through advancing fundamental organizational change encompassing a company's strategy, processes, structures, identity and culture (Kungl and Hess, 2021).

A company's value creation, however, is not only centered in the company itself, but also in supply chains through the purchase of *inputs* (Thorlakson et al., 2018). In many cases, activities in supply chains and therefore inputs are linked to sustainability problems such as pollution, transport emissions, hazardous materials, conflict minerals, poverty and workers' health (Yawar and Seuring, 2015; Silva and Schaltegger, 2019). By sourcing sustainable materials and engaging in sustainable supply chain management practices, companies can actively advance sustainability in supplier companies and support mainstreaming sustainable practices in supplier markets (Metta and Badurdeen, 2012; Boström et al., 2015; Broemer et al., 2019).

On the *output* side, companies have the power to support sustainability-oriented change in purchasing and product use behavior of consumers through sustainable product design and respective marketing strategies (White et al., 2019). To change consumption patterns in a market towards sustainability, involves creating a critical mass needed for market transitions and requires companies to go beyond targeting 'the green consumer segment' and instead address mass market consumers (Schaltegger 2002; White et al., 2019).

A company's actions can further trigger changes in competitor behavior, the *substitute* (e.g. the companies that could substitutes the company if it is not successful). With sustainability becoming a competitive topic in a market, other companies may copy sustainable practices, re-

sulting in a co-evolution of increasingly sustainable companies and therefore the transition towards increasingly sustainable markets (Hockerts and Wüstenhagen, 2010; Schaltegger et al., 2016). Companies may also choose to collaborate with competitors directly, for example by codeveloping or co-marketing sustainable products (Bengtsson and Kock, 1999; Christ et al., 2017; Tian et al., 2019).

Lastly, companies are not just regulation takers bound by societal conditions, but can also actively influence regulations, public policies, norms and mindsets representing the industry regime, i.e. the *arena* (Bendell and Kearins, 2005; Markard and Truffer, 2008; Geels et al., 2014). Approaches of corporate politics have been discussed under the notion of responsible and ethical lobbying (Scherer and Palazzo, 2011; Barron and Skountridaki, 2020). Many involve collaborations with societal and political actors (Moldovan et al., 2016), the aim being to establish industry initiatives, codes of conduct, standards and norms for sustainability (Giovannucci and Ponte, 2005). The academic literature has addressed these phenomena differently through e.g. the notion of "institutional entrepreneurship" (Nasra and Dacin, 2010; Schaltegger and Wagner, 2011) or intermediaries (Kivimaa et al., 2019). The sustainable entrepreneurship literature has further emphasized that entrepreneurial companies can correct market and government failure (e.g. Dean and McMullen, 2007).

For all five scopes of company contributions, sustainability transitions of markets can be facilitated through the innovation of processes, products and organizations. However, innovation efforts need to be *managed* to create considerable effects (Tushman and Nadler, 1986; Tidd, 2001).

### 4.2 Managing sustainability innovations across scopes of market transition

Combining the management of three innovation approaches, i.e. process, product and organizational innovations, across scopes of market transition results in various management measures that can contribute to the sustainability transition of markets. Summarized in Table 2, this section discusses how managerial approaches can contribute to transitions and provides exemplary management measures. Management measures differ not only with the desired scope of the contribution but also with the type of innovation employed. Taking the example of supply chains, a company can increase its contribution with regard to the *input* by managing process, product and organizational innovations, measures which have been discussed in the sustainable supply chain management literature (e.g. Seuring and Müller, 2008). For instance, managing a company's contribution to supply chain sustainability through *process* innovation requires a company to train suppliers for new sustainability practices and support and incentivize their implementation and/or sanction non-compliance with supplier codes of conduct. Through continuous monitoring and assessment, a suppliers' progress in sustainability performance can be ensured. Managing the same contribution scope of *inputs*, but through *product* innovation, already requires a different set of managerial skills. Here, conducting life cycle assessments of purchased products and drawing the necessary conclusions for sustainable sourcing strategies is vital for achieving sustainable product supplies. Lastly, *organizational* innovations in supply chains require the focal company to support and incentivize thorough integration of sustainability standards in supplier companies. This can be achieved through collaborative business modelling, sharing best practices, training and educating (e.g. Harms et al., 2013).

	Scopes of market transitions						
		CORE (the company)	<b>INPUT</b> (supply chains)	OUTPUT (consumption)	SUBSTITUTE (competition)	ARENA (regulation, govern- ance and culture)	
Managing types of sustainability innovations       Organization     Products	Processes	<ul> <li>Adapting knowledge to sustainability requirements</li> <li>Shifting resources to align operations/production with sustainability standards</li> <li>Assessing and reporting progress (e.g. Environmental management accounting &amp; systems)</li> </ul>	<ul> <li>Training suppliers</li> <li>Incentivizing and supporting sustainable production pro- cesses</li> <li>Assessing and monitoring suppliers' sustainability measures (through supplier questionnaires; codes of conducts)</li> <li>Implementing sanction</li> </ul>	<ul> <li>Monitoring of consumption patterns of customers/con- sumers (surveys of what consumers do)</li> <li>Managing the type, fre- quency and geographical lo- cation of delivery, distribu- tion and communication channels</li> </ul>	<ul> <li>Collaborating with competitors on sustainable industry standards and business practices</li> <li>Sharing best practices</li> </ul>	<ul> <li>Managing actor networks</li> <li>Holding stakeholder dia- logues (e.g. for creating shared industry standards)</li> </ul>	Innovated processes
	Products	<ul> <li>Knowledge management (research &amp; knowledge inte- gration)</li> <li>Integrating sustainability metrics in product design</li> <li>Engaging in open innovation &amp; forming strategic alliances</li> </ul>	<ul> <li>Managing supply chains for sustainable products</li> <li>Conducting life cycle as- sessments</li> <li>Super scouting resource materials; developing sus- tainable sourcing strategies</li> <li>Supply chain transparency management</li> </ul>	<ul> <li>Increasing consumer dialogues to achieve product co-development</li> <li>Implementing consumer education and communication campaigns</li> <li>Acquiring an established sustainable product label or certication</li> </ul>	- Managing knowledge & rela- tionships for co-develop- ment of products with com- petitors	<ul> <li>Lobbying responsibly &amp; ethically for sustainable product standards</li> <li>Building public-private partnerships for new product standards and labels</li> <li>Cooperating with media to educate consumers on sustainability issues of product supply chains</li> </ul>	Innovated products
	Organization	<ul> <li>Adopting a transformational &amp; systems approach to lead- ership</li> <li>Fostering a learning- and sustainability-oriented cul- ture</li> <li>Training employees</li> <li>Implementing company initi- atives to increase internal visibility and communication</li> </ul>	<ul> <li>Training, mobilizing and incentivizing suppliers for integrating sustainability</li> <li>Engaging in collaborative business modelling with suppliers</li> </ul>	<ul> <li>Targeting mainstream consumers</li> <li>Implementing marketing strategies that consider psychological aspects of consumer behavior</li> <li>Nudging through use of descriptive norms and prediction requests</li> </ul>	<ul> <li>Creating and innovating collaborative business models for sustainability</li> <li>Mutual learning, harmonizing, coordinating operations and standards</li> <li>Establishing joint supplier assessment and training centers with competitors</li> </ul>	<ul> <li>Cooperating with e.g. NGOs, media, regulators to develop new industry standards/regulations</li> <li>Co-creating multi-organi- zational sustainable busi- ness models that create new sustainability-oriented markets</li> </ul>	Innovated organization
	Transformed core     Transformed inputs     Transformed outputs     Transformed substitutes     Transformed arena						

### **4.3** Managerial intent and choice of measures

Our framework is based on two main assumptions: First, managerial intent is a core requirement for contributing to lasting, large-scale sustainability transition of markets. Second, the selection of adequate management measures differs with each company, for example with regard to size, resource base, market position, and business environment.

Given global climate change predictions and other sustainability challenges (Rockström et al., 2009; Stern and Stern, 2007), companies are increasingly required to select and manage their innovations with the concrete aim of contributing to sustainability transitions of markets (Schaltegger et al., 2022). We argue that having a *managerial intent for sustainability* is essential for achieving the desired contribution to market transition. In doing so, the proposed framework differs from previous approaches that may have strategic purpose with regard to sustainability or competitiveness, but often neither put the sustainability transition of markets in the center nor take an innovation management approach to contributing intentionally to sustainability transitions. Our framework, in contrast, emphasizes a company's deliberate intent to contribute to a sustainability transition of markets through the careful selection of targeted management measures. As a basis for managerial intent, sustainability transition and innovation research emphasize the importance of a company's sustainability orientation (Kennedy et al., 2017; van Lieshout et al., 2021).

Once a company targets the sustainability transition of markets, a core issue is the starting point for management. At a first glance, the framework seems to suggest that a company's contribution to sustainability transitions could be maximized if it succeeds implementing various management measures at the intersection of all three innovation types and five scopes of market transitions. This may, however, not necessitate the simultaneous pursuit of all managerial approaches nor may it be attainable or desirable. The potential inadequacy of applying all managerial approaches equally and simultaneously is demonstrated by the following reasoning: Depending on a company's size, resources, value proposition and market environment, the required focus and the feasibility to contribute successfully to a sustainability transition of the market may differ. Choosing an adequate starting point therefore depends on a company's and measure's impact potential, i.e. its potential to contribute successfully to a sustainability transition of markets. The intentional choice of management measures may help a company stay within its financial possibilities and at the same time maximizing its contribution to market transition. While our framework displays the range of areas that can be addressed to create sustainability contributions, the choice of the adequate managerial focus needs to be assessed specifically for each company, product and market environment. We discuss how a maximial contribution to sustainability transitions of markets can be managed in the section on company contribution pathways (Section 5).

Taking the case of synthetic fertilizers as an example, their main sustainability problems are caused in the use on agricultural land (Lykogianni et al., 2021). For a multinational fertilizer producer, managing process innovations at the output scope and contributing to changed agricultural practices (e.g. training of farmers to prevent overuse and ensure efficient application as well as better monitoring of how much is needed at what moment of time) may therefore be crucial. A start-up offering fairly and sustainably produced smartphones might prioritize to manage product innovations at the input scope, as smartphone production is linked to sustainability issues in its supply chains such as conflict materials and high emissions (Wernink and Strahl, 2015). A key challenge for any company managing their innovations aimed at sustainability transitions is therefore to intentionally identify the appropriate management measure(s).

Geels (2014), in his triple embeddedness framework, suggests that continuous learning and sensemaking can help a company to select appropriate approaches to influence different industry environments. Through these continuous activities, companies can interpret environmental developments, organize information and evaluate appropriate management measures. Based on Geels' ideas, it can be assumed that the sustainable phone producer opted for managing sourcing strategies and product design after having evaluated the market for smartphones and having identified unsustainable supply chains as an important issue to address. The company's focus on product innovation and its consequences for innovations in the supply chain results from these initial sensemaking and learning activities. While the prerequisites of strategy and management selection have been discussed in the literature, the eventual contribution of the chosen managerial focus to different market environments has not been explored, so far. Particularly, it has remained unclear whether companies can influence the extent of their contributions based on the selection of management measures.

This section has developed a new framework that offers companies different fields of actions for managing their innovations aimed at the sustainability transition of markets. In the following, the framework will be discussed with regard to its implications for the extent of individual company contributions.

### 5. Company contribution pathways

Previous research has established that companies can engage in different sustainability innovations to contribute to sustainable development in general (Klewitz and Hansen, 2014; Silvestre and Ţîrcă, 2019) and sustainable market transitions in particular (Bolton and Hannon, 2016; Geels and Schot, 2007; Schaltegger et al., 2016). With our developed theoretical framework, we conceptualized possible management measures at the intersection of scopes of market transitions and types of sustainability innovations. The framework postulates that for companies to contribute to an effective sustainability transition of markets, they can and need to intentionally select suitable managerial starting points that correspond with a company's individual context. However, the question remains whether and how companies can influence the eventual outcome for market transition, i.e. whether and how companies can maximize their transition contribution. In this regard, we suggest that each management measure has a specific *contribution pathway (CP)*. The pathway concept proposes that each selected management measure not only manages the direct innovation in one scope of market transition but can trigger further changes across different scopes. One management measure can thus become the starting point of a wider dissemination throughout the market.

To analyze the extent of a company's contribution to the sustainability transformation of markets, we therefore distinguish different CPs with distinct measures of sustainability transition management. At first sight, the term contribution pathway suggests a link to the typology of sociotechnical transition pathways (Geels and Schot, 2007; Geels et al., 2016). However, Geels and Schot's original typology takes a system perspective and discusses pathways based on differences in timing and nature of multi-level interactions and pressures. This research, does not focus on external pressures, but highlights management choice in one organization's transition scope, which triggers change in other scopes. Both sustainability transition and corporate sustainability transformation research emphasize the effectiveness of an interplay between comprehensive changes at the intersection of core and arena (mindset, norms, beliefs) as well as inputs, outputs and substitutes (co-evolution of actors) (Geels and Schot, 2007; Hockerts and Wüstenhagen, 2010; Schaltegger et al., 2016). In this regard, Schaltegger and Johnson (2019, p.23) argue that "the combination of developing ventures with sustainability-oriented values at the core [...] and co-evolution of market actors [...] appears as an effect pathway for mesolevel sustainability-oriented market innovations and transformations".

Further, the progressive nature of innovation-driven market transition has been addressed in transition research (Geels and Schot, 2007; Loorbach et al., 2017; Schaltegger and Wagner,

2010) and links to arguments of actor interaction and co-evolution in sustainability transformation (Hockerts and Wüstenhagen, 2010; Schaltegger et al., 2016). Most closely related to this work, Geels (2014) has conceptualized causal mechanisms within different market environments: Pressures are first exerted by activists, then spill-over to the wider public, thereupon to policymakers or consumers and then to companies, who can react strategically to these pressures. Instead of adopting such an outside-in-perspective, our work deliberately takes an insideout-perspective, making management activities of individual companies the starting point for market transition such as suggested in the sustainable entrepreneurship literature (e.g. Jolink and Niesten, 2015; Parrish, 2010). For the sake of clarity, the following discussion focuses on one management choice per transition scope and presents *idealized contribution pathways* triggered by one choice. Table 3 distinguishes the resulting five *contribution pathways* (triggered by one exemplary management measure) and identifies the resulting contribution of the organization to a sustainability transition of markets.

The first contribution pathway, CP1, starts from management measures for innovating the *core*. CP1 provides a foundation for all following choices. If a company lacks a sustainable core, it risks that all further management measures aimed at a sustainability transition of markets are considered greenwashing (Laufer, 2003). Thus, ensuring a sustainable core is a pre-requisite for other changes. A sustainable core covers the company's sustainability mindset and values impact how seriously the intent becomes to implement management measures aimed at other contribution pathways. By changing the focal organization, i.e. the core, the company contribution is identified as transformed core,  $CC^1$ .

CP2 starts from management measures for innovating the *input*, which, in turn, affects the core. For instance, if a company engages in management measures for sustainable sourcing, internal purchasing policies, product development and marketing strategies will be adapted, leading to a transformation of both, input and core. By transforming the input, change affects multiple level such as new materials, processes, mindsets of individuals of input organizations, i.e. different suppliers as well as the core, resulting in a transformed input and core and a corporate contribution to the power of two,  $CC^2$ .

CP3 starts from measures for innovating the *output*, including, for instance, offering a novel sustainable product including nudging strategies. This measure can, in turn, affect product development and marketing at the *core* and sourcing at the *input*. A transformed output can affect a large part of consumers by not only offering alternative products but creating alternative mindsets. An example could be the marketing campaign by Patagonia stating "Do not buy this

jacket". While a transformed output focuses on consumer level change, it requires pre-requisite transformations at input and core, resulting in a corporate contribution to the power of 3, CC<sup>3</sup>.

CP4 starts from measures for innovating the *substitute*, such as co-creating sustainable innovations with competitors. This measure not only changes competitor's offering, but can also affect activities at the *core*, *output* and *input*, resulting in a company contribution to the power of four. Finally, CP5 starts from measures for innovating the *arena*, creating the potentially longest pathway. It includes managing actor networks or private-public partnerships, often involving the co-creation of sustainability innovations. While most immediately, the resulting innovations will influence institutional conditions (culture, regulations), the increased sustainability requirements will require adaptations at all organizations, including *core* and *substitute*, which, in turn, might lead to further changes at the *input* and the *output*. With the subsequent triggers, a contribution to a sustainability transformation of markets affects all five areas, resulting in a company contribution to the power of five, i.e.  $CC^5$ . While the ultimate contribution is difficult to quantify, we argue that it is more than mere addition because these changes are exponentiated by affecting a variety of (interrelated) individuals and organizations.

Contribu- tion path- way (CP)	Starting point of the managing organiza- tion Example of manage- ment measure	<b>Transitions triggered in SCOPE:</b> (potential causal chains)	Company contribution (CC) to the sustainabil- ity transition of mar- kets
CP1	Innovating the CORE Example measure: Business transfor- mation	<b>CORE</b> Fundament for all further innovations and management measures aimed at a sustainability transition of markets	CC <sup>1</sup> = Transformed core
CP2	<b>Innovating the</b> <b>INPUT</b> Increasing socio-eco- logical standards in supply chains	<ul> <li>INPUT → CORE</li> <li>Input: Supplier trainings to transform suppliers to achieve envisioned standards or switching suppliers thereby signaling the procurement market (of inputs) that more sustainable inputs are desired</li> <li>Core, e.g. organization introduces supplier code of conduct at core organization, adapting procurement policies</li> </ul>	CC <sup>2</sup> =Transformed core and input

Table 3: Idealized contribution pathways to the sustainability transition of markets

СРЗ	<b>Innovating the</b> <b>OUTPUT</b> Example measure: Increasing consumers sustainability mindset	<ul> <li>OUTPUT → CORE → INPUT</li> <li>Output: Sustainable product is offered and incentivizes sustainable consumer behavior</li> <li>Core: Marketing strategy ensures the offered product is sustainable, e.g. changes in purchasing / suppliers</li> <li>Input: Product input is manufactured sustainably</li> </ul>	CC <sup>3</sup> = Transformed out- put, core and input
CP4	Innovating the SUBSTITUTE Example measure: Co-creating sustaina- ble market structures with competitor(s)	<ul> <li>SUBSTITUTE → OUTPUT →</li> <li>CORE → INPUT</li> <li>Substitute: Competitors switch to more sustainable products, changing their own output, input and core</li> <li>Output: Consumers choices change to more sustainable offers</li> <li>Core: Collaborations improve sustainability of own organization</li> <li>Input: Higher sustainability standards of suppliers, such as higher supplier compensations (e.g. fair trade)</li> </ul>	CC <sup>4</sup> =Transformed core, input, output and substi- tute
CP5	Innovating the ARENA Example measure: Elevating sustainabil- ity point of reference by (co-)creating sus- tainability certifica- tion	<ul> <li>ARENA → SUBSTITUTE → OUT- PUT → CORE → INPUT</li> <li>Arena: Introduce certificate, ensure high trust through higher market standards, regulations and collabora- tions with NGOs</li> <li>Substitute: Competitors restructure offer to comply with certification</li> <li>Output: Availability of certified products</li> <li>Core: Collaboration with NGO to develop sustainability certification</li> <li>Input: Ensure product is aligned with high sustainability standard of certificate</li> </ul>	CC <sup>5</sup> =Transformed core, input, output, substitute and arena

Linking the CPs to practice shows different interrelations between the scopes, confirming that, to trigger CPs, each company needs to carefully manage their measures with the intent of sustainability transitions of markets (see table with practice examples in Appendix B). CP5 could be observed, for instance, at Unilever. The company partnered with the WWF to co-develop the Marine Stewardship Council certification (MSC, 2022; Ponte, 2012 REF). Most fundamentally, the activity contributed to a market transition at the arena by introducing a new standard, a private governance institution and label (Ponte, 2012). This, in turn, resulted in changes at the core, as Unilever shifted resources to accommodate for the changed purchasing policies. The altered purchasing policies, based on the new standards, resulted in changes at the input by increasing the number of certified sustainable fisheries. Identifying sustainable inputs on product packaging then increased consumer awareness of sustainability in fish products, thus triggering changes at the output level. Finally, the increased awareness and demand for sustainable

fish increased the overall amount of sustainably certified fish of other food producing companies, large food service and food distribution companies, thus impacting the substitute. Today, 14% of marine catch have been certified with the MSC, arguably a transformative contribution of 14% after 25 years (Arton et al., 2020; MSC, 2022).

In essence, a company's contribution (CC) is the achieved extent of transition, meaning the degree of replacement of unsustainable market structures by sustainable alternatives at the core, the input, the output, the substitute and/or the arena. How much a company can contribute to sustainability transitions of markets with each management measure depends on the careful and focused management of the contribution pathway which promises furthest-reaching impact in the market.

### 6. Conclusion

By taking a management perspective and linking the innovation management and sustainability transition literature, we propose a theoretical framework for managing innovations aimed at the sustainability transition of markets. The framework distinguishes five scopes of market transitions - the core, input, output, substitute and arena -, and offers management measures by linking each transition scope to the management of sustainable process, product and organizational innovations. The effective implementation of management measures depends on (1) a company's management intent to achieve a sustainability transformation of markets and (2) the careful selection of adequate management measures based on the potential company contribution to the sustainability transition of markets. The previous section discussed potential contribution pathways for each selected starting point based on scope. The proposed conceptual framework aims to support companies in carefully selecting management measures with the aim of a sustainability transition of markets and to become a game changer in its market.

Future research benefits from the theoretical framework to structure systematic approaches for managerial interventions aimed at the sustainability transition of markets. For management research, the framework can support developing tools for measuring specific contributions. So far, the maximization of company contributions to market transition can merely be estimated by identifying the type of innovation measure employed and comparing it to the current market share and impact of the respective company. However, for higher rigor, concrete and ideally measurable indicators are needed. Therefore, developing this contribution further to assess its impacts and distilling indicators to use for controlling and reporting relevant data may be a worthwhile future research endeavor. In addition, further research may want to explore how the

framework can capture all categories without risking to attribute effects of one category to another. With its conceptual focus, this contribution has not dealt with underlying drivers and levers of sustainability transitions. Recent sustainability transformation literature for example, considers underlying "deep" changes in mindsets, beliefs, values, norms and competencies as most important leverage points for large-scale transitions (Abson et al., 2017; Ives et al., 2020; Buhr et al., 2023). The link between actor mindsets in different scopes and different innovation management approaches to innovation foci and scopes of sustainability contributions therefore need to be investigated further. Lastly, while referencing some illustrative practice examples, it was beyond the scope of this conceptual research to conduct empirical research analyzing company contributions and respective innovation management techniques in practice. Future indepth case studies can provide a more direct analysis oriented towards the presented framework, revealing crucial company activities in the different categories.

Lastly, the framework assists companies to consider their sustainability commitments and respective impacts more intentionally. It does so by (1) encouraging companies to consider their contributions beyond the organizational boundaries and (2) engage in a process that allows them to select adequate management measures for maximum impact.

# References

Adams, R., Jeanrenaud, S., Bessant, J., Denyer, D., & Overy, P. (2016). Sustainability-oriented innovation: A systematic review. *International Journal of Management Reviews*, *18*(2), 180-205. https://doi.org/10.1111/ijmr.12068

Antikainen, M., & Valkokari, K. (2016). A framework for sustainable circular business model innovation. *Technology Innovation Management Review*, 6(7). http://doi.org/10.22215/timre-view/1000

Altham, W. (2007). Benchmarking to trigger cleaner production in small businesses: drycleaning case study. *Journal of Cleaner Production*, 15(8-9), 798-813. https://doi.org/10.1016/j.jclepro.2006.07.005

Arton, A., Leiman, A., Petrokofsky, G., Toonen, H., & Longo, C. S. (2020). What do we know about the impacts of the Marine Stewardship Council seafood ecolabelling program? A systematic map. *Environmental Evidence*, *9*(1), 1-20. https://doi.org/10.1186/s13750-020-0188-9

Austin, J. E., & Leonard, H. B. D. (2008). Can the virtuous mouse and the wealthy elephant live happily ever after?. *California Management Review*, *51*(1), 77-102. https://doi.org/10.2307/41166469

Ballot, G., Fakhfakh, F., Galia, F., & Salter, A. (2015). The fateful triangle: Complementarities in performance between product, process and organizational innovation in France and the UK. *Research Policy*, *44*(1), 217-232. https://doi.org/10.1016/j.respol.2014.07.003

Barron, A., & Skountridaki, L. (2022). Toward a Professions-Based Understanding of Ethical and Responsible Lobbying. *Business & Society*, 61(2), 340-371. https://doi.org/10.1177/0007650320975023

Battilana, J., Leca, B., & Boxenbaum, E. (2009). 2 how actors change institutions: towards a theory of institutional entrepreneurship. *Academy of Management Annals*, 3(1), 65-107. https://doi.org/10.5465/19416520903053598

Bendell, J. & Kearins, K. (2005): The political bottom line: the emerging dimension to corporate responsibility for sustainable development, *Business Strategy and the Environment*, 14(6), 372–383. https://doi.org/10.1002/bse.439

Berggren, C., Magnusson, T., & Sushandoyo, D. (2015). Transition pathways revisited: established firms as multi-level actors in the heavy vehicle industry. *Research Policy*, 44(5), 1017-1028. https://doi.org/10.1016/j.respol.2014.11.009

Bhamra, T., Lilley, D., & Tang, T. (2008). Sustainable use: changing consumer behaviour through product design. *Changing the Change: Design Visions, Proposals and Tools, Turin, 2008, Proceedings.* 

Bidmon, C. M., & Knab, S. F. (2018). The three roles of business models in societal transitions: New linkages between business model and transition research. *Journal of Cleaner Production*, 178, 903-916. https://doi.org/10.1016/j.jclepro.2017.12.198

Blom, E. M., Burg, E., Verhagen, P., & Hillen, M. (2014). Tony's Chocolonely. How a social entreprise is changing the chocolate industry. Stichting Social Entreprise NL. Available at: https://www.archief.social-enterprise.nl/files/1814/4198/0581/Tonys\_Chocolonely\_Teaching\_Case\_Version\_Beta.pdf

Blum-Kusterer, M., & Hussain, S. S. (2001). Innovation and corporate sustainability: An investigation into the process of change in the pharmaceuticals industry. *Business strategy and the environment*, *10*(5), 300-316. https://doi.org/10.1002/bse.300

Bodur, H. O., Duval, K. M., & Grohmann, B. (2015). Will you purchase environmentally friendly products? Using prediction requests to increase choice of sustainable products. *Journal of Business Ethics*, 129(1), 59-75. https://doi.org/10.1007/s10551-014-2143-6

Boer, H., & During, W. E. (2001). Innovation, what innovation? A comparison between product, process and organizational innovation. *International Journal of Technology Management*, 22(1-3), 83-107. https://www.academia.edu/30825605/

Bolton, R., & Hannon, M. (2016). Governing sustainability transitions through business model innovation: Towards a systems understanding. *Research Policy*, *45*(9), 1731-1742. https://doi.org/10.1016/j.respol.2016.05.003

Boons, F., & Lüdeke-Freund, F. (2013). Business models for sustainable innovation: state-of-the-art and steps towards a research agenda. *Journal of Cleaner Production*, *45*, 9-19. https://doi.org/10.1016/j.jclepro.2012.07.007

Borin, N., Cerf, D. C., & Krishnan, R. (2011). Consumer effects of environmental impact in product labeling. *Journal of Consumer Marketing*. https://doi.org/10.1108/07363761111101976

Boström, M., Jönsson, A. M., Lockie, S., Mol, A. P., & Oosterveer, P. (2015). Sustainable and responsible supply chain governance: challenges and opportunities. *Journal of Cleaner Production*, 107, 1-7. https://doi.org/10.1016/j.jclepro.2014.11.050

Broemer, J., Brandenburg, M., & Gold, S. (2019). Transforming chemical supply chains toward sustainability. A practice-based view. *Journal of Cleaner Production*, 236, 117701. https://doi.org/10.1016/j.jclepro.2019.117701

Buhr, M., Harms, D., & Schaltegger, S. (2023). Individual change agents for corporate sustainability transformation: a systematic literature review. *Benchmarking: An International Journal*, (ahead-of-print). https://doi.org/10.1108/BIJ-09-2021-0551

Carboni, O. A., & Russu, P. (2018). Complementarity in product, process, and organizational innovation decisions: Evidence from European firms. *R&D Management*, 48(2), 210-222. https://doi.org/10.1111/radm.12284

Cascade (2021). How Patagonia Became The Benchmark in Sustainable Clothing. Cascade. Available at: https://www.cascade.app/strategy-factory/studies/patagonia-strategy-study [Accessed 22 Nov 2022].

Chang, D., Lee, C. K. M., & Chen, C. H. (2014). Review of life cycle assessment towards sustainable product development. *Journal of Cleaner Production*, *83*, 48-60. https://doi.org/10.1016/j.jclepro.2014.07.050

Chappin, E. J.L. & Ligtvoet, A. (2014). Transition and transformation: a bibliometric analysis of two scientific networks researching socio-technical change, *Renewable and Sustainable Energy Reviews*, 30, 715–723. https://doi.org/10.1016/j.rser.2013.11.013

Chen, Y. S., & Chang, C. H. (2013). The determinants of green product development performance: Green dynamic capabilities, green transformational leadership, and green creativity. *Journal of Business Ethics*, 116(1), 107-119. https://doi.org/10.1007/s10551-012-1452-x

Chertow, M. R. (2000). Industrial symbiosis: literature and taxonomy. *Annual Review of Energy and the Environment*, 25(1), 313-337. https://doi.org/10.1146/annurev.energy.25.1.313

Chertow, M. R. (2007). "Uncovering" industrial symbiosis. *Journal of Industrial Ecology*, 11(1), 11-30. https://doi.org/10.1162/jiec.2007.1110

Chesbrough, H. (2010). Business model innovation: opportunities and barriers. *Long Range Planning*, 43(2-3), 354-363. https://doi.org/10.1016/j.lrp.2009.07.010

Chouinard, Y., & Stanley, V. (2013). *The responsible company: What we've learned from Patagonia's first 40 years*. Patagonia.

Christ, K. L., Burritt, R. L., & Varsei, M. (2017). Coopetition as a potential strategy for corporate sustainability. *Business Strategy and the Environment*, 26(7), 1029-1040. https://doi.org/10.1002/bse.1967

Dixon, S.E.A., & Clifford, A. (2007). Ecopreneurship – a new approach to managing the triple bottom line, 20(3), 326-345. https://doi.org/10.1108/09534810710740164

Coenen, L., Benneworth, P., & Truffer, B. (2012). Toward a spatial perspective on sustainability transitions. *Research Policy*, 41(6), 968-979. https://doi.org/10.1016/j.respol.2012.02.014

Cormican, K., & O'Sullivan, D. (2003). A collaborative knowledge management tool for product innovation management. *International Journal of Technology Management*, 26(1), 53-67. https://doi.org/10.1504/IJTM.2003.003144

Dangelico, R. M., & Pujari, D. (2010). Mainstreaming green product innovation: Why and how companies integrate environmental sustainability. *Journal of Business Ethics*, 95(3), 471-486. https://doi.org/10.1007/s10551-010-0434-0

Dean, T. J., & McMullen, J. S. (2007). Toward a theory of sustainable entrepreneurship: Reducing environmental degradation through entrepreneurial action. *Journal of Business Venturing*, 22(1), 50-76. https://doi.org/10.1016/j.jbusvent.2005.09.003

De Jesus, A., & Mendonça, S. (2018). Lost in transition? Drivers and barriers in the eco-innovation road to the circular economy. *Ecological Economics*, 145, 75-89. https://doi.org/10.1016/j.ecolecon.2017.08.001

Delmas, M. A., Lyon, T. P., & Maxwell, J. W. (2019). Understanding the role of the corporation in sustainability transitions. *Organization & Environment*, 32(2), 87-97. https://doi.org/10.1177/1086026619848255

De Medeiros, J. F., Ribeiro, J. L. D., & Cortimiglia, M. N. (2014). Success factors for environmentally sustainable product innovation: a systematic literature review. *Journal of Cleaner Production*, 65, 76-86. https://doi.org/10.1016/j.jclepro.2013.08.035

Demarque, C., Charalambides, L., Hilton, D. J., & Waroquier, L. (2015). Nudging sustainable consumption: The use of descriptive norms to promote a minority behavior in a realistic online shopping environment. *Journal of Environmental Psychology*, 43, 166-174. https://doi.org/10.1016/j.jenvp.2015.06.008

Elzen, B., Geels, F. W., & Green, K. (Eds.). (2004). *System innovation and the transition to sustainability: theory, evidence and policy*. Edward Elgar Publishing.

Evans, S., Vladimirova, D., Holgado, M., Van Fossen, K., Yang, M., Silva, E. A., & Barlow, C. Y. (2017). Business model innovation for sustainability: Towards a unified perspective for creation of sustainable business models. *Business Strategy and the Environment*, 26(5), 597-608. https://doi.org/10.1002/bse.1939

Fairphone (2020). Press Release. Impact Report 2020: Fairphone nearly doubles sales in 2020 as its industry influence increases. Available at: https://www.fairphone.com/wp-content/up-loads/2021/06/Impact-Report-2020-Fairphone-nearly-doubles-sales-in-2020-as-its-industry-influence-increases.pdf [Accessed 22 Nov 2022]

Farla, J. C. M., Markard, J., Raven, R., & Coenen, L. E. (2012). Sustainability transitions in the making: A closer look at actors, strategies and resources. *Technological Forecasting and Social Change*, 79(6), 991-998. https://doi.org/10.1016/j.techfore.2012.02.001

Ferdig, M. A. (2007). Sustainability leadership: Co-creating a sustainable future. *Journal of Change Management*, 7(1), 25-35. https://doi.org/10.1080/14697010701233809

Frishammar, J., Kurkkio, M., Abrahamsson, L., & Lichtenthaler, U. (2012). Antecedents and consequences of firms' process innovation capability: a literature review and a conceptual framework. *IEEE Transactions on Engineering Management*, *59*(4), 519-529. https://doi.org/10.1109/TEM.2012.2187660

Galpin, T., Whittington, J.L. and Bell, G. (2015), "Is your sustainability strategy sustainable? Creating a culture of sustainability", Corporate Governance, Vol. 15 No. 1, pp. 1-17. https://doi.org/10.1108/CG-01-2013-0004

Gaziulusoy, A. İ.; Boyle, C. & McDowall, R. (2013): System innovation for sustainability: a systemic double-flow scenario method for companies, *Journal of Cleaner Production*, 45, 104–116. https://doi.org/10.1016/j.jclepro.2012.05.013

Geels, F. W., Hekkert, M. P., & Jacobsson, S. (2008). The dynamics of sustainable innovation journeys, *Technology Analysis & Strategic Management*, 20, 521-536. https://doi.org/10.1080/09537320802292982

Geels, F. W. (2011): The multi-level perspective on sustainability transitions: responses to seven criticisms, *Environmental Innovation and Societal Transitions*, 1(1), 24–40. https://doi.org/10.1016/j.eist.2011.02.002

Geels, F. W. (2014). Reconceptualising the co-evolution of firms-in-industries and their environments: Developing an inter-disciplinary Triple Embeddedness Framework. *Research Policy*, 43(2), 261-277. https://doi.org/10.1016/j.respol.2013.10.006

Geels, F. W. (2020). Micro-foundations of the multi-level perspective on socio-technical transitions: Developing a multi-dimensional model of agency through crossovers between social constructivism, evolutionary economics and neo-institutional theory. *Technological Forecasting and Social Change*, *152*, 119894. https://doi.org/10.1016/j.techfore.2019.119894

Geels, F. W., & Schot, J. (2007). Typology of sociotechnical transition pathways. *Research Policy*, *36*(3), 399-417. https://doi.org/10.1016/j.respol.2007.01.003

Giovannucci, D. & Ponte, S. (2005). Standards as a new form of social contract? Sustainability initiatives in the coffee industry. *Food Policy*, 30(3), 284-301. https://doi.org/10.1016/j.foodpol.2005.05.007

GMP (2022). Green Mountain Power Named to TIME's List of the TIME100 Most Influential Companies. Green Mountain Power. Available at: https://greenmountainpower.com/news/gmp-named-to-time100-most-influential-companies/ [Accessed 22 Nov 2022]

Goldstein, D. R., & Russo, M. V. (2017). Seventh Generation: Balancing Customer Expectations with Supply Chain Realities. In *Case Studies in Sustainability Management and Strategy* (pp. 30-51). Routledge. Gopalakrishnan, S., Bierly, P., & Kessler, E. H. (1999). A reexamination of product and process innovations using a knowledge-based view. *Journal of High Technology Management Research*, 1(10), 147-166. https://doi.org/10.1016/S1047-8310(99)80007-8

Gurbuz, S., Kiran-Ciliz, N., & Yenigun, O. (2004). Cleaner production implementation through process modifications for selected SMEs in Turkish olive oil production. *Journal of Cleaner Production*, 12(6), 613-621. https://doi.org/10.1016/S0959-6526(03)00121-5

Hansen, E., & Schaltegger, S. (2013). 100 per cent organic? A sustainable entrepreneurship perspective on the diffusion of organic clothing. *Corporate Governance*, 13(5), 583–598. https://doi.org/10.1108/CG-06-2013-0074

Harms, D., Hansen, E. G., & Schaltegger, S. (2013). Strategies in sustainable supply chain management: an empirical investigation of large German companies. *Corporate Social Responsibility and Environmental Management*, 20(4), 205-218. https://doi.org/10.1002/csr.1293

Haugh, H. M., & Talwar, A. (2010). How do corporations embed sustainability across the organization? *Academy of Management Learning & Education*, 9(3), 384-396. https://doi.org/10.5465/amle.9.3.zqr384

Heikkilä, M., & Heikkilä, J. (2013). *Collaborative business model innovation process for net-worked services*. In International conference on electronic commerce (pp. 133-147). Springer, Berlin, Heidelberg.

Hidalgo, A., & Albors, J. (2008). Innovation management techniques and tools: a review from theory and practice. *R&D Management*, 38(2), 113-127. https://doi.org/10.1111/j.1467-9310.2008.00503.x

Hockerts, K. & Wüstenhagen, R. (2010): Greening Goliaths versus emerging Davids: theorizing about the role of incumbents and new entrants in sustainable entrepreneurship, *Journal of Business Venturing*, 25(5), 481–492. https://doi.org/10.1016/j.jbusvent.2009.07.005

Hoffmann, E. (2007). Consumer integration in sustainable product development. *Business Strategy and the Environment*, 16(5), 322-338. https://doi.org/10.1002/bse.577

Hörisch, J. (2018). How business actors can contribute to sustainability transitions: A case study on the ongoing animal welfare transition in the German egg industry. *Journal of Cleaner Production*, 201, 1155–1165. https://doi.org/10.1016/j.jclepro.2018.08.031

Hübel, C. (2022). Entrepreneurship-driven organizational transformation for sustainability: a sensemaking lens. *Journal of Organizational Change Management*, *35*(1), 240-256. https://doi.org/10.1108/JOCM-03-2021-0067

Hübel, C., Weissbrod, I., & Schaltegger, S. (2022). Strategic alliances for corporate sustainability innovation: The 'how'and 'when'of learning processes. *Long Range Planning*, 102200. https://doi.org/10.1016/j.lrp.2022.102200

Iyer, G., & Soberman, D. A. (2016). Social responsibility and product innovation. *Marketing Science*, 35(5), 727-742. https://doi.org/10.1287/mksc.2015.0975

Jaffry, S., Pickering, H., Ghulam, Y., Whitmarsh, D., & Wattage, P. (2004). Consumer choices for quality and sustainability labelled seafood products in the UK. *Food Policy*, 29(3), 215-228. https://doi.org/10.1016/j.foodpol.2004.04.001

Johnson, M. P., & Schaltegger, S. (2020). Entrepreneurship for sustainable development: A review and multilevel causal mechanism framework. *Entrepreneurship Theory and Practice*, 44(6), 1141-1173. https://doi.org/10.1177/1042258719885368

Jolink, A., & Niesten, E. (2015). Sustainable development and business models of entrepreneurs in the organic food industry. *Business Strategy and the Environment*, 24(6), 386-401. https://doi.org/10.1002/bse.1826

Kennedy, S., Whiteman, G., & van den Ende, J. (2017). Radical innovation for sustainability: The power of strategy and open innovation. Long Range Planning, 50(6), 712-725. https://doi.org/10.1016/j.lrp.2016.05.004

Kivimaa, P., Boon, W., Hyysalo, S., & Klerkx, L. (2019). Towards a typology of intermediaries in sustainability transitions: A systematic review and a research agenda. Research Policy, 48(4), 1062-1075. https://doi.org/10.1016/j.respol.2018.10.006

Kivimaa, P., & Rogge, K. S. (2022). Interplay of policy experimentation and institutional change in sustainability transitions: The case of mobility as a service in Finland. Research Policy, 51(1), 104412. https://doi.org/10.1016/j.respol.2021.104412

Klewitz, J., & Hansen, E. G. (2014). Sustainability-oriented innovation of SMEs: a systematic review. Journal of Cleaner Production, 65, 57-75. https://doi.org/10.1016/j.jcle-pro.2013.07.017

Köhler, J.; Geels, F. W.; Kern, F.; Markard, J.; Wieczorek, A. & Alkemade, F. et al. (2019): An agenda for sustainability transitions research: State of the art and future directions, *Environmental Innovation and Societal Transitions*, 31, 1-32. https://doi.org/10.1016/j.eist.2019.01.004

Köhler, J., Wietschel, M., Whitmarsh, L., Keles, D. & Schade, W. (2010). Infrastructure investment for a transition to hydrogen automobiles. *Technological Forecasting and Social Change*, 77(8), 1237-1248. https://doi.org/10.1016/j.techfore.2010.03.010

Kraaijenbrink, J. (2019). How To Bring Sustainability To The Masses: Tony's Chocolonely's Impact Strategy. Forbes. Available at: https://www.forbes.com/sites/jeroenkraaijenbrink/2019/11/08/how-to-bring-sustainability-to-the-masses-tonys-chocolonely-impact-strategy/?sh=2d3dd80b712a [Accessed 22 Nov 2022]

Kralisch, D., Ott, D., Lapkin, A. A., Yaseneva, P., De Soete, W., Jones, M., ... & Finkbeiner, M. (2018). The need for innovation management and decision guidance in sustainable process design. *Journal of Cleaner production*, 172, 2374-2388. https://doi.org/10.1016/j.jcle-pro.2017.11.173

Kungl, G., & Hess, D. J. (2021). Sustainability transitions and strategic action fields: A literature review and discussion. *Environmental Innovation and Societal Transitions*, *38*, 22-33. https://doi.org/10.1016/j.eist.2020.10.004

Laufer, W. S. (2003). Social accountability and corporate greenwashing. *Journal of business ethics*, 43, 253-261. https://doi.org/10.1023/A:1022962719299

Lehner, M., Mont, O., & Heiskanen, E. (2016). Nudging–A promising tool for sustainable consumption behaviour?. *Journal of Cleaner Production*, 134, 166-177. https://doi.org/10.1016/j.jclepro.2015.11.086

Loorbach, D., & Wijsman, K. (2013). Business transition management: exploring a new role for business in sustainability transitions. *Journal of Cleaner Production*, *45*, 20-28. https://doi.org/10.1016/j.jclepro.2012.11.002

Loorbach, D., Frantzeskaki, N., & Avelino, F. (2017). Sustainability transitions research: Transforming science and practice for societal change. *Annual Review of Environment and Resources*, 42, 599–626. https://doi.org/10.1146/annurev-environ-102014-021340 Lykogianni, M., Bempelou, E., Karamaouna, F., & Aliferis, K. A. (2021). Do pesticides promote or hinder sustainability in agriculture? The challenge of sustainable use of pesticides in modern agriculture. *Science of The Total Environment*, 795, 148625. https://doi.org/10.1016/j.scitotenv.2021.148625

Markard, J.; Raven, R. & Truffer, B. (2012): Sustainability transitions: an emerging field of research and its prospects, *Research Policy*, 41(6), 955–967.

Markard, J., & Truffer, B. (2008). Technological innovation systems and the multi-level perspective: Towards an integrated framework. *Research Policy*, 37(4), 596-615. https://doi.org/10.1016/j.respol.2008.01.004

Melander, L. (2017). Achieving sustainable development by collaborating in green product innovation. *Business Strategy and the Environment*, 26(8), 1095-1109. https://doi.org/10.1002/bse.1970

Metta, H., & Badurdeen, F. (2012). Integrating sustainable product and supply chain design: modeling issues and challenges. *IEEE Transactions on Engineering Management*, 60(2), 438-446. https://doi.org/10.1109/TEM.2012.2206392

Michelsen, O., & Fet, A. M. (2010). Using eco-efficiency in sustainable supply chain management; a case study of furniture production. *Clean Technologies and Environmental Policy*, 12(5), 561-570. https://doi.org/10.1007/s10098-009-0266-8

Miles, M. P., Munilla, L. S., & Darroch, J. (2009). Sustainable corporate entrepreneurship. *International Entrepreneurship and Management Journal*, 5(1), 65-76. https://doi.org/10.1007/s11365-008-0074-3

Moldovan, A. M., Greenley, M., & Lakatos, E. S. (2016). Corporate social responsibility, NGOs and business partnerships for social sustainability. *Review of Applied Socio-Economic Research*, 11(1), 51-56.

Mont, O. K. (2002). Clarifying the concept of product–service system. *Journal of Cleaner Production*, *10*(3), 237-245. https://doi.org/10.1016/S0959-6526(01)00039-7

Montecchi, M., Plangger, K., & West, D. C. (2021). Supply chain transparency: A bibliometric review and research agenda. *International Journal of Production Economics*, 238, 108152. https://doi.org/10.1016/j.ijpe.2021.108152

Morehouse, C. (2019). Green Mountain Power aims for 100% renewables by 2030 with up to 100 MW storage. Utility Dive. Available at: https://www.utilitydive.com/news/green-moun-tain-power-aims-for-100-renewables-by-2030-with-up-to-100-mw-sto/552852/ [Accessed 22 Nov 2022].

Morioka, S. N., & Carvalho, M. M. (2016). Measuring sustainability in practice: exploring the inclusion of sustainability into corporate performance systems in Brazilian case studies. *Journal of Cleaner Production*, 136, 123-133. https://doi.org/10.1016/j.jclepro.2016.01.103

Mousavi, S. & Bossink, B. (2020). Corporate-NGO partnership for environmentally sustainable innovation: Lessons from a cross-sector collaboration in aviation biofuels, *Environmental Innovation and Societal Transitions*, 34, 80-95. https://doi.org/10.1016/j.eist.2019.12.005

Moyano-Fuentes, J., Maqueira-Marín, J. M., & Bruque-Cámara, S. (2018). Process innovation and environmental sustainability engagement: An application on technological firms. *Journal of Cleaner Production*, 171, 844-856. https://doi.org/10.1016/j.jclepro.2017.10.067

Murray, J. G. (2000). Effects of a green purchasing strategy: the case of Belfast City Council. *Supply Chain Management: An International Journal*, 5(1), 37-44. https://doi.org/10.1108/13598540010312954 MSC (2022). Our collective impact. Marine Stewardship Council. Available at: https://www.msc.org/what-we-are-doing/our-collective-impact [Accessed 22 Nov 2022]

Nasra, R., & Dacin, M. T. (2010). Institutional arrangements and international entrepreneurship: the state as institutional entrepreneur. *Entrepreneurship Theory and Practice*, 34(3), 583-609. https://doi.org/10.1111/j.1540-6520.2009.00354.x

Norris, S., Hagenbeck, J., & Schaltegger, S. (2021). Linking sustainable business models and supply chains—Toward an integrated value creation framework. *Business Strategy and the Environment*. 2021, 1-15. https://doi.org/10.1002/bse.2851

Nylund, P. A., Brem, A., & Agarwal, N. (2021). Innovation ecosystems for meeting sustainable development goals: The evolving roles of multinational enterprises. *Journal of Cleaner Production*, 281, 125329. https://doi.org/10.1016/j.jclepro.2020.125329

Ornetzeder, M., & Rohracher, H. (2006). User-led innovations and participation processes: lessons from sustainable energy technologies. *Energy Policy*, *34*(2), 138-150. https://doi.org/10.1016/j.enpol.2004.08.037

Ørsted (2021). Our green business transformation: What we did and lesson learned. Ørsted White Paper. Available at: https://orsted.com/en/insights/white-papers/green-transformation-lessons-learned [Accessed 22 Nov 2022]

Madsen, H. L., & Ulhøi, J. P. (2021). Sustainable visioning: Re-framing strategic vision to enable a sustainable corporate transformation. *Journal of Cleaner Production*, 288, 125602. https://doi.org/10.1016/j.jclepro.2020.125602

O'Rourke, D., & Strand, R. (2017). Patagonia: Driving sustainable innovation by embracing tensions. *California Management Review*, *60*(1), 102-125. https://doi.org/10.1177/0008125617727748

Parrish, B. D. (2010). Sustainability-driven entrepreneurship: Principles of organization design. *Journal of Business Venturing*, 25(5), 510-523. https://doi.org/10.1016/j.jbusvent.2009.05.005

Pattberg, P. (2004). *Private environmental governance and the sustainability transition: functions and impacts of NGO-business partnerships*. Klaus Jacob, Manfred Binder and Anna Wieczorek (eds.), 52-66.

Pieroni, M. P., McAloone, T. C., & Pigosso, D. C. (2019). Business model innovation for circular economy and sustainability: A review of approaches. *Journal of Cleaner Production*, 215, 198-216. https://doi.org/10.1016/j.jclepro.2019.01.036

Pinkse, J., & Bohnsack, R. (2021). Sustainable product innovation and changing consumer behavior: Sustainability affordances as triggers of adoption and usage. *Business Strategy and the Environment*. https://doi.org/10.1002/bse.2793

Ponte, S. (2012). The Marine Stewardship Council (MSC) and the making of a market for 'sustainable fish'. *Journal of Agrarian Change*, *12*(2-3), 300-315. https://doi.org/10.1111/j.1471-0366.2011.00345.x

Rattalino, F. (2018). Circular advantage anyone? Sustainability-driven innovation and circularity at Patagonia, Inc. *Thunderbird International Business Review*, 60(5), 747-755. https://doi.org/10.1002/tie.21917

Razavi, S. H., & Attarnezhad, O. (2013). Management of organizational innovation. *International Journal of Business and Social Science*, 4(1). Rennings, K. (2000): Redefining innovation: eco-innovation research and the contribution from ecological economics, *Ecological Economics*, 32(2), 319–332. https://doi.org/10.1016/S0921-8009(99)00112-3

Rennings, K. (2000). Redefining innovation—eco-innovation research and the contribution from ecological economics. *Ecological economics*, *32*(2), 319-332. https://doi.org/10.1016/S0921-8009(99)00112-3

Robertson, P. L., Casali, G. L., & Jacobson, D. (2012). Managing open incremental process innovation: absorptive capacity and distributed learning. *Research policy*, *41*(5), 822-832. https://doi.org/10.1016/j.respol.2012.02.008

Rockström, J., Steffen, W., Noone, K., Persson, Å., Chapin, F. S., Lambin, E. F., ... & Foley, J. A. (2009). A safe operating space for humanity. *nature*, *461*(7263), 472-475. https://doi.org/10.1038/461472a

Rogge, K. S., & Reichardt, K. (2016). Policy mixes for sustainability transitions: An extended concept and framework for analysis. *Research Policy*, 45(8), 1620-1635. https://doi.org/10.1016/j.respol.2016.04.004

Rohrbeck, R., Konnertz, L., & Knab, S. (2013). Collaborative business modelling for systemic and sustainability innovations. *International Journal of Technology Management* 22, 63(1-2), 4-23. https://doi.org/10.1504/IJTM.2013.055577

Schaltegger, S., Beckmann, M., & Hockerts, K. (2018). Sustainable entrepreneurship: creating environmental solutions in light of planetary boundaries. *International Journal of Entrepreneurial Venturing*, 10(1), 1-16. https://doi.org/10.1504/IJEV.2018.090990

Schaltegger, S., Loorbach, D., & Hörisch, J. (2022). Managing entrepreneurial and corporate contributions to sustainability transitions. *Business Strategy and the Environment*. https://doi.org/10.1002/bse.3080

Schaltegger, S., Hörisch, J., & Loorbach, D. (2020). Corporate and entrepreneurial contributions to sustainability transitions. *Business Strategy and the Environment*, 29(3), 1617-1618. https://doi.org/10.1002/bse.2454

Schaltegger, S., Lüdeke-Freund, F., & Hansen, E. G. (2012). Business cases for sustainability: the role of business model innovation for corporate sustainability. *International Journal of Innovation and Sustainable Development*, 6(2), 95-119. https://doi.org/10.1177/1086026616633272

Schaltegger, S., Lüdeke-Freund, F., & Hansen, E. G. (2016). Business models for sustainability: A co-evolutionary analysis of sustainable entrepreneurship, innovation, and transformation. *Organization & Environment*, 29(3), 264-289. https://doi.org/10.1177/1086026616633272

Schaltegger, S., & Wagner, M. (2011). Sustainable entrepreneurship and sustainability innovation: categories and interactions. *Business Strategy and the Environment*, 20(4), 222-237. https://doi.org/10.1002/bse.682

Scherer, A. G., & Palazzo, G. (2011). The new political role of business in a globalized world: A review of a new perspective on CSR and its implications for the firm, governance, and democracy. *Journal of Management Studies*, 48(4), 899-931. https://doi.org/10.1111/j.1467-6486.2010.00950.x

Schiederig, T., Tietze, F., & Herstatt, C. (2012). Green innovation in technology and innovation management–an exploratory literature review. *R&D Management*, 42(2), 180-192. https://doi.org/10.1111/j.1467-9310.2011.00672.x Schumpeter, J. A. (Clemence, R.V.; ed.) (1991). *Essays: On entrepreneurs, innovations, business cycles, and the evolution of capitalism.* Transaction Publishers.

SCM (2021). Fairphone's industry influence continues to grow. Supply Chain Movemement. Available at: https://www.supplychainmovement.com/fairphones-industry-influence-continues-to-grow/ [Accessed 22 Nov 2022]

Seebode, D., Jeanrenaud, S., & Bessant, J. (2012). Managing innovation for sustainability. *R&D Management*, 42(3), 195-206. https://doi.org/10.1111/j.1467-9310.2012.00678.x

Senge, P., Hamilton, H., & Kania, J. (2015). The dawn of system leadership. *Stanford Social Innovation Review*, 13(1), 27-33.

Seuring, S., & Müller, M. (2008). From a literature review to a conceptual framework for sustainable supply chain management. *Journal of Cleaner Production*, 16(15), 1699-1710. https://doi.org/10.1016/j.jclepro.2008.04.020

Seuring, S. (2004). Industrial ecology, life cycles, supply chains: differences and interrelations. *Business Strategy and the Environment*, 13(5), 306-319. https://doi.org/10.1002/bse.418

Shen, J., & Benson, J. (2016). When CSR is a social norm: How socially responsible human resource management affects employee work behavior. *Journal of Management*, 42(6), 1723-1746. https://doi.org/10.1007/s10551-017-3544-0

Silva, S., & Schaltegger, S. (2019). Social assessment and management of conflict minerals: a systematic literature review. *Sustainability Accounting, Management and Policy Journal*. 10(1), 157-182. https://doi.org/10.1108/SAMPJ-02-2018-0029

Silvestre, B. S., & Ţîrcă, D. M. (2019). Innovations for sustainable development: Moving toward a sustainable future. *Journal of cleaner production*, 208, 325-332. https://doi.org/10.1016/j.jclepro.2018.09.244

Smink, M. M.; Hekkert, M. P. & Negro, S. O. (2015). Keeping sustainable innovation on a leash? Exploring incumbents' institutional strategies. *Business Strategy and the Environment*, 24(2), 86–101. https://doi.org/10.1002/bse.1808

Smith, M. T. (2001). Eco-innovation and market transformation. *The Journal of Sustainable Product Design*, 1(1), 19-26. https://doi.org/10.1023/A:1014426028474

Stern, N., & Stern, N. H. (2007). *The economics of climate change: the Stern review*. cambridge University press.

Stubbs, W. (2019). Strategies, practices, and tensions in managing business model innovation for sustainability: The case of an Australian BCorp. *Corporate Social Responsibility and Environmental Management*, 26(5), 1063-1072. https://doi.org/10.1002/csr.1786

Tang, T., & Bhamra, T. (2008). *Changing energy consumption behaviour through sustainable product design*. In DS 48: Proceedings DESIGN 2008, the 10th International Design Conference, Dubrovnik, Croatia.

Tavassoli, S., & Karlsson, C. (2015). Persistence of various types of innovation analyzed and explained. *Research Policy*, 44(10), 1887-1901. https://doi.org/10.1016/j.respol.2015.06.001

Tian, F., Sošić, G., & Debo, L. (2019). Manufacturers' competition and cooperation in sustainability: Stable recycling alliances. *Management Science*, 65(10), 4733-4753. https://doi.org/10.1287/mnsc.2018.3178

Tidd, J. (2001). Innovation management in context: environment, organization and performance. *International Journal of Management Reviews*, 3(3), 169-183. https://doi.org/10.1111/1468-2370.00062 Thorlakson, T., de Zegher, J. F., & Lambin, E. F. (2018). Companies' contribution to sustainability through global supply chains. *Proceedings of the National Academy of Sciences*, *115*(9), 2072-2077. https://doi.org/10.1073/pnas.1716695115

Throop, W., & Mayberry, M. (2017). Leadership for the sustainability transition. *Business and Society Review*, 122(2), 221-250. https://doi.org/10.1111/basr.12116

Tony's Open Chain (2022). Who's in. Tony's Open Chain. Available at: https://www.to-nysopenchain.com/#whos-in [Accessed 22 Nov 2022]

Turnheim, B., & Geels, F. W. (2012). Regime destabilisation as the flipside of energy transitions: Lessons from the history of the British coal industry (1913–1997). *Energy Policy*, 50, 35-49. https://doi.org/10.1016/j.enpol.2012.04.060

Tushman, M., & Nadler, D. (1986). Organizing for innovation. *California Management Review*, 28(3), 74-92. https://doi.org/10.2307/41165203

Van Dam, Y. K., & De Jonge, J. (2015). The positive side of negative labelling. *Journal of Consumer Policy*, 38(1), 19-38. https://doi.org/10.1007/s10603-014-9274-0

van Lieshout, J. W., Nijhof, A. H., Naarding, G. J., & Blomme, R. J. (2021). Connecting strategic orientation, innovation strategy, and corporate sustainability: A model for sustainable development through stakeholder engagement. *Business Strategy and the Environment*, 30(8), 4068-4080. https://doi.org/10.1002/bse.2857

Van de Ven, A. H. (1995). Managing the process of organizational innovation. In G.P. Huber & W.H. Glick (Eds.). *Organizational change and redesign: Ideas and insights for improving performance* (269-294). Oxford: Oxford University Press.

Van Rijnsoever, F. J. (2022). Intermediaries for the greater good: How entrepreneurial support organizations can embed constrained sustainable development startups in entrepreneurial ecosystems. *Research Policy*, *51*(2). https://doi.org/10.1016/j.respol.2021.104438

Wagner, M. (2008). Empirical influence of environmental management on innovation: Evidence from Europe. *Ecological Economics*, 66(2-3), 392-402. https://doi.org/10.1016/j.ecolecon.2007.10.001

Walton, S. V., Handfield, R. B., & Melnyk, S. A. (1998). The green supply chain: integrating suppliers into environmental management processes. *International Journal of Purchasing and Materials Management*, 34(1), 2-11. https://doi.org/10.1111/j.1745-493X.1998.tb00042.x

Weber, K. M., & Rohracher, H. (2012). Legitimizing research, technology and innovation policies for transformative change: Combining insights from innovation systems and multi-level perspective in a comprehensive 'failures' framework. *Research Policy*, *41*(6), 1037-1047. https://doi.org/10.1016/j.respol.2011.10.015

Weissbrod, I., & Bocken, N. M. (2017). Developing sustainable business experimentation capability–A case study. *Journal of Cleaner Production*, 142, 2663-2676. https://doi.org/10.1016/j.jclepro.2016.11.009

Wernink, T., & Strahl, C. (2015). Fairphone: Sustainability from the inside-out and outside-in. In *Sustainable value chain management* (pp. 123-139). Springer, Cham.

Whiteman, G., Walker, B., & Perego, P. (2013). Planetary boundaries: Ecological foundations for corporate sustainability. *Journal of Management Studies*, 50(2), 307-336. https://doi.org/10.1111/j.1467-6486.2012.01073.x White, K., Habib, R., & Hardisty, D. J. (2019). How to SHIFT consumer behaviors to be more sustainable: A literature review and guiding framework. *Journal of Marketing*, 83(3), 22-49. https://doi.org/10.1177/0022242919825649

Yawar, S. A., & Seuring, S. (2017). Management of social issues in supply chains: a literature review exploring social issues, actions and performance outcomes. *Journal of Business Ethics*, *141*(3), 621-643. https://doi.org/10.1007/s10551-015-2719-9

Zint, M., & Frederick, R. (2001). Marketing and Advertising a 'Deep Green'Company: The Case of Patagonia, Inc. *Journal of Corporate Citizenship*, (1), 93-113. https://www.jstor.org/stable/jcorpciti.1.93

# I. APPENDIX A: Application of the framework to two practice examples

For illustration and applicability purposes of the framework, we discussed our framework drawing on two examples, a large incumbent firm, Unilever, and a small niche actor, Lichtblick.

		Scopes of market transitions				
		CORE	INPUT	OUTPUT	SUBSTITUTE	ARENA
iovations	Pro- cesses	Zero waste to landfill pro- gramme	Sustainable code of con- ducts	#1RinseIsEnou gh campaign	The Sustaina- bility Consor- tium (TSC)	Collaboration with Global Re- porting Initia- tive
Managing types of sustainability inn	Products	90% cut of phos- phates in deter- gents, 50% fewer emissions, same performance	Various tools for suppliers, e.g. cool farm tool (GHG cal- culator)	Lifebuoy soap to increase handwashing and hygiene practices in de- veloping coun- tries	Refrigerants Naturally! Campaign to encourage cli- mate friendly alternatives for refrigeration	Marine Stew- ardship Council organisation and certificate developed in cooperation with WWF
	Organization	Sustainable Liv- ing Plan encom- passes entire com- pany and its im- pact on markets and society	Training pro- grammes, e.g. "I am Walls", unemployed to entrepreneur	5 Levers for change: Pro- gram to make consumers more sustaina- ble	Initiatives at World Business Council for Sustainable De- velopment for joint industry action	Involvement and support for SDGs

Table 1 (Appendix A): Unilever's management measures for contributing to the sustainability transition of markets

Table 2 (Appendix A): Lichtblick's management measures for contributing to the sustainability transition of markets

		Scopes of market transitions				
	CORE INPUT OU			OUTPUT	SUBSTI-	ARENA
					TUTE	
ovations	Processes	"Ökoprofit" Environmental Management System	Sustainable purchasing guidelines	Business custom- ers adapt pur- chasing policies to sustainability	Transparency for feed-in tar- iffs (through le- gal measures)	Comment regu- lations / pro- posals in en- ergy markets in favour of sus- tainability
types of sustainability inn	Products	Vision includes innovative products with the objective of energy market transformation, e.g. "swarm energy ("Schwarmen- ergie")	Consumers be- come suppliers by creating their own green energy; Licht- blick provides IT solution	Consumers to suppliers by cre- ating their own green energy; Lichtblick pro- vides IT solution	Transparency for feed-in tar- iffs (through le- gal measures)	Collaboration with municipal- ity to offer green power for e-cars
Managing	Organization		investments in new wind, so- lar	Inform consumer and offer calcula- tor for green power	called for trans- parency of feed-in tariffs with court ac- tion at federal court	2005 - increase transparency of feed-in tariffs at Federal Su- preme Court

Contri- bution pathway (CP)	Example company	Contribution pathway (CP)	Company contribu- tion (CC)	Measurable market impact
CP1	<b>Green Mountain</b> <b>Power (GMP):</b> Business transfor- mation towards green energy, as discussed in Throop and May- berry (2017)	CORE: Business model innovation to wards renewable energy (Orsted GMP) & smart-grid technology (GMP <i>Potential further impacts:</i> <i>ARENA</i> : Nationwide system of electric vehicle charging stations <i>SUBSTITUTE</i> : Inspiration for other con ventional energy providers to orient to wards sustainability	- CC <sup>1</sup>	Green Mountain Power serves ca. 270,000 customers, with 90% carbon-free power (More- house, 2019). Green Mountain Power has been named to TIME's List of the TIME100 Most Influential Companies (GMP, 2022).
CP2	<b>Fairphone:</b> Introduction of fairly produced modular phones, discussed in Wernink and Strahl (2015); Nylund et al. (2021)	INPUT: Increasing socio-ecologica standards in smartphone supply chains OUTPUT: Increased demand for fairly produced phones SUBSTITUTE: Pressure on other phone manufacturers to increase supply chain transparency and to consider modularity	<b>I</b> CC <sup>3</sup> s	Doubling of sales in 2020, in- creasing influence in smartphone market (Fairphone, 2020). Fairphone sold nearly 95,000 phones in 2020 com- pared to almost 59,000 in 2019, representing a sales increase of 76%. (SCM, 2021)
CP3	Patagonia: as discussed in Zint and Frederick (2001) Chouinard and Stan- ley (2012); O'Rourke and Strand (2017); Rattalino (2018)	OUTPUT: Consumer consider slow fashion, responsible consumption and sufficiency ARENA: Increased social discourse about throwaway culture and sustainable clothing	v CC <sup>2</sup>	10% market share in the out- door apparel market, making it the number 1 market share holder. 10% year-on-year growth rate. (Cascade, 2021)
CP4	<b>Tony's Chocolone-</b> ley: Co-creation of market for 100% sal- very free chocolate with other choclote producers; discussed in Blom et al. (2014)	SUBSTITUTE: Co-creation of a 100% slavery-free chocolate industry OUTPUT: Raising awareness on slavery in chocolate value chains; Encouraging consumers to make responsible consump tion choices	$\frac{1}{2}$	In 2019, Tony's Chocolonely was the biggest chocolate brand in the Netherlands with a mar- ket share of around 19% and growth of 27% compared to the previous year (Kraaijenbrink, 2019). The open chain program has four big allies until now, in- cluding ALDI and Ben & Jer- ries (Tony's Open Chain, 2022)
CP5	Unilever: Co-development of the Marine Steward- ship Council (MSC) certification with the WWF, discussed in Ponte (2012)	ARENA: Co-creating a mainstream market for sustainable fish by intro- ducing new standards, a private gov- ernance institution and label CORE: Supplier code of conduct, sus- tainability strategy INPUT: Increasing the number of sus- tainable certified fisheries OUTPUT: Increasing consumer aware- ness of sustainability in fish purchases SUBSTITUTE: Increasing amount of sustainable fish at large food service and	CC <sup>5</sup>	14% of marine catch have been certified with the MSC Fisher- ies Standards according to the organization itself. => Argua- bly a contribution of 14% after 25 years (Arton et al., 2020; MSC, 2022)

II.	<b>APPENDIX B:</b>	Practice examples :	for contribution	pathways

# **D.** Processes of exploration and reorientation

Hübel, C., Weissbrod, I. & Schaltegger, S. (2022): Strategic alliances for corporate sustainability innovation: The 'how' and 'when' of learning processes. Long Range Planning, 102200. https://doi.org/10.1016/j.lrp.2022.102200

Hübel, C. (2022): Entrepreneurship-driven organizational transformation for sustainability: A sensemaking lens, Journal of Organizational Change Management, 35(1), 240-256. https://doi.org/10.1108/JOCM-03-2021-0067

Long Range Planning xxx (xxxx) xxx



Contents lists available at ScienceDirect

# Long Range Planning



journal homepage: www.elsevier.com/locate/lrp

# Strategic alliances for corporate sustainability innovation: The 'how' and 'when' of learning processes

Charlott Hübel<sup>\*</sup>, Ilka Weissbrod, Stefan Schaltegger

Centre for Sustainability Management (CSM), Leuphana Universität Lüneburg Universitätsallee 1, 21335 Lüneburg, Germany

### ABSTRACT

Mounting sustainability pressures challenge established firms to engage with sustainability innovations, which are often introduced by startups. Research on alliance learning has established the potential of learning from startups to advance corporate innovation. Here, scholars have outlined alliance learning processes and outcomes and have distinguished learning *about* and learning *from* alliance partners as two key learning types. The saliency of learning *from* the operational alliance process is stressed. To date, however, no study has investigated alliance learning processes and outcomes for sustainability innovations. This is despite the fact that sustainability research suggests learning processes in the sustainability context have a distinct nature. This study addresses this research gap by analyzing the sustainability-specific learning processes and outcomes of a large European meat producer and wholesaler with a turnover of \$2.7bn in the fiscal year 2019/2020. The firm formed alliances with nine startups for sustainable plant, insect-based and cell-based protein solutions. Our analysis (1) identifies three distinct characteristics of sustainability-related alliance learning processes and outcomes, and (2) specifies the temporal occurrence and outcomes of learning types in alliance learning phases. In contrast to findings of prior research, our study reveals that learning *about* alliance learning outcomes may support an established firm's contrability-oriented alliance learning process. In addition, the findings highlight that alliance learning outcomes may support an established firm's contribution to the sustainability transformation of mass markets.

#### 1. Introduction

Increasing societal and environmental challenges such as poverty and climate change have moved sustainability - i.e. the aspiration to "safeguard intergenerational equity" (Bansal and DesJardine, 2014: 70) in an economy operating within the space of planetary boundaries (Rockström et al., 2009) - from a niche to a mainstream issue (Bocken and Geradts, 2020). With mounting sustainability regulations and stakeholder pressure, established firms in different industries are challenged to analyze their strategies and practices with regard to sustainability requirements (Schaltegger and Hörisch, 2017). Firms can, and need to, respond to this new "sustainability imperative" (Lubin and Esty, 2010: 2) by engaging with sustainability innovations, often involving fundamental reconfigurations of products and processes so that they not only target economic but also environmental and/or social benefits (Arnold and Hockerts, 2011; Klewitz and Hansen, 2014). In doing so, established firms can secure their existence (Schaltegger and Hörisch, 2017), gain competitive advantage (Hall and Vredenburg, 2004; Hermundsdottir and Aspelund, 2021) and, at the same time, contribute to the transformation of markets and industries toward sustainable development (Hockerts and Wüstenhagen, 2010; Schaltegger and Wagner, 2011).

Established firms, however, frequently face difficulties in developing and adopting sustainability innovations (Bocken and Geradts, 2020). Such innovations are different from conventional innovations (Kennedy et al., 2017; Weissbrod, 2019), as they come with

\* Corresponding author.

https://doi.org/10.1016/j.lrp.2022.102200

Received 28 May 2020; Received in revised form 5 February 2022; Accepted 25 February 2022

Available online 2 March 2022

E-mail address: charlott.huebel@leuphana.de (C. Hübel).

<sup>0024-6301/© 2022</sup> The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

#### C. Hübel et al.

directional risks<sup>1</sup> (Hansen et al., 2009), involve different externalities in innovation and diffusion phases (Rennings, 2000) and require interdisciplinary learning toward systems thinking (Adams et al., 2016; Wals and Corcoran, 2012). While established firms struggle to implement sustainability innovations, a growing number of sustainability startups has started to introduce sustainable product innovations to the market (Hockerts and Wüstenhagen, 2010). Sustainability startups differ from conventional startups by integrating social and environmental value creation into their core business (Dean and McMullen, 2007) and following a values-oriented (Parrish, 2010), as well as a stakeholder-oriented (Freudenreich et al., 2019) approach.

In studies of how established firms advance innovation, strategic management research emphasizes the potential of strategic alliances. Leading scholars argue that strategic alliances can drive corporate innovation through organizational learning (Inkpen, 2008; Inkpen and Dinur, 1998; Kavusan et al., 2016; Schildt et al., 2012). Researchers have studied development and interaction patterns of alliance learning (Bingham et al., 2015; Davis and Eisenhardt, 2011; Doz, 1996; Kale and Singh, 2007), alliance learning types (Das and Kumar, 2007; Inkpen and Currall, 2004; Inkpen and Tsang, 2007) and alliance learning outcomes (Doz, 1996; Kavusan et al., 2016; Liu and Lui, 2020). Regarding learning types, initial research suggests that learning *about* alliance partners may dominate in the early phases of an alliance, whereas learning *from* dominates in the operational alliance phases (Das and Kumar, 2007). Empirical research confirms that alliances with startups hold high learning potential for established firms (Alvarez and Barney, 2001; Gopalakrishnan et al., 2008; McCutchen and Swamidass, 2004; Mittra, 2007; Weiblen and Chesbrough, 2015; Rothaermel, 2001). At the same time, sustainability research shows that sustainability startups possess sustainability innovation capabilities related to a sustainability-specific value proposition, creation and delivery, which established firms often lack (Keskin et al., 2020; Schaltegger et al., 2016). Learning from sustainability startups in the context of strategic alliances, therefore, might help established firms to develop capabilities for sustainability innovation.

The distinct nature of sustainability innovation and startups might suggest that sustainability-oriented alliance learning processes differ from those in conventional alliances. To date, however, no study has empirically investigated alliance learning processes and outcomes for sustainability innovations. Our study addresses this research gap and is guided by the research question: How do learning processes in strategic alliances between an established firm and sustainability startups unfold and influence sustainability innovation in the established firm? To answer this research question, we develop a *process model for alliance-driven learning for corporate sustainability innovation*, which depicts both the processes and outcomes of alliance learning from the perspective of an established firm. The analysis aims to gain new insights into the temporal occurrence and outcomes of two learning types (i.e. *from* and *about*) in alliance processes in a highly topical context. Using a qualitative methodology, we investigate processual dynamics in the established firm with regard to its alliances with sustainability startups, depicting how interactions evolve over time (Langley et al., 2013). An exploratory single case study design (Yin, 2009) was adopted to achieve an in-depth understanding of the interorganizational and personal interactions inherent to alliance learning.

Our findings contribute to the strategic management literature by establishing three distinct characteristics of alliance learning processes and outcomes in the context of sustainability innovations. This is done by specifying the temporal occurrence and outcomes of the two established learning types *about* and *from* and by describing novel alliance experiences in three distinct alliance learning phases.

Our review of literature on alliance learning processes and alliance learning outcomes is followed by a description of the methodology we adopted for data collection and analysis. After presenting the findings and the discussion of the results, we develop the process model. Our conclusions address areas for further research.

#### 1.1. Literature review

Sustainability research claims that for sustainability innovations to address complex global challenges, they require interdisciplinary learning in collaborative environments (Bradbury-Huang et al., 2010; Mülling Neutzling et al., 2018; Rohrbeck et al., 2013). Strategic management research has identified strategic alliances as a valuable collective learning environment (Doz, 1996; Inkpen, 1998; Inkpen and Tsang, 2007; Kale and Singh, 2007) that can spur corporate innovation (Inkpen, 2008; Inkpen and Dinur, 1998; Kavusan et al., 2016; Mittra, 2007; Rothaermel, 2001; Schildt et al., 2012; Weiblen and Chesbrough, 2015). In strategic alliances, partners "exchange, share or co-develop [ ...] resources, competencies and capabilities" (Russo and Cesarani, 2017: 1) and seek "solutions to long-term needs, rather than temporary fixes" (Chen and Chen, 2002: 1008). To answer this study's research question of *how* learning in strategic alliances advances corporate sustainability innovation therefore requires a deeper understanding of alliance learning processes and outcomes.

#### 1.2. Alliance learning processes

The literature provides two complementary perspectives on alliance processes. The first relates to the overall alliance life cycle and is broken down into a linear sequence of partner selection, partner management and partner termination (Heimeriks et al., 2015) or formation, operation and outcome (Das and Teng, 2002; Russo and Cesarani, 2017). The second perspective, on which our study focuses, zooms in on the operational phase of an alliance starting with its formation. In this perspective, processes occur in a dynamic, iterative manner and are strongly related to learning (Inkpen and Tsang, 2007; Kumar and Nti, 1998). The alliance learning process

<sup>&</sup>lt;sup>1</sup> Directional risk is the risk of not knowing beforehand the direction of an innovation's actual effects on sustainable development.

#### C. Hübel et al.

generally involves articulation, codification, sharing and internalization (Kale and Singh, 2007). Doz (1996) and Doz and Hamel (1998) introduce an evolutionary view of alliance learning, as based on earlier work by Ring and Van de Ven (1992). This alliance learning theory suggests that successful alliances move through several cycles of learning, re-evaluation and readjustment, with each learning cycle making the alliance more efficient over time. Whether learning occurs and whether the alliance evolves favorably is determined by the initial conditions of the alliance, including partner expectations and strategies as well as absorptive capacities (Doz, 1996; Kumar and Nti, 1998). As the learning process evolves, alliance conditions change and shape subsequent behavior. Researchers argue that a favorable evolutionary path requires that partners are sufficiently compatible for learning to occur (Doz, 1996; Hamel, 1991; Lane and Lubatkin, 1998). Research on alliances between large and small firms suggests that due to differences in resource portfolios, market experience and cultures such alliances can involve unequal processes to the detriment of the smaller firm (Barabel et al., 2014; Doz, 1988; Minshall et al., 2010; Pérez et al., 2012; Prashantham and Birkinshaw, 2008).

Strategic management scholars have distinguished two main types of learning that occur in the alliance process: learning *about* the alliance partner and learning *from* the alliance partner (Doz and Hamel, 1998; Khanna et al., 1998; Inkpen and Currall, 2004; Inkpen and Tsang, 2007). Doz's (1996) seminal work on alliance learning has largely focused on the first type of learning, learning *about*, which is partner-specific learning that occurs along five dimensions: environment, goals, skills, task and process. This type of learning focuses on the management of individual alliances and is linked to the concept of alliance management learning (Ireland et al., 2002). Learning *from*, which is also referred to as content learning (Das and Kumar, 2007), involves knowledge that has value to a firm outside the scope of the alliance, as firms can internalize knowledge to enhance their own operations (Khanna et al., 1998; Inkpen and Currall, 2004). Individual-level components, including personal interactions, are key building blocks for developing such knowledge integration capabilities (Felin et al., 2012).

Some studies explicitly link learning to different phases in the alliance life cycle (Das and Kumar, 2007; Heimeriks et al., 2015; Schildt et al., 2012). In their conceptual paper, Das and Kumar (2007) suggest that the partner selection phase is dominated by learning *about* the partner, while the partner management phase is increasingly dominated by learning *from* the partner. This is in line with research on interorganizational trust, highlighting that partner familiarity and understanding is needed for knowledge transfer to occur (Inkpen and Currall, 2004; Nielsen, 2005; Nielsen and Nielsen, 2009).

### 2. Alliance learning outcomes

Alliance learning research suggests that the two types of learning can result in different outcomes for the alliance and its members. Learning *about* the alliance partner helps firms to build trust, improve coordination and increase overall alliance performance (Inkpen and Currall, 2004; Khanna et al., 1998; Liu and Lui, 2020). In particular, positive experiences related to gaining knowledge *about* the partners increase a firm's motivation to continue the alliance (Das and Kumar, 2007). It has also been suggested that learning *about* an alliance partner can influence decisions on entering into and learning in further alliances (Gulati et al., 2009; Heimeriks et al., 2007; Heimeriks and Duysters, 2006). The more alliances a firm enters and the more diverse the partner firms are, the more learning can occur (Cui and O'Connor, 2012; Jiang et al., 2010). By contrast, Grant and Baden-Fuller (2004) propose with their knowledge accessing theory of strategic alliances that the more firms focus on only accessing knowledge, i.e., learning *about*, without the intention of integrating and applying that knowledge, the more alliances they can enter. Learning *about* therefore does not only relate to advancing an individual alliance, but also to whether a firm enters further alliances. With increasing experience with alliances, firms develop an alliance learning capability, which results in increased learning and alliance success over time (Anand and Khanna, 2000; Kale and Singh, 2007; Schilke and Goerzen, 2010).

The impact of learning *from* goes beyond improved alliance performance and can also substantially advance a firm's internal innovation. The link between learning success and corporate innovation has received substantial attention in the strategic management literature (e.g. Berghman et al., 2013; Cohen and Levinthal, 1990; Jiménez-Jiménez and Sanz-Valle, 2011). With regard to learning in the context of alliances, it is argued that a firm's ability to absorb and exploit a partner's knowledge is particularly high in alliances with a high degree of technological overlap or a complementary knowledge base (Kavusan et al., 2016; Schildt et al., 2012; Shenkar and Li, 1999). Subramanian and Soh (2017) further argue that learning *from* a wider range of alliance partners can increase the breadth of resulting innovation produced by the focal firm. However, learning *from* can also have a negative impact on alliances or even lead to early alliance termination, when one partner "outlearns" the other, takes advantage of its increased bargaining power and starts using its acquired knowledge competitively (Hamel, 1991; Inkpen and Beamish, 1997; Inkpen and Currall, 2004; Van de Ven and Walker, 1984). Research indicates that in alliances between large firms and startups the large firm is much more likely to win the learning race, rendering these types of partnerships difficult to maintain (Alvarez and Barney, 2001). Winning the learning race, however, requires that the large firm is able to overcome learning challenges attributable to a potential lack of management commitment and core business-focused corporate structures (Kohler, 2016).

#### 2.1. Research gap

Extant research has neglected to explore alliance learning processes and outcomes in the context of sustainability innovations. Learning for sustainability innovations might be different, as it is influenced by the degree of commitment to a firm's sustainability strategy (Kennedy et al., 2017) and it involves further requirements, including the unlearning of prior knowledge that contradicts sustainability principles (Adams et al., 2016; Bossink, 2007; Magnusson et al., 2003). With their purpose to contribute to social and environmental goals beyond organizational boundaries, sustainability innovations require learning towards systems and about empathetic thinking (Wals and Corcoran, 2012) as well as learning about the interrelationships between organization, society and the

#### C. Hübel et al.

environment (Johnson and Schaltegger, 2020). In addition, differences in motivation, values and ideologies between mainstream and sustainability-oriented actors may complicate or prolong alliance learning processes, as actors tend to be resistant to learning from one another (London et al., 2004; O'Mahony and Bechky, 2008; Rondinelli and London, 2003; Stern and Hicks, 2000). Alliance learning outcomes might differ too, as sustainability innovations can support the sustainability transformation of whole markets and industries (Hockerts and Wüstenhagen, 2010; Schaltegger et al., 2016; Schaltegger and Wagner, 2011).

The investigation of sustainability-related alliance learning processes addresses the outlined research gap. Fig. 1 synthesizes previous research findings, illustrates the main research gap and highlights three specific aspects that have not been addressed by research, so far.

This study's in-depth analysis attends to all three of the outlined aspects, thereby contributing to alliance learning research. First, the investigation may provide empirical insights into how and when established firms learn *about* and *from* sustainability startups, thereby confirming or deviating from Das and Kumar's (2007) suggestion that the alliance process is increasingly dominated by learning *from*. Second, the investigation may shed light on how differences in the temporal occurrence and purpose of learning types influence experiences in the sustainability-related alliance process and, vice versa, affect the evolutionary path of alliances. Lastly, the investigation may give indications of broader sustainability-related alliance learning outcomes beyond organizational boundaries – an aspect that has not been investigated so far, neither in the sustainability nor in the alliance learning literature.

### 3. Methods

#### 3.1. Empirical approach

In order to investigate the relationship characteristics of alliance learning for corporate sustainability innovation, we employ a qualitative, exploratory case study design (Yin, 2009). The qualitative study design allows for an in-depth analysis of alliance learning processes in a unique setting, "capturing the nuances of processes in and around organizations" (Langley et al., 2013: 10). Given the study's focus on corporate innovation, the analysis adopts the perspective of the established firm, i.e. its learning cycles and outcomes as partner-specific learning processes. We investigated innovation processes in a large established European meat producer and wholesaler with a turnover of \$2.7bn in the fiscal year 2019/2020. This firm, anonymized as BIGMEAT, entered nine strategic sustainability-oriented alliances with international startup firms for alternative protein products. Given the social discourse on meat consumption and production (Sapontzis, 2004) and the radicalness of new meat replacement innovations (Shapiro, 2018), alliances between a meat producing firm and alternative protein startups constitute an intriguing and highly topical research case.

BIGMEAT is family owned and run and the alliances are part of the firm's strategy to diversify its product portfolio toward meat alternatives and thereby future-proof its business. The startup alliance partners offered ethical and ecological food tech innovations, including plant-based, insect-based, cell-based and 3D-printed meat analogues. All of the startups aimed to make their products available to the European mass market. BIGMEAT supported the alternative protein product startups by providing finance, infrastructure and market know-how, while the startups granted BIGMEAT access to state-of-the-art protein product innovations. Nine



Fig. 1. Synthesis of previous alliance learning research and identified research gap.
### C. Hübel et al.

strategic alliances were established between February 2016 and mid 2019 (Table 1) in the form of strategic investments (S, B, E) and/ or distribution partnerships (A, C, D, F, G, H). All but one alliance (G) are still ongoing at the time of the completion of the research.

The first alliance, Alliance S, remained undisclosed to the public due to reputational concerns voiced by both partners. While the owner of Startup S was concerned about potential negative responses from its vegan customer base, BIGMEAT's management had not yet formulated and communicated their strategic intentions to become involved in the alternative protein field at the point of alliance formation.

### 3.2. Research process

We designed the case research to ensure internal validity, construct validity, and reliability (Gibbert et al., 2008). The research process (Fig. 2) was abductive (Dubois and Gadde, 2002) and alternated between the literature, empirical findings and data analysis (Thomas, 2010).

The link between alliances and corporate innovation (e.g. Weiblen and Chesbrough, 2015) and sustainability-oriented interactions between startups and established firms (e.g. Hockerts and Wüstenhagen, 2010) provided the foundation for this study. Our empirical observations of alliance learning processes, however, could not be fully explained in this initial theoretical framework. We therefore engaged in an iterative abductive process of "theory matching," resulting in a revised theoretical framework that incorporated constructs of alliance learning (Kovács and Spens, 2005). By alternating between data collection, data analysis and theory matching, we gained an increasingly comprehensive understanding of the phenomenon of alliance-driven learning for corporate sustainability innovation. This deeper understanding encouraged us to develop a new process model.

### 4. Data collection and preparation

The lead researcher sought to collect salient data without influencing the alliance process or interviewees using a variety of data collection methods. First, the data collection included semi-structured interviews (Wengraf, 2001) as well as informal conversational interviews (Turner, 2010) with managers from BIGMEAT and the startup partners (see Appendix A, Tables A.1 and A.2). For those interviews that were conducted in German, relevant text sections were translated by the lead researcher. Top managers interviewed included BIGMEAT's CEO and Chairman, the Board Lead for Alternative Proteins (previous Head of M&A), and the CEO of two BIGMEAT subsidiaries. Middle managers included managers from sales, marketing, project management, product development, communication, sustainability management and the new head for the alternative protein unit. The interviews with startups were conducted taking part in the ongoing activities of the participants while consciously observing and, after the interaction, recording the observations made. In doing so, we aimed for a balance between observation and participation (DeWalt and DeWalt, 2002). Third, we collected data in the form of documents (Prior, 2003). Each data source fulfilled a different purpose during analysis (Table 2). In total, our database comprises 40 formal and informal interviews, observations at 11 events and 71 documents.

As proposed by Dwyer and Buckle (2009), we adopted an insider-outsider perspective to take into account the complexity of human experiences and the impossibility for qualitative researchers to remain completely separate from the study. The lead researcher travelled to the company headquarters and participated in industry events and conferences while the co-researchers acted as 'sparring partners' to reflect on the observations at a critical distance. Interactions with study participants took place in work settings, at meetings and at industry events. The Board Lead for Alternative Proteins was the main point of contact during the 18-month long investigation.

Data was collected from August 2018 until February 2020. To reconstruct the strategic decisions and alliance processes that occurred prior to August 2018, interviewees were asked to recall the alliance process back to February 2016 and these recollections were used as retrospective data (Pettigrew, 1990). The final three interviews with top and middle management at BIGMEAT (Interviews 10; 11; 12) corroborated the data findings and were used to seek final clarification of some factual information (e.g. people involved, timelines). Interviews were transcribed ad verbatim and the observational data, including interviews, were uniformly protocolled (McLellan et al., 2003).

Table 1	
The nine strategic alliances of the BIGMEAT research	case.

Alliance	Core product value proposition of startup	Alliance formed	Alliance status
S	Plant-based protein	Beginning of 2016	Commercialization active
Α	Cell-based protein	End of 2017	Commercialization planned for 2022/23
В	Insect-based protein	Beginning of 2018	Commercialization active
С	Plant-based protein	April 2018	Commercialization active
D	Plant-based protein	July 2018	Commercialization planned for 2021
Е	Insect-based protein	July 2018	Commercialization planned for 2021
F	Plant-based and cell-based protein	November 2018	Commercialization planned for 2021
G	Plant-based protein	November 2018	Planned commercialization stopped
		Terminated January 2020	
н	Plant-based protein and technology platform	Mid-2019	Commercialization planned for 2021

#### Long Range Planning xxx (xxxx) xxx



Fig. 2. Abductive research process used in the case study (adapted from Kovács and Spens, 2005, 139).

Table 2							
Data sources	and	their	use	in	the	analy	sis.

Data types	Sources	Use in the analysis
Semi-structured interviews at BIGMEAT	12 formal interviews with top and middle managers (see Appendix A, Table A1)	<ul> <li>Reconstruct the process of alliance development</li> <li>Investigate the motivations behind alliance formation and learning intentions</li> <li>Examine changes over time/whether and how organizational learning occurred</li> </ul>
Semi-structured interviews at startups	8 formal interviews with founders/top managers (see Appendix A, Table A1)	<ul> <li>Reconstruct the process of alliance formation and development</li> <li>Investigate startup motivations</li> <li>Corroborate statements about alliance formation and processes by BIGMEAT managers</li> </ul>
Informal interviews	20 informal interviews at BIGMEAT with top and middle management (see Appendix A, Table A2)	<ul> <li>Supplement formal interviews</li> <li>Gain deeper insights into personal motivations and learning over time</li> </ul>
Balanced participatory observation	6 industry events 4 meetings 1 press event	<ul> <li>Investigate the relationship and fit of partners</li> <li>Investigate the direct collaborative learning processes</li> <li>Analyze the external communication of the alliances</li> </ul>
Publicly available documents	30 firm reports & press releases 29 media reports 4 firm magazines	<ul> <li>Examine external communication of alliances</li> <li>Examine external perceptions (e.g. media coverage) of alliances and its implications</li> </ul>
Internal documents	8 presentations and written communication	• Examine internal communication of alliances and related learning processes and outcomes

### 4.1. Data analysis

The data analysis was composed of three steps. Step one used the qualitative data analysis software program MAXQDA (see Appendix B, Figure B1) to conduct a coding process. Table 3 shows the resulting 106 first-order descriptive, process and emotion codes (Saldaña, 2015), which were grouped into 8 second-order codes. All first-order and second-order codes were compiled in a codebook (DeCuir-Gunby et al., 2011; Saldaña, 2015) and were discussed between two researchers.

In step two of the data analysis, we marked codes and specific text sections according to the alliance the statements referred to. The alliance-specific text sections are organized in matrix tables (Miles et al., 2014). The process was informed by alliance learning research (Doz, 1996; Doz and Hamel, 1998; Inkpen and Currall, 2004) and identified the learning cycles (i.e. cycles of learning, re-evaluation and readjustment), including the temporal occurrence of learning types (i.e. learning *about* and learning *from*), and their respective outcomes (columns in Table 4). We systematically identified learning *about* and learning *from* alliances in the collected data set (rows in Table 4). Data sections referring to one or more of Doz's learning dimensions, i.e. the partners' environment and market context, the partners' goals and motives, the partners' skills, the alliance task and the alliance process, all indicated learning *about*. Statements were classified as instances of learning *from* if the data referred to knowledge from the partners with a clear intention to internalize and use the knowledge, unconnected to the alliance process (Inkpen and Currall, 2004). Both learning processes and outcomes were analyzed from the perspective of BIGMEAT in order to identify the scope and dominant type of learning at BIGMEAT. The process generated nine matrix tables, one for each alliance. Table 4 shows the simplified matrix table of Alliance F.

### C. Hübel et al.

### Long Range Planning xxx (xxxx) xxx

### Table 3

First-order and second-order codes identified in qualitative data.

First-order codes		Second-order codes
• Openness	Careful decision-making	Established firm characteristics & strategy
Pioneer thinking	Family business	
Growth through diversity	Market orientation	
Traditionalism	Corporate identity	
Sustainability views	• Uniqueness	
• Future of meat business	• Strategy	
Customer orientation	Personal conviction	
Finduct focus     Finduct focus	Compatitive advantage	
• Employee orientation	Authenticity	
	Profit orientation	
	Collaboration focus	
Fascination/hype	Skepticism	Established firm initial conditions
High expertise	No defined process	
<ul> <li>Seriousness/long-term thinking</li> </ul>	Market opportunity	
Vegan vs. meat	Trust in leadership	
Low awareness	Existential fear	
<ul> <li>Perception of startup</li> </ul>	Acceptance	
New knowledge	<ul> <li>Ambivalent work tasks</li> </ul>	Established firm internal processes
Generation 2.0	<ul> <li>Negative experience</li> </ul>	
Structural change	<ul> <li>Limits to internal change</li> </ul>	
<ul> <li>New sustainability perspective</li> </ul>	<ul> <li>Disagreements/discussions</li> </ul>	
<ul> <li>Positive feedback</li> </ul>	<ul> <li>Employee encouragement</li> </ul>	
<ul> <li>More involvement</li> </ul>	<ul> <li>Staff shortages</li> </ul>	
New innovativeness	Company image	
Opportunities for meat business	Strategy development	
Risks for meat business	• Newness	
Positive experience	Product development challenge	
	Learning challenge	
Planning & management	• Learning chanelige	NewCo/Accelerator Platform
Structural integration		
• Enthusiasm		
Compatibility	Goals	Alliance conditions
Pragmatism	Differences	
<ul> <li>Personal contact</li> </ul>	<ul> <li>Philosophy &amp; values</li> </ul>	
Partner fit	<ul> <li>Unstructured process</li> </ul>	
<ul> <li>Type of partnership</li> </ul>	Competition	
Challenges	<ul> <li>Startup dominance</li> </ul>	Alliance processes
<ul> <li>Established firm support</li> </ul>	<ul> <li>Time-consuming</li> </ul>	
Open communication	<ul> <li>Mutual partnership</li> </ul>	
<ul> <li>Further alliance potential</li> </ul>	Alliance re-evaluation	
	Functioning collaboration	
	Alliance synergies	
Strategy/approach	New knowledge	Startup perspective
Motivation for partnership	Market orientation	
Praising established firm	Independence     Endependence	
Sustainability inission     Business motivation	External recuback     Expertise	
Alliance goal	<ul> <li>Experiesc</li> <li>Established firm products</li> </ul>	
Initial concerns	Technological innovation	
Personal story	Growth	
- 1 croonar story	New market access	
	Business challenges	
	Welcoming partner learning	
Industry change	Political awareness	External conditions
Startup hype	Stakeholder pressure	
Competition	Decreasing hype	
Negative reputation	Local market challenges	
Uncertainties	5	

Note: The code book and related coding process were the basis for creating nine matrix tables, which provide insights on learning processes and outcomes of each alliance.

### C. Hübel et al.

### Table 4

Matrix table of BIGMEAT's learning processes and learning outcomes in Alliance F.

ALLIANCE		Learning processes at BIGMEAT (learning cycles)				
F		Initial conditions	Learning	Re-evaluation	Readjustment	
Learning outcomes at BIGMEAT	About	Personal fit:Alignment with startup management, c ompatibility of competencies	Local market challenges: Regulatory barriers	Dealing with challenges: Open communication, more support for startup	Goal readjustment: Postponing market launch, adjusting alliance management	
	From	Fascination with R&D process, awareness of o pportunities for core business	New R&D knowledge for development of plant-based protein products, potential for c ompetitive advantage New sustainability perspectives related to global food security	<i>Re-orientation:</i> Desire to focus on own strength	New innovativeness: Accelerating development of own plant-based protein products	

In the third and final data analysis step we reconstructed the timeline of alliance learning and linked it to BIGMEAT's innovation process. To increase the findings' validity, we conducted three triangulation interviews (Interviews 9; 10; 11) as well as a presentation and group meeting with four key BIGMEAT managers toward the end of data analysis in late February 2020. These measures did not result in additional second-order codes of the previously collected data on the alliance learning process.

#### 4.1.1. Findings

The findings on BIGMEAT's alliance learning are presented in three parts. The first two parts cover alliance learning processes and alliance learning outcomes, with tables summarizing the key data findings and relating them to prior research. The third part merges the key findings into a process model for alliance-driven learning for corporate sustainability innovation.

### 4.2. BIGMEAT's alliance learning processes

The data indicates that while BIGMEAT learned extensively in its alliances with sustainability startups, the learning scope and intensity varied. This variation in learning over time is seen in the different temporal occurrence of learning *about* and learning *from*, impacting the frequency of learning cycles, as well as in the different experiences for BIGMEAT in distinct learning phases. Fig. 3 depicts the learning cycles for BIGMEAT in each of the nine alliances as boxes encompassing instances of learning *about* and/or *from*, re-evaluation and readjustment in the alliance learning phases.

Occurrence of learning types: The learning cycles of BIGMEAT include instances of both learning *about* and learning *from*. Learning *about* occurred in all nine alliances and continuously throughout the alliance processes. This type of learning was the main trigger for learning cycles, confirming the importance of partner familiarity and trust on the scope of alliance learning (Inkpen and Currall, 2004; Nielsen, 2005; Nielsen and Nielsen, 2009). In total, five alliances (Alliances B, C, D, F and G) went through one or two full cycles of learning, re-evaluation and readjustment, with Alliance C being the only alliance that saw the completion of two full learning cycles. Accordingly, Alliance C offered extensive learning opportunities to BIGMEAT managers (Interviews 5; 8; 10). For middle management, including sales (Interviews 5; 7; 8) and marketing (Interview 12), it was the first alliance the managers were involved in and the one that required the most intensive contact. Late 2019 saw a cluster of parallel learning cycles, pointing to an alliance learning capability developed by BIGMEAT (Anand and Khanna, 2000; Kale and Singh, 2007; Schilke and Goerzen, 2010). In line with this, BIGMEAT managers perceived learning to be particularly high at later phases of the alliance processes (Interviews 4; 9; 10).

The data indicates that learning *from* only occurred after extensive learning *about* had taken place and that it only occurred in the early stages of Alliances C, D and F. Interestingly, learning *from* was limited to those three alliances and did not reoccur in later stages of the alliance process. BIGMEAT's top management offered explanations for the lack of learning from Startups B and G, who they considered "*outliers*" in terms of learning and integration opportunities due to startup focus on niche products such as insect-based meat analogues or plant-based mozzarella (Board discussion February 2020; Interviews 10; 8).

No.	Finding	Main topic	Relation to prior research
1.1	While the established firm continuously learned <i>about</i> all partners, the occurrence of learning <i>from</i> was limited to three instances and partners, and was not relevant in later alliance phases.	The diverging temporal occurrence of learning <i>about</i> and learning <i>from</i> in the alliance process	The finding contrasts with earlier research suggesting that the partner management phase would be dominated by learning <i>from</i> (Das and Kumar, 2007).

*Three learning phases:* The data findings suggest that BIGMEAT's overall alliance learning process can be separated into three distinct phases (bottom, Fig. 3). The first phase involved overcoming a perceived *"inhibition threshold"* to learning (Interview 10). Before entering the first alliance, Alliance S, the startups were perceived as *"ideological stereotype images of enemies"* (Informal interview 1) and reservations existed on both sides about collaboration between a meat firm and a vegan firm (Interviews 1; 3; 10; 13). The differences in values and missions (see Appendix C) strongly impacted the early alliance phase (Interviews 1; 10; 13). Barriers were



Fig. 3. BIGMEAT timeline of 'learning from' and 'learning about' along distinct alliance learning phases.

perceived as high due to stakeholder expectations, including employees and customers, which is also why the first alliance was never disclosed to the public (Interviews 1; 10; 13). However, with Alliance S, BIGMEAT managers recognized the feasibility of collaborating with and learning from vegan sustainability startups: *"Everyone can learn from the other. It's not that one is bad and the other one is good. Let's just try this experiment"* (Interview 10). Learning was especially facilitated by the pragmatism expressed by both parties, i.e. their ability to look beyond differences in values and missions and focus on the joint alliance objective (Interviews 1; 10; 13; 14; 16; 19). Alliance S created the conditions necessary for forming Alliances A, B and C and thus for entering a second phase marked by positive alliance experiences (Interview 10).

The second phase included acquiring valuable knowledge for BIGMEAT *about* and *from* the alliance partners. Some of these positive experiences continued to occur further in the process whenever BIGMEAT entered alliances with new alternative protein startup partners. In initial positive learning experiences with startup partners, BIGMEAT managers learned *about* the pragmatic approach and collaborative potential of vegan companies and startups (Interview 10; 6), *about* the market potential of *"second generation"*.<sup>2</sup> plant-based products (Interview 6), *about* the products' compatibility with BIGMEAT's existing production know-how (Interviews 1; 8) and customer base, the so called *"flexitarians"* <sup>3</sup> (Interviews 5; 9), *about* the products' global sustainability potential (Interviews 4; 5; 2; 9), *about* new markets including the fish, egg and cheese markets (Interviews 10; 11; Informal interview 17), and *about* the potential applications of new technologies such as 3D-printing (Interview 9). Particularly valuable knowledge was generated in the context of BIGMEAT's learning *from* three startups, Startups C, D and F. This type of learning adopted a dual nature and included both product-related and sustainability-related learning. In the former, BIGMEAT managers learned how to develop and produce second generation products for proper target group reach (Interviews 10) and, what was perceived as the most valuable learning, the positioning of such products for proper target group reach (Interviews 8; 10). Concerning the latter, individual managers, in particular BIGMEAT's Board Lead for Alternative Proteins, learned from the startups how business can serve as a powerful vehicle for achieving fundamental change for sustainability (Informal interview 9).

The third learning phase was overwhelmingly marked by negative alliance experiences. In the course of Alliances B, C, D, F and G, BIGMEAT managers had negative experiences with regard to alliance processes relating to what were perceived as bold demands by startup partners, communication misunderstandings and project delays. All this was particularly evident in Alliance C. In the first learning cycle with Alliance C, BIGMEAT had already learned *about* the startup's demand for "collaboration among equals" and that it was not willing to "*let go of the reins*" (Interview 10) of strategic product placement decisions. The second learning cycle involved increasing frustration in the BIGMEAT top management team and among various middle managers (Observation 09/2019), which

<sup>&</sup>lt;sup>2</sup> Second generation products are in appearance, texture, preparation method and taste almost undistinguishable from the animal-based original.

<sup>&</sup>lt;sup>3</sup> Flexitarians are customers who are keen to reduce meat consumption while still eating (some) animal meat.

#### C. Hübel et al.

resulted in delivery delays (Interviews 5; 8) and what was perceived as uncooperative behavior by the startups (Interview 3; Informal interview 15). The impacts were mostly felt by the sales managers who coordinated interactions between the startups and BIGMEAT's sales partners:

"What is happening right now [in Alliance C] is painful. We are consciously inflicting pain on ourselves. And right now, you need a very, very, very high tolerance limit, because it isn't working out as it should." (Interview 8)

No.	Finding	Main topic	Relation to prior research
1.2	Learning <i>about</i> and <i>from</i> sustainability startups occurs in three phases marked by (1) overcoming an inhibition threshold, (2) positive experiences and (3) mixed positive and negative experiences.	Subdivision of alliance learning processes into three distinct phases	This subdivision adds to alliance process research (e.g. Heimeriks et al., 2015). The diverging experiences in alliance learning phases specify evolutionary alliance learning paths (e.g. Doz, 1996).

### 4.3. BIGMEAT's alliance learning outcomes

BIGMEAT's learning *about* and *from* the sustainability startups in the three phases of alliance learning had different outcomes within and beyond organizational boundaries.

*Learning about:* This type of learning enabled the BIGMEAT top management team to make decisions regarding further alliances, establish processes to ensure efficient alliance handling, and make new structural considerations. The first phase of learning in Alliances S and A helped the leadership team to overcome initial resentment toward vegan startups (Interviews 10; 4; 6) and made them "open to dealing with technologies that may represent a massive competition to our core business" (Interview 10). The second phase of valuable learning outcomes, particularly in the context of Alliance C, created a newfound enthusiasm for second-generation plant-based products among BIGMEAT employees (Interviews 10; 7; 6; 3; 4). This was driven by the first taste experience with the startup product: "*I was able to taste it and I said: My goodness, this is a product that can become something*" (Interview 5). The data suggests that initial positive experiences and related valuable knowledge were the main triggers for formulating the new firm strategy of "growth through diversity" (Interview 11; Firm magazine 12/2018) and marked the beginning of a deliberate alliance portfolio approach. Valuable learning in Alliance C also facilitated processes and helped avoid pitfalls in subsequent alliances, especially D, E and F:

"Due to the way we learned to work with each other in [Alliance C], a lot of progress is being made here at the firm [BIGMEAT]. Thus, for the next startup, we already know from the outset: Okay, we also have to pay close attention to this or that aspect, we have to approach that differently." (Interview 4)

BIGMEAT's alliance decisions increasingly shifted from startups for meat analogues (S, A, B, C) to startups that can be placed under a much wider "*protein umbrella*" (Observation November 2018; D, F, G). Engaging with diverse alternative protein startups further away from BIGMEAT's core product range (i.e. meat) stressed the firm's willingness to engage in sustainability-oriented product innovation in the broader alternative protein field. In line with this, the data indicates that learning processes sped up significantly over the third learning phase and BIGMEAT's decisions on alliances were made faster (e.g. F, G, H).

The negative experiences in the third learning phase in Alliances B, D, F and G prompted the top management team to renew financial support for Startup B (Interviews 9; 11), to postpone planned market launches with Startups D and F (Interviews 3; 4; 10), and to eventually terminate Alliance G in January 2020 (Board Discussion 02/2020). Even though experiences were particularly negative with Alliance C and evoked the desire among BIGMEAT managers to become independent from startup demands (Informal interview 15), the benefit of Alliance C in enabling BIGMEAT to change its product portfolio outweighed the frustration (Interview 10). Instances of negative alliance experiences led to the acceleration of BIGMEAT's own plant-based product development (Interview 5; Informal interviews 13; 15). At the organizational level, "We try to really push our own product development forward now" (Interview 9). Individuals were also affected by negative alliance experiences: "[This] spurred me on personally and the same happened with my team, I mean those people working for me" (Interview 3).

No.	Finding	Main topic	Relation to prior research
2.1	Continuous positive experiences related to valuable knowledge <i>about</i> the startup partners accelerated decisions on alliance partners further away from the core business. Later negative experiences related to learning <i>about</i> startup partners accelerated the established firm's innovation processes.	The changing nature and purpose of learning <i>about</i> over time	The finding extends previous work on alliance learning types (e.g. Inkpen and Currall, 2004) and outcomes of learning <i>about</i> (Doz, 1996) by considering learning over time.

*Learning from:* Product-related learning *from* the startups prompted BIGMEAT to abandon their previous focus on vegan products serving market niches and instead started to develop second-generation plant-based protein products for the mass market (Interviews 5; 8; 9):

"We increasingly focus on pimping our existing products in terms of marketing and taste, maybe integrating something that the [product of Startup C] has that we haven't had yet." (Interview 5)

#### C. Hübel et al.

Realizing the boundaries of its own meat focused product development (Interviews 3; 10), BIGMEAT's management first increased collaboration with external product developers and then in early 2020 created new positions in the product development department (Interviews 4; 10; Board discussion 02/2020). This served the future objective to have a separate product development department for alternative proteins: "*The more in-house the better*" (Interview 10). Before engaging in alliances, BIGMEAT's sales manager thought of positioning the alternative protein products with its veggie food service and product brand, targeting vegan or vegetarian consumers (Interview 8). In contrast to this original product placement strategy, BIGMEAT learned from Alliance C to ask retailers to position these products directly in the same aisle as conventional meat products.

# "[Startup C] knows today that 92 percent of the customers that buy [products of Startup C] also buy meat. It shows us that the positioning and orientation of [Startup C] is exactly right." (Interview 8)

Based on this learning from Startup C, all new BIGMEAT products will now be positioned in close proximity to animal-derived meat. Concerning product marketing, BIGMEAT managers learned how to implement an alternative protein branding strategy. Before the alliances, BIGMEAT's operations focused on the conventional high-volume meat market, which typically does not involve any brands. In late 2019 and early 2020, BIGMEAT formulated its first social media strategy and planned a separate communication channel for BIGMEAT's own new alternative protein brand (Interview 11; Board discussion 02/2020). This activity was not directly triggered by the alliances, but *"It helps to see, of course, which successes a partner [...] has with which strategy. In terms of social media, [Startup C] is certainly a good example"* (Interview 12). The more knowledge about plant-based protein innovations filtered into BIGMEAT's products (Interview 4; Informal interview 9). BIGMEAT now aims at increasing the share of total revenue for alternative protein products from 5% in January 2018 to 25% in 2025 (Media report 08/2019). However, the rapid implementation of alternative protein products was only possible through learning enabled by BIGMEAT's existing expertise and the infrastructure of the conventional meat industry.

Sustainability-related learning that occurred in the context of the three alliances prompted individual BIGMEAT employees to adopt novel perspectives and integrate global sustainability concerns into decision-making (Interview 4; Informal interview 15). BIGMEAT's project team member and trend scouter, for instance, recounts how the interaction with the founder of Startup F gave him a new "*awareness to find alternatives*" and encouraged him to consider global food security concerns. According to him, "*These are perspectives that I have only adopted in the last few years because of such people and companies*" (Interview 4). Similarly, other BIGMEAT managers, after learning about the sustainability potential of the startups' products, began to question the long-term viability of conventional meat production and became more open toward alternative protein solutions (Interview 3; 5; 8; Informal interview 7). Adopting the startups' approaches to the development, positioning and marketing of BIGMEAT's plant-based product innovations was increasingly seen as a way to combine the goals of increasing sales revenue in the mass market and gaining competitive advantage as well as the proactive goal of contributing to sustainability (Interview 9; Informal interview 9). The data indicate that knowledge spillovers of sustainability protein field as contributing to their agenda of transforming the market toward sustainability (Interviews 14; 15; 17). The owner of startup A states, for instance:

"We are definitely an ideologically-driven startup [...] Maybe in other ways [BIGMEAT] wouldn't have been exposed to the potential of the ideologically-driven activities [we display in our business]. I think in that sense we are affecting them. And through this, [BIGMEAT] is also exposed to the audiences that very much relate to this ideology. And [BIGMEAT] sees other potential value in that as well. I think it's extremely beneficial for all sides." (Interview 14)

No.	Finding	Main topic	Relation to prior work
2.2	Product-related knowledge for sustainability innovation advanced the established firm's innovation capability. Sustainability-related learning <i>from</i> sustainability startups resulted in changes to perspectives and behaviors of individuals in the established firm, advancing their ambition for corporate sustainability.	The dual nature and purpose of learning <i>from</i> in the context of sustainability innovations	The specification of learning <i>from</i> in the context of sustainability innovations extends work on alliance learning types (Inkpen and Currall, 2004) and outcomes of learning <i>from</i> (e.g. Alvarez and Barney, 2001)

*Combined learning over time*: The continuous flow of valuable knowledge related to learning *about* the startup's high market potential and the products' compatibility with BIGMEAT's existing expertise, and the occasional occurrence of learning *from*, which was related to developing and marketing plant-based products, coupled with later negative experiences related to alliance processes and coordination resulted in a powerful outcome for sustainability innovation. The combination of different learning types and experiences over time made BIGMEAT's top management eager to find ways to "*extremely expand*" in the alternative protein field while simultaneously avoiding financial risks, dependencies and challenges that come with an increasing number of alliance partners (Informal interview 6). As a result, they decided to co-found a global accelerator platform for plant-based protein products as well as a joint venture for the production and distribution of these products across Europe (Interview 9; 10; 11). With these new activities, BIGMEAT was aiming at the "*accelerated development of strong brands, high quality standards and a wide range of products for a mass market that* 

Long Range Planning xxx (xxxx) xxx

### extends far beyond vegan and vegetarian niches" (Press release 02/2020).

No.	Finding	Main topic	Relation to prior research
2.3	The combined effect of alliance learning types and alliance learning experiences over time resulted in sustainability-oriented corporate entrepreneurial activities aiming at broader market impact	The large-scale sustainability impact of accumulated alliance learning experiences	Adds a market perspective to the inter- and intra- organizational perspective on alliance learning outcomes suggested by previous research focused on alliance performance (e.g. Liu and Lui, 2020), alliance decisions (e.g. Heimeriks et al., 2007) learning capabilities (e.g. Schilke and Goerzen, 2010) and corporate innovation (e. g. Kavusan et al., 2016)

The findings show that learning from various startups in strategic alliances continuously increased BIGMEAT's sustainability innovation in the alternative protein field.

### 4.4. Process model development

C. Hübel et al.

A new process model for alliance-driven learning for corporate sustainability innovation depicts how the findings are sequentially linked (Fig. 4). The grey area encompassing alliance learning processes combines Findings 1.1 and 1.2 on alliance learning types and experiences in three alliance learning phases. The underlying light grey area encompassing alliance learning outcomes combines Findings 2.1, 2.2 and 2.3 on alliance decisions, in-house sustainable product innovation and sustainable mass market transformation. The temporal sequences as well as direct and indirect influences of alliance learning are indicated by dark colored arrows.

In line with Finding 1.2, the conceptual model distinguishes three learning phases in the evolutionary alliance learning process. The different scopes and contents of the learning cycles, which are represented in Finding 1.1, suggest an at least equal importance of learning *about* and learning *from* for corporate innovation. The larger distance between the first and second learning cycles indicates the inhibition threshold identified in Finding 1.2. In line with Finding 2.3, the dark arrows at the top and bottom show how the combination of alliance learning types and experiences can advance and accelerate corporate entrepreneurial activities for sustainable mass market transformation. Such transformation is understood as the replacement of conventional products, services and market structures by superior environmental and social products and services (Hockerts and Wüstenhagen, 2010; Schaltegger and Wagner, 2011). In the case of BIGMEAT, this happened directly through (1) the formation of alliances further away from the core business (Finding 2.1, arrows at top), and (2) the acceleration of in-house product innovation (Finding 2.2, arrows at bottom). The dotted arrows show the indirect transformation outcomes of sustainability learning (Finding 2.2). They visualize how personal learning from startup owners advanced the sustainability ambition of individual managers at BIGMEAT, influencing the firm's innovation activities and therefore also driving market change. The dark colored areas in the process model indicate those alliance learning characteristics



sustainability-related alliance learning

Fig. 4. Process model of alliance-driven learning for corporate sustainability innovation.

#### C. Hübel et al.

# **ARTICLE IN PRESS**

#### Long Range Planning xxx (xxxx) xxx

that were specific to the sustainability context. The following section elaborates on how these insights help close the research gap identified in this paper.

### 5. Discussion

Previous research dealing with alliance learning processes and outcomes has described learning as highly evolutionary (Doz, 1996) and has defined two main types of learning, learning *about* and learning *from*, which influence decision-making and the success of alliances or corporate innovation in the established partner firm (Inkpen and Currall, 2004; Das and Kumar, 2007). In this regard, there are some general findings about the temporal occurrence of learning *about* and *from* in the alliance process (Das and Kumar, 2007). So far however, no empirical study has explored alliance learning processes in the context of sustainability innovations, in particular different learning experiences, challenges and outcomes. Our case findings address the research gap by offering insights into the processes and outcomes of an established firm's learning in alliances with sustainability startups. Our study makes important contributions to the existing literature as discussed in the following.

### 5.1. Characteristics of alliance learning processes and outcomes for sustainability

The study findings are in line with the initially posed assumption that alliance learning processes and outcomes differ in the context of sustainability innovations. Our process model shows the three distinct characteristics of sustainability-related alliance learning (see dark colored areas in Fig. 4). Each of these characteristics extends prior research in alliance learning.

Inhibition threshold: The inhibition threshold identified in the study, which is seen in an initially reduced learning scope and pace, provides a new temporal perspective on differences of learning processes between established firms and sustainability startups. The observed hesitancy and only gradual increase in learning pace over time extends previous alliance learning research that has emphasized the facilitation role of partner familiarity and trust for learning in alliances (Inkpen and Currall, 2004; Nielsen, 2005; Nielsen and Nielsen, 2009). Our study's findings combine insights on alliance learning processes with previous sustainability research, which has pointed to potential complications in alliance learning due to differences between mainstream and sustainability actors (London et al., 2004; O'Mahony and Bechky, 2008; Rondinelli and London, 2003; Stern and Hicks, 2000). In the case of BIGMEAT, extensive learning *about* sustainable startup partners and their collaborative potential (particularly Alliance S) was necessary for the firm to eventually learn *from* the first startup (Alliance C) in later phases of the alliance process. The inhibition threshold experienced in sustainability-oriented alliances can thus help to explain the observed time lag between the first experience of learning *about* and the first experience of learning *from*, and the gradual acceleration of alliance learning over time. This indicates that the more alliances an established firm enters into, the greater the likelihood is to find sustainability partners to learn *from* and to find the one special alliance that becomes the main driver of learning for sustainability innovation.

*Dual nature of learning from*: By outlining the temporal occurrence and outcomes of sustainability-related learning, our study adds a new dimension to learning *from* that becomes relevant in the context of sustainability innovations and that has not yet been discussed in strategic management literature (Doz and Hamel, 1998; Khanna et al., 1998; Inkpen and Currall, 2004; Inkpen and Tsang, 2007). The adoption from startup owners of new sustainability perspectives by some BIGMEAT managers highlights the relevance of personal interactions for sustainability innovations. This finding relates to Felin et al.'s (2012) microfoundation view of capabilities. The finding that sustainability startup partners welcomed sustainability-related learning at the established firm conflicts with previous alliance learning research. It was previously assumed that accessing and internalizing knowledge from partners constitutes a learning race and struggle for dominance that negatively impacts the relationship between the partner firms (Alvarez and Barney, 2001; Hamel, 1991; Inkpen and Beamish, 1997; Inkpen and Currall, 2004; Van de Ven and Walker, 1984). While this might be true for conventional product-related learning outcomes, the opposite may be true for sustainability-related learning outcomes. As BIGMEAT only internalized sustainability-related knowledge from the three startups that also provided relevant product-related knowledge, our study further suggests that sustainability-related learning *requires* previous valuable product-related learning.

Learning outcomes for market transformation: The identification of alliance learning outcomes beyond organizational boundaries extends previous alliance learning research, which has focused on learning outcomes for alliance learning capability and alliance performance (e.g. Kale and Singh, 2007; Liu and Lui, 2020), decisions on alliance formation (e.g. Gulati et al., 2009; Heimeriks et al., 2007), and corporate innovation (e.g. Subramanian and Soh, 2017). The case of BIGMEAT gives indications of alliance learning outcomes that benefit the transformation of mass markets toward sustainability. It is thus in line with previous suggestions that large established companies have significant impact on the marketplace by dedicating resources toward addressing sustainability challenges (Hockerts and Wüstenhagen, 2010). The learning-triggered formation of additional distribution alliances increases the availability of meat alternatives in mainstream distribution channels. Early impacts were observed in the context of Alliance C, with products having reached conventional meat consumers, as well as non-organic retail and food service chains in May 2019, thirteen months after Alliance C was established. Thus, our finding also supports that there is a positive relationship between corporate sustainability innovations and competitiveness (Hall and Vredenburg, 2004; HermundsdottirAspelund, 2021). The learning alliance with BIGMEAT further accelerated the expansion rates of individual startups, thereby increasing their impact on conventional markets. For instance, Startup C's revenue quadrupled between 2018 and 2020. Considering that the goal of sustainability innovations is to solve broader problems relating to planetary boundaries or the United Nations' Sustainable Development Goals (e.g. Johnson and Schaltegger, 2020), market transformation triggered by alliance learning has the potential to contribute to the sustainable development of society as a whole.

#### C. Hübel et al.

#### 5.2. Temporal occurrence and purpose of learning types in specific learning phases

Our process model highlights the study's new findings on the temporal occurrence of learning *about* and learning *from* in the three alliance learning phases, and differentiates the learning type and purpose depending on contextual alliance experiences (see learning cycles, types and phases in Fig. 4). While these findings arise from our sustainability-oriented case study, our findings suggest that they are not necessarily limited to the sustainability context.

The crucial role of learning about: The continual occurrence and far-reaching impact of learning *about* emphasize the crucial role of this type of learning for advancing corporate innovation. This finding is distinct from Das and Kumar's (2007) suggestion that learning *from* dominates learning *about* in the alliance operation phase. In contrast, our empirical study identified the repeated dominance of learning *about* over learning *from* in the alliance process. Furthermore, our findings extend research that has limited the outcomes of learning *about* to alliance-specific processes (Doz, 1996; Ireland et al., 2002). Our study suggests that learning *about* directly accelerates innovation in the established firm, particularly in the later stages of the alliance process. The prevalence of learning *about* enabled BIGMEAT to enter a wide range of alliances with product offers increasingly distanced from BIGMEAT's core business of animal derived meat products. Confirming Grant and Baden-Fuller (2004) knowledge accessing theory, we find that learning *about* can trigger opportunities for strategic re-orientation over time. Our study, however, does not support Subramanian and Soh's (2017) finding that increased partner diversity positively impacts the breadth of the resulting corporate innovation. Even though BIGMEAT acquired knowledge with regard to plant-based meat, egg and dairy as well as cell-based meat, the BIGMEAT product innovations remained focused on plant-based meat substitutes.

*Positive and negative experiences*: Phase 2 and 3 of our process model suggest that alliance learning involves positive experiences in the early phase and mixed positive and negative experiences in later phases. Indeed, past research has addressed positive experiences related to the acquisition of valuable knowledge (Das and Kumar, 2007) and negative experiences related to learning challenges in alliances between established firms and startups (Barabel et al., 2014; Doz, 1988; Minshall et al., 2010; Pérez et al., 2012). Our case study, however, introduces a novel process perspective on these different experiences within alliances. Most importantly, we show that negative experiences in Phase 3 of the alliance process, despite being detrimental to the alliance process as such, can become key drivers of radical innovation activities in an established firm. In the case of BIGMEAT, negative alliance experiences had a very significant impact due to an aggregation of parallel learning cycles within different alliances. In the case of Alliance C, for example, negative experiences were linked to the startup's strong sales relationships and this resulted in an unanticipated sense of dependency at BIGMEAT. This finding conflicts with previous research suggesting that partner asymmetry implies unequal processes to the detriment of the smaller firm (Barabel et al., 2014; Doz, 1988; Minshall et al., 2010; Pérez et al., 2012; Prashantham and Birkinshaw, 2008).

### 6. Conclusion, limitations and future research

This study provides empirical insights into the processes and outcomes of alliance learning in the context of sustainability innovation and establishes a conceptual link to their role in fostering market transformation toward sustainability.

The analysis provides valuable contributions to the alliance learning literature by identifying three distinct characteristics of sustainability-related alliance learning. We found that learning in alliances with sustainability startups (1) requires overcoming an inhibition threshold, and is thus initially slow and only accelerates over time, (2) includes not only product-related learning (i.e. development, marketing), but also sustainability-related learning (i.e. perspectives, goals), and (3) can indirectly and directly advance a firm's contribution to sustainable mass market transformation. The case of BIGMEAT shows how sustainability-related alliance learning processes can be separated into three distinct alliance learning phases with positive and negative alliance experiences, in which the temporal occurrence of learning types and their purpose (*about* and *from*) differ. Most importantly, we found the crucial role of learning *about* for advancing corporate sustainability innovation over time, which conflicts with previous research's strong focus on learning *from*.

BIGMEAT constitutes a unique case in the meat industry, and accordingly case research was designed for internal validity, construct validity and reliability (Gibbert et al., 2008). To increase external validity (i.e. transferability) of the findings and therefore our alliance learning process model to other established firms, future research could further investigate processes and outcomes of direct collaborations between established firms and sustainability startups. In addition, the startup perspective could be explicitly explored by analyzing learning cycles and outcomes for small firms collaborating with large firms. These analyses would constitute a valuable addition to the findings presented here. The processes of intra-organizational knowledge diffusion and potentially diverging learning outcomes in different organizational departments or among individual employees have a high impact on corporate innovation. It would therefore be valuable to explore these differences with regard to learning *about* and learning *from* alliance partners and their potential impact on corporate sustainability innovation. Further, it might be worthwhile to investigate to what extent reputational concerns – as those voiced in Alliance S – can hamper corporate engagement with sustainability innovation. Lastly, since our study indicates only the potential for mass market transformation, future longitudinal studies could provide more detailed accounts of actual market changes following alliance processes.

### Role of funding source

This research has profited from funding by the Robert Bosch Stiftung, project no. 12.5.F082.0021.0. The Robert Bosch Foundation had no involvement in the study design, the collection, analysis and interpretation of data, in the writing of the article or in the decision to submit the article for publication.

### C. Hübel et al.

### Author statement

Charlott Hübel: Writing- Original Draft, Investigation, Formal Analysis, Conceptualization Ilka Weissbrod: Writing- Review & Editing, Conceptualization, Formal Analysis Stefan Schaltegger: Supervision, Conceptualization, Writing- Review & Editing.

### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

### Appendix A. Interview details

 Table A.1

 Details of semi-structured interviews at BIGMEAT and startups

Interview No.	Firm of interviewee	Position of interviewee(s)	Duration of interview
1	BIGMEAT	CEO/Chair	74 min
2	BIGMEAT	CEO of key subsidiaries	47 min
3	BIGMEAT	Project Manager/Product Developer	52 min
4	BIGMEAT	Project Manager/Trend Scouter	47 min
5	BIGMEAT	CEO of international sales firm	45 min
6 (group interview)	BIGMEAT	A: Head of Sustainability Management & PR	42 min
		B: Sustainability Manager & Board Secretary	
7	BIGMEAT	Alliance Coordination Manager	34 min
8	BIGMEAT	Sales Manager	51 min
9	BIGMEAT	Head of Alternative Protein Unit	49 min
10	BIGMEAT	Board Lead for Alternative Proteins	73 min
11	BIGMEAT	Press Spokesperson	47 min
12	BIGMEAT	Marketing Manager	47 min
13	Startup S	Founder/CEO	40 min
14	Startup A	Co-Founder/CEO	52 min
15	Startup B	Co-Founder/CEO	49 min
16	Startup C	Board member	32 min
17	Startup D	Co-CEO/Chair	30 min
18	Startup E	Executive Vice President	36 min
19	Startup F	Co-Founder/CEO	29 min
20	Startup G	Founder/CEO	52 min

### Table A.2

Details of informal conversational interviews at BIGMEAT

Interview No.	Interviewee	Date of interview	Interview location/context
1	Board Lead for Alternative Proteins	August 2018	Phone conversation
2	CEO of key subsidiaries	September 2018	Firm headquarters
3	Project Manager	October 2018	Firm headquarters
4	Board Lead for Alternative Proteins	November 2018	Industry event
5	Board Lead for Alternative Proteins	January 2019	Firm headquarters
6	Board Lead for Alternative Proteins	February 2019	Phone conversation
7	Project Manager	March 2019	Press event
8	Board Lead for Alternative Proteins	May 2019	Meeting with potential alliance partner
9	Board Lead for Alternative Proteins	May 2019	Industry event
10	Board Lead for Alternative Proteins	June 2019	Industry event
11	International Sales Manager	June 2019	Industry event
12	Board Lead for Alternative Proteins	July 2019	Firm headquarters
13	Board Lead for Alternative Proteins	August 2019	Firm headquarters
14	Sales Manager	August 2019	Firm headquarters
15	Board Lead for Alternative Proteins	September 2019	Industry event
16	Board Lead for Alternative Proteins	October 2019	Industry event
17	Head of Alternative Protein Unit	October 2019	Industry event
18	Head of Alternative Protein Unit	November 2019	Firm headquarters
19	Marketing Manager	November 2019	Firm headquarters
20	Board Lead for Alternative Proteins	November 2019	Firm headquarters

## Long Range Planning xxx (xxxx) xxx

# Appendix B. MAXQDA coding process

Start Import Codes Memos Variablen Analyse M	Methods Visual Tools Reports MAXIDictio	うて く <b>な</b> © ②
Image: Second	Image: Section of the sectio	
📔 Liste der Dokumente 📔 🌓 💽 🕣 🔎 🛱	🖓 🖛 🗴 🕜 Dokument-Browser: Interview, Board Lead Alternative Proteins 5.11.2019 (114 Absätze)	▶ ୬ ♥ ♥ @ ≖ ×
Character, Send Spert Workhey, 170,2019     Conversion, Meetry Dirker, ever, 150,2019     Conversion, Meetry Dirker, ever, 150,2019     Conversion, Meetry Dirker, ever, 150,2019     Conversion, Meetry Dirker, 2011,2019     Conversion, Meetry Dirker, 2012,2019     Conversion, Meetry Dirker, 2012,2019     Conversion, Meetry Dirker, 2012,2019     Conversion, Project Manager, 2012,2019     Conversion, Project Manager, 2012,2019     Conversion, Project Manager, 2012,2019     Conversion, 2012,2019     Conversion, Project Manager, 2012,2019     Conversion, 2012,2019     Con	<ul> <li>Establied film duracterises &amp; strategy</li> <li>Istablied film duracterises &amp; strategy</li> <li>Istab</li></ul>	Instrugeauer was war zu voert ent? Imsted? VWe geht sie auch mit in Bette- und da war keiner inzu #87024# sis echt aus ale Leute auf der anderen ninnen zusammen arbeiten gab es vohrte nicht. Es war lichschrühe nicht. Aber aft tickt, wie wir Kunden war von Anfang an extrem
Codesystem     C	Percent Interface	ussen often den neuen nn ist das schon auch das segründet. Ja, es macht ung damals. In den anderen in ist. Was hat sich mit laube, mit ##Startup A## hat ssive Konkurrenz darstellen. dere möglicherweise sagen: eventeil machen werden. Wit
) 2 Strange C 2 Strange D 2 Strange F 2 Strange C 2	and matchen genau das Gegenteil. Wir werden es nicht bekämpfen, wir werden es nicht versuchen klein zu halten, sondern wir sagen: "Wir müssen uns 56 57 58 59 50 50 50 50 50 50 50 50 50 50	ogeneratinkeneratinkenerationen ma h admit aktiv beschäftigen." h mal sehr viel näher am r gemacht haben.





Table C.1						
BIGMEAT	and startup	motivations	for	entering	the alliand	es

Alliance	Articulated motivation for entering alliance				
	BIGMEAT	Startups			
S	"We got to know each other through contacts and then started the talks. And we said: You know what? We can actually both learn from each other. Everyone can learn from the other. It's not that one is bad and the other one good. Let's just try the experiment." (Interview 10)	"For me it was more like the pressure is taken away from me It was more like saying it is great that it is taken away from me, that you have a partner who would catch you if you stumble. I did wish for such opportunities that they would help you to gain a foothold abroad, but I didn't know whether everything would really happen that way or whether they really did have the opportunity. This makes it especially great that it worked out too." (Interview 13)			
Α	"We have to deal with it, because if we don't do it, others will. And we cannot understand this technology if we are not involved. And we cannot prevent this technology." (Interview 10)	"We are definitely an ideologically-driven startup. I think it is one of our advantages. From the start we thought that partnering with the meat industry and with major players like [BIGMEAT] would be extremely important and beneficial to cultured meat firms. [] We wanted to partner with them, because we believed that, in order to really make an impact, partnering with an existing industry - that would allow us to get the type of [impact]" (Interview 14)			
В	"[Alliance B] enables us as [BIGMEAT] to get a taste of a new area and to benefit a little from the startup world." (Interview 10)	"I would say: innovation meets experience. Especially at the beginning. I think this is extremely fruitful and there should be a lot more of it. [] Maybe I'm only speaking for myself here, but I think we now know what sustainability means. We try to live accordingly. I think many of the older generation are becoming familiar with the term, but not yet properly. This is why you have to combine innovation and experience." (Interview 15)			
С	"But in the case of [Startup C], when you bite into [the product] for the first time, I thought: This is a completely new league, which we don't even know in Europe. This is how the idea came up: Can't we also offer this to our customers, who we already have in the meat sector, and thereby possibly turn it into an extremely attractive product?" (Interview 10)	"Together we can achieve something that one cannot achieve alone, at least not in time. The meat producer cannot bring the product to the consumer if he doesn't get it from us. We cannot address the consumer in [Europe] because we do not have the infrastructure, the network, the distribution and the cold chain. If our common goal is to bring the product to the European consumer, then we need a partner in Europe who enables us to do this, and so we have an ideal situation." (Interview 16)			
		(continued on next page)			

(continued on next page)

#### C. Hübel et al.

### Table C.1 (continued)

Long Range Planning xxx (xxxx) xxx

Alliance	Articulated motivation for entering alliance				
	BIGMEAT	Startups			
D	"Doesn't it make sense to include these areas, as we also have the sales logistics and production skills to cover these products accordingly?" (Interview 10)	"For us, because we have a mission, because time is of the essence, who better to help us speed that up than the firms that are already really well established? [] I don't care whether you are looking from animal welfare or environmental degradation or climate change or health care, [for] all of those things, time is not on our side. So, speed to market, speed to innovation, all of that strategic help accelerate and that's why we partner with them." (Interview 17)			
Ε	"We are investing in [Startup E] with a small amount to reduce the use of soy as much as possible."	<sup>•</sup> It allowed us to accelerate discussions with our partners for the various products. It allowed us to start development of plant two and three. It was very meaningful to us in terms of going from demonstration to pure commercial." (Interview 18)			
F	[Startup F] fits under [BIGMEAT's] new "protein umbrella." It is more about "enlarging the pie," creating completely new categories, and giving customers a choice. (Meeting 10/2018)	"If firms can help us do more good faster, I am going to partner with them. It's as simple as that." (Interview 19)			
G	The owner of [Startup G] approached [BIGMEAT] and he is "a great guy." The cheese has a lot of possible uses and can enhance the existing product range. (Informal interview 15)	"Through [BIGMEAT], I have the opportunity to be represented in so many supermarkets at the same time, and discounters too, and thus I can do TV advertising [] [BIGMEAT] opens the doors for us to bring the product to the midst of society." (Interview 20)			
н	"In the case of [Startup H], it is also a long-term partnership, where still a lot of development has to be put it. But if it is going to happen, it is a really great technology. And we will be a part of it." (Interview 9)	no startup interview -			

### References

- Anand, B.N., Khanna, T., 2000. Do firms learn to create value? The case of alliances. Strat. Manag. J. 21, 295–315. https://doi.org/10.1002/(SICI)1097-0266(200003) 21:3<295::AID-SMJ91>3.0.CO:2-O.
- Adams, R., Jeanrenaud, S., Bessant, J., Denyer, D., Overy, P., 2016. Sustainability-oriented innovation: a systematic review. Int. J. Manag. Rev. 18, 180–205. https://doi.org/10.1111/ijmr.12068.
- Alvarez, S.A., Barney, J.B., 2001. How entrepreneurial firms can benefit from alliances with large partners. Acad. Manag. Perspect. 15, 139–148. https://doi.org/ 10.5465/ame.2001.4251563.
- Arnold, M.G., Hockerts, K., 2011. The greening dutchman: Philips' process of green flagging to drive sustainable innovations. Bus. Strat. Environ. 20, 394–407. https://doi.org/10.1002/bse.700.
- Bansal, P., DesJardine, M.R., 2014. Business sustainability: it is about time. Strat. Organ. 12, 70–78. https://doi.org/10.1177/1476127013520265.
- Barabel, M., Meier, O., Soparnot, R., 2014. Asymmetric alliances between SMEs and large firms in the area of innovation: strategic determinants and cultural effects. Gestion 31, 87–106. https://doi.org/10.3917/g2000.316.0087, 2000.
- Berghman, L., Matthyssens, P., Streukens, S., Vandenbempt, K., 2013. Deliberate learning mechanisms for stimulating strategic innovation capacity. Long. Range Plan. 46, 39–71. https://doi.org/10.1016/j.lrp.2012.11.003.
- Bingham, C.B., Heimeriks, K.H., Schijven, M., Gates, S., 2015. Concurrent learning: how firms develop multiple dynamic capabilities in parallel. Strat. Manag. J. 36, 1802–1825. https://doi.org/10.1002/smj.2347.
- Bocken, N.M., Geradts, T.H., 2020. Barriers and drivers to sustainable business model innovation: organization design and dynamic capabilities. Long. Range Plan. 53, 101950 https://doi.org/10.1016/j.lrp.2019.101950.
- Bossink, B.A., 2007. The interorganizational innovation processes of sustainable building: a Dutch case of joint building innovation in sustainability. Build. Environ. 42, 4086–4092. https://doi.org/10.1016/j.buildenv.2006.11.020.
- Bradbury-Huang, H., Lichtenstein, B., Carroll, J.S., Senge, P.M., 2010. Relational space and learning experiments: the heart of sustainability collaborations. Res. Organ. Change Dev. 18, 109–148. https://doi.org/10.1108/S0897-3016(2010)0000018008.
- Chen, H., Chen, T.J., 2002. Asymmetric strategic alliances: a network view. J. Bus. Res. 55, 1007–1013. https://doi.org/10.1016/s0148-2963(02)00284-9.
- Cohen, W.M., Levinthal, D.A., 1990. Absorptive capacity: a new perspective on learning and innovation. Adm. Sci. Q. 128–152. https://doi.org/10.2307/2393553. Cui, A.S., O'Connor, G., 2012. Alliance portfolio resource diversity and firm innovation. J. Market. 76, 24–43. https://doi.org/10.1509/jm.11.0130.
- Das, T.K., Kumar, R., 2007. Learning dynamics in the alliance development process. Manag. Decis. https://doi.org/10.1108/00251740710745980.
  Das, T.K., Teng, B.S., 2002. The dynamics of alliance conditions in the alliance development process. J. Manag. Stud. 39, 725–746. https://doi.org/10.1111/1467-6486.00006
- Davis, J.P., Eisenhardt, K.M., 2011. Rotating leadership and collaborative innovation: recombination processes in symbiotic relationships. Adm. Sci. Q. 56, 159–201. https://doi.org/10.1177/0001839211428131.
- Dean, T.J., McMullen, J.S., 2007. Toward a theory of sustainable entrepreneurship. Reducing environmental degradation through entrepreneurial action. J. Bus. Ventur. 22, 50–76. https://doi.org/10.1016/j.jbusvent.2005.09.003.
- DeCuir-Gunby, J.T., Marshall, P.L., McCulloch, A.W., 2011. Developing and using a codebook for the analysis of interview data: an example from a professional development research project. Field Methods 23, 136–155. https://doi.org/10.1177/1525822X10388468.
- DeWalt, K.M., DeWalt, B.R., 2002. Participant Observation: A Guide for Fieldworkers. AltaMira Press, Walnut Creek.
- Doz, Y.L., 1988. Technology partnerships between larger and smaller firms: some critical issues. Int. Stud. Manag. Organ. 17, 31–57. https://doi.org/10.1080/00208825.1987.11656466.
- Doz, Y.L., 1996. The evolution of cooperation in strategic alliances: initial conditions or learning processes? Strat. Manag. J. 17, 55–83. https://doi.org/10.1002/ smj.4250171006.
- Doz, Y.L., Hamel, G., 1998. Alliance Advantage: the Art of Creating Value through Partnering. Harvard Business Press, Brighton
- Dubois, A., Gadde, L.E., 2002. Systematic combining: an abductive approach to case research. J. Bus. Res. 55, 553–560. https://doi.org/10.1016/S0148-2963(00) 00195-8.
- Dwyer, S.C., Buckle, J.L., 2009. The space between: on being an insider-outsider in qualitative research. Int. J. Qual. Methods 8, 54–63. https://doi.org/10.1177/160940690900800105.
- Felin, T., Foss, N.J., Heimeriks, K.H., Madsen, T.L., 2012. Microfoundations of routines and capabilities: individuals, processes, and structure. J. Manag. Stud. 49, 1351–1374. https://doi.org/10.1111/j.1467-6486.2012.01052.x.

#### C. Hübel et al.

#### Long Range Planning xxx (xxxx) xxx

Freudenreich, B., Lüdeke-Freund, F., Schaltegger, S., 2019. A stakeholder theory perspective on business models: value creation for sustainability. J. Bus. Ethics 1–16. https://doi.org/10.1007/s10551-019-04112-z.

Gibbert, M., Ruigrok, W., Wicki, B., 2008. What passes as a rigorous case study? Strat. Manag. J. 29, 1465–1474. https://doi.org/10.1002/smj.722.

Gopalakrishnan, S., Scillitoe, J.L., Santoro, M.D., 2008. Tapping deep pockets: the role of resources and social capital on financial capital acquisition by biotechnology firms in biotech-pharma alliances. J. Manag. Stud. 45, 1354–1376. https://doi.org/10.1111/j.1467-6486.2008.00777.x.

Grant, R.M., Baden-Fuller, C., 2004. A knowledge accessing theory of strategic alliances. J. Manag. Stud. 41, 61–84. https://doi.org/10.1111/j.1467-6486.2004.00421.x.

Gulati, R., Lavie, D., Singh, H., 2009. The nature of partnering experience and the gains from alliances. Strat. Manag. J. 30, 1213–1233. https://doi.org/10.1002/smj.786.

Hall, J., Vredenburg, H., 2004. Introduction: sustainable development innovation and competitive advantage: implications for business, policy and management education. Corp. Sustain.: Governance, Innov. Strat., Dev. Methods 129–140. https://doi.org/10.5172/impp.2004.6.2.129.

Hamel, G., 1991. Competition for competence and interpartner learning within international strategic alliances. Strat. Manag. J. 12, 83–103. https://doi.org/ 10.1002/smj.4250120908.

Hansen, E.G., Grosse-Dunker, F., Reichwald, R., 2009. Sustainability innovation cube: a model to evaluate sustainability innovations. Int. J. Innovat. Manag. 13, 683–713. https://doi.org/10.1142/S1363919609002479.

Heimeriks, K.H., Duysters, G., 2006. Alliance capability as a mediator between experience and alliance performance: an empirical investigation into the alliance capability development process. J. Manag. Stud. 44, 25–49. https://doi.org/10.1111/j.1467-6486.2006.00639.x.

Heimeriks, K.H., Duysters, G., Vanhaverbeke, W., 2007. Learning mechanisms and differential performance in alliance portfolios. Strat. Organ. 5, 373–408. https://doi.org/10.1177/1476127007083347.

Heimeriks, K.H., Bingham, C.B., Laamanen, T., 2015. Unveiling the temporally contingent role of codification in alliance success. Strat. Manag. J. 36, 462–473. http://www.jstor.org/stable/43897780.

Hermundsdottir, F., Aspelund, A., 2021. Sustainability innovations and firm competitiveness: a review. J. Clean. Prod. 280, 124715 https://doi.org/10.1016/j. iclepro.2020.124715.

Hockerts, K., Wüstenhagen, R., 2010. Greening Goliaths versus emerging Davids - theorizing about the role of incumbents and new entrants in sustainable entrepreneurship. J. Bus. Ventur. 25, 481–492. https://doi.org/10.1016/j.jbusvent.2009.07.005.

Inkpen, A.C., 1998. Learning and knowledge acquisition through international strategic alliances. Acad. Manag. Perspect. 12, 69–80. https://doi.org/10.5465/ ame.1998.1333953.

Inkpen, A.C., 2008. Knowledge transfer and international joint ventures: the case of NUMMI and General Motors. Strat. Manag. J. 29, 447–453. https://doi.org/ 10.1002/smj.663.

Inkpen, A.C., Beamish, P.W., 1997. Knowledge, bargaining power, and the instability of international joint ventures. Acad. Manag. Rev. 22 (1), 177–202. https://doi.org/10.5465/amr.1997.9707180263.

Inkpen, A.C., Currall, S.C., 2004. The coevolution of trust, control, and learning in joint ventures. Organ. Sci. 15, 586–599. https://doi.org/10.1287/orsc.1040.0079. Inkpen, A.C., Dinur, A., 1998. Knowledge management processes and international joint ventures. Organ. Sci. 9, 454–468. https://doi.org/10.1287/orsc.9.4.454.

Inkpen, A.C., Tsang, E.W., 2007. Learning and strategic alliances. Acad. Manag. Ann. 1, 479–511. https://doi.org/10.1080/078559815.

Ireland, R.D., Hitt, M.A., Vaidyanath, D., 2002. Alliance management as a source of competitive advantage. J. Manag. 28, 413–446. https://doi.org/10.1016/S0149-2063(02)00134-4.

Jiang, R.J., Tao, Q.T., Santoro, M.D., 2010. Alliance portfolio diversity and firm performance. Strat. Manag. J. 31, 1136–1144. https://doi.org/10.1002/smj.869. Jiménez-Jiménez, D., Sanz-Valle, R., 2011. Innovation, organizational learning, and performance. J. Bus. Res. 64, 408–417. https://doi.org/10.1016/j.

jbusres.2010.09.010. Johnson, M.P., Schaltegger, S., 2020. Entrepreneurship for sustainable development: a review and multilevel causal mechanism framework. Enterpren. Theor. Pract. 44, 1141–1173. https://doi.org/10.1177/1042258719885368.

Kale, P., Singh, H., 2007. Building firm capabilities through learning: the role of the alliance learning process in alliance capability and firm-level alliance success. Strat. Manag. J. 28, 981–1000. https://doi.org/10.1002/smj.616.

Kavusan, K., Noorderhaven, N.G., Duysters, G.M., 2016. Knowledge acquisition and complementary specialization in alliances: the impact of technological overlap and alliance experience. Res. Pol. 45, 2153–2165. https://doi.org/10.1016/j.respol.2016.09.013.

Kennedy, S., Whiteman, G., van den Ende, J., 2017. Radical innovation for sustainability: the power of strategy and open innovation. Long. Range Plan. 50, 712–725. https://doi.org/10.1016/j.lrp.2016.05.004.

Keskin, D., Wever, R., Brezet, H., 2020. Product innovation processes in sustainability-oriented ventures: a study of effectuation and causation. J. Clean. Prod. 263, 121210 https://doi.org/10.1016/j.jclepro.2020.121210.

Khanna, T., Gulati, R., Nohria, N., 1998. The dynamics of learning alliances: competition, cooperation, and relative scope. Strat. Manag. J. 19, 193–210. https://doi. org/10.1002/(SICI)1097-0266(199803)19:3<193::AID-SMJ949>3.0.CO;2-C.

Klewitz, J., Hansen, E.G., 2014. Sustainability innovation of SMEs: a systematic review. J. Clean. Prod. 65, 57–75. https://doi.org/10.1016/j.jclepro.2013.07.017. Kohler, T., 2016. Corporate accelerators: building bridges between corporations and startups. Bus. Horiz. 59, 347–357. https://doi.org/10.1016/j. bushor.2016.01.008.

Kovács, G., Spens, K.M., 2005. Abductive reasoning in logistics research. Int. J. Phys. Distrib. Logist. Manag. 35, 132–144. https://doi.org/10.1108/ 09600030510590318

Kumar, R., Nti, K.O., 1998. Differential learning and interaction in alliance dynamics: a process and outcome discrepancy model. Organ. Sci. 9, 356–367. https://doi.org/10.1287/orsc.9.3.356.

Lane, P.J., Lubatkin, M., 1998. Relative absorptive capacity and interorganizational learning. Strat. Manag. J. 19, 461–477. https://doi.org/10.1002/(SICI)1097-0266 (199805)19:5<461::AID-SMJ953>3.0.CO;2-L.

Langley, A.N.N., Smallman, C., Tsoukas, H., Van de Ven, A.H., 2013. Process studies of change in organization and management: unveiling temporality, activity, and flow. Acad. Manag. J. 56, 1–13. https://doi.org/10.5465/amj.2013.4001.

Liu, C.L., Lui, S., 2020. Comparing the impact of alliance-learning activities on alliance performance across small and large firms. Knowl. Manag. Res. Pract. 18, 188–198. https://doi.org/10.1080/14778238.2019.1673674.

London, T., Rondinelli, D.A., O'Neill, H.U.G.H., 2004. Exploring uneasy learning alliances between corporations and non-profit organizations. In: Academy of Management Proceedings. Academy of Management, New York, pp. B1–B6, 2004, No. 1. Academy of Management.

Lubin, D.A., Esty, D.C., 2010. The sustainability imperative. Harv. Bus. Rev. 88, 42–50.

Magnusson, T., Lindström, G., Berggren, C., 2003. Architectural or modular innovation? Managing discontinuous product development in response to challenging environmental performance targets. Int. J. Innovat. Manag. 7, 1–26. https://doi.org/10.1142/S1363919603000714.

McCutchen Jr., W.W., Swamidass, P.M., 2004. Motivations for strategic alliances in the pharmaceutical/biotech industry: some new findings. J. High Technol. Manag. Res. 15, 197–214. https://doi.org/10.1016/j.hitech.2004.03.003.

McLellan, E., MacQueen, K.M., Neidig, J.L., 2003. Beyond the qualitative interview: data preparation and transcription. Field Methods 15, 63–84. https://doi.org/ 10.1177/1525822X02239573.

Miles, M.B., Huberman, A., Saldaña, J., 2014. Qualitative Data Analysis: A Methods Sourcebook. Sage, Thousand Oaks.

Minshall, T., Mortara, L., Valli, R., Probert, D., 2010. Making "asymmetric" partnerships work. Res. Technol. Manag. 53, 53–63. https://doi.org/10.1080/ 08956308.2010.11657631.

Mittra, J., 2007. Life science innovation and the restructuring of the pharmaceutical industry: merger, acquisition and strategic alliance behaviour of large firms. Technol. Anal. Strat. Manag. 19, 279–301. https://doi.org/10.1080/09537320701281540.

#### C. Hübel et al.

#### Long Range Planning xxx (xxxx) xxx

Mülling Neutzling, D., Land, A., Seuring, S., do Nascimento, L.F.M., 2018. Linking sustainability innovation to supply chain relationship integration. J. Clean. Prod. 172, 3448–3458. https://doi.org/10.1016/j.jclepro.2017.11.091.

Nielsen, B.B., 2005. The role of knowledge embeddedness in the creation of synergies in strategic alliances. J. Bus. Res. 58, 1194–1204. https://doi.org/10.1016/j. ibusres.2004.05.001.

Nielsen, B.B., Nielsen, S., 2009. Learning and innovation in international strategic alliances: an empirical test of the role of trust and tacitness. J. Manag. Stud. 46, 1031–1056. https://doi.org/10.1111/j.1467-6486.2009.00840.x.

O'Mahony, S., Bechky, B.A., 2008. Boundary organizations: enabling collaboration among unexpected allies. Adm. Sci. Q. 53, 422–459. https://doi.org/10.2189/ asqu.53.3.422.

Parrish, B.D., 2010. Sustainability-driven entrepreneurship: principles of organization design. J. Bus. Ventur. 25, 510–523. https://doi.org/10.1016/j. ibusvent 2009 05 005

Pérez, L., Florin, J., Whitelock, J., 2012. Dancing with elephants: the challenges of managing asymmetric technology alliances. J. High Technol. Manag. Res. 23, 142–154. https://doi.org/10.1016/j.hitech.2012.06.007.

Pettigrew, A.M., 1990. Longitudinal field research on change: theory and practice. Organ. Sci. 1, 267–292. https://doi.org/10.1287/orsc.1.3.267.

Prashantham, S., Birkinshaw, J., 2008. Dancing with gorillas: how small firms can partner effectively with MNCs. Calif. Manag. Rev. 51, 6–23. https://doi.org/ 10.2307/41166466.

Prior, L.F., 2003. Using Documents in Social Research. Sage, London.

Rennings, K., 2000. Redefining innovation. Eco-innovation research and the contribution from ecological economics. Ecol. Econ. 32, 319–332. https://doi.org/ 10.1016/S0921-8009(99)00112-3.

Ring, P.S., Van de Ven, A.H., 1992. Structuring cooperative relationships between organizations. Strat. Manag. J. 13, 483–498. https://doi.org/10.1002/ smj.4250130702.

Rockström, J., Steffen, W., Noone, K., Persson, Å., Chapin III, F.S., Lambin, E., Lenton, T.M., Scheffer, M., Folke, C., Schellnhuber, H.J., Nykvist, B., Wit, C. A. de, Hughes, T., van der Leeuw, S., Rodhe, H., Sörlin, S., Snyder, P.K., Costanza, R., Svedin, U., Falkenmark, M., Karlberg, L., Corell, R.W., Fabry, V.J., Hansen, J., Walker, B., Liverman, D., Richardson, K., Crutzen, P., Foley, J., 2009. Planetary boundaries: exploring the safe operating space for humanity. Ecol. Soc. 14 https:// doi.org/10.5751/ES-03180-140232.

Rohrbeck, R., Konnertz, L., Knab, S., 2013. Collaborative business modelling for systemic and sustainability innovations. Int. J. Technol. Manag. 63, 4–23. https://doi. org/10.1504/IJTM.2013.055577.

Rondinelli, D.A., London, T., 2003. How corporations and environmental groups cooperate: assessing cross-sector alliances and collaborations. Acad. Manag. Perspect. 17, 61–76. https://doi.org/10.5465/ame.2003.9474812.

Rothaermel, F.T., 2001. Complementary assets, strategic alliances, and the incumbent's advantage: an empirical study of industry and firm effects in the biopharmaceutical industry. Res. Pol. 30, 1235–1251. https://doi.org/10.1016/S0048-7333(00)00142-6.

Russo, M., Cesarani, M., 2017. Strategic alliance success factors: a literature review on alliance lifecycle. Int. J. Bus. Adm. 8, 1–9. https://doi.org/10.5430/ijba. v8n3p1.

Saldaña, J., 2015. The Coding Manual for Qualitative Researchers. Sage, Thousand Oaks.

Sapontzis, S.F., 2004. Food for Thought: the Debate over Eating Meat. Prometheus Books, New York.

Schaltegger, S., Hörisch, J., 2017. In search of the dominant rationale in sustainability management: legitimacy-or profit-seeking? J. Bus. Ethics 145, 259–276. https://doi.org/10.1007/s10551-015-2854-3.

Schaltegger, S., Wagner, M., 2011. Sustainable entrepreneurship and sustainability innovation: categories and interactions. Bus. Strat. Environ. 20, 222–237. https://doi.org/10.1002/bse.682.

Schaltegger, S., Lüdeke-Freund, F., Hansen, E.G., 2016. Business models for sustainability: a Co-evolutionary analysis of sustainable entrepreneurship, innovation and transformation. Organ. Environ. 29, 3–10. https://doi.org/10.1177/1086026616633272.

Schildt, H., Keil, T., Maula, M., 2012. The temporal effects of relative and firm-level absorptive capacity on interorganizational learning. Strat. Manag. J. 33, 1154–1173. https://doi.org/10.1002/smj.1963.

Schilke, O., Goerzen, A., 2010. Alliance management capability: an investigation of the construct and its measurement. J. Manag. 36, 1192–1219. https://doi.org/ 10.1177/0149206310362102.

Shapiro, P., 2018. How Growing Meat without Animals Will Revolutionize Dinner and the World. Gallery Books, New York.

Shenkar, O., Li, J., 1999. Knowledge search in international cooperative ventures. Organ. Sci. 10, 134–143. https://doi.org/10.1287/orsc.10.2.134.

Stern, A.J., Hicks, T., 2000. The Process of Business/environmental Collaborations: Partnering for Sustainability. Greenwood Publishing Group, Westport. Subramanian, A.M., Soh, P.H., 2017. Linking alliance portfolios to recombinant innovation: the combined effects of diversity and alliance experience. Long. Range Plan. 50, 636–652. https://doi.org/10.1016/j.lrp.2016.11.001.

Thomas, G., 2010. Doing case study: abduction not induction, phronesis not theory. Qual. Inq. 16, 575–582. https://doi.org/10.1177/1077800410372601. Turner III, D.W., 2010. Qualitative interview design: a practical guide for novice investigators. Qual. Rep. 15, 754. https://doi.org/10.12307/2392941. Van de Ven, A.H., Walker, G., 1984. The dynamics of interorganizational coordination. Adm. Sci. Q. 598–621. https://doi.org/10.2307/2392941. Wals, A.E., Corcoran, P.B. (Eds.), 2012. Learning for Sustainability in Times of Accelerating Change. Wageningen Academic Publishers, Wageningen. Weiblen, T., Chesbrough, H.W., 2015. Engaging with startups to enhance corporate innovation. Calif. Manag. Rev. 57, 66–90. https://doi.org/10.1525/ cmr.2015.57.2.66.

Weissbrod, I., 2019. Leadership and goal setting for sustainable innovation in large businesses. In: Bocken, N.M.P., Ritala, P., Albareda, L., Verburg, R. (Eds.), Innovation for Sustainability: Business Transformations towards a Better World. Palgrave Macmillan, London, pp. 135–155. https://doi.org/10.1007/978-3-319-97385-2.8

Wengraf, T., 2001. Qualitative Research Interviewing. Sage, London.

Yin, R.K., 2009. Case Study Research: Design and Methods, fourth ed. Sage, Thousand Oaks.



Charlott Hübel is the Robert Bosch Foundation scholar of Management and Entrepreneurship in the interdisciplinary research project "Processes of Sustainability Transformation" at Leuphana University Lüneburg, Germany. Her research focuses on entrepreneurial and managerial approaches and processes through which mainstream firms strategically reorient towards sustainability and contribute to industry transformations. Email charlott.huebel@leuphana.de

### Long Range Planning xxx (xxxx) xxx





Ilka Weissbrod, PhD, teaches and researches at the intersection of Innovation and Strategy in Sustainability Management. 10 years as innovation advisor to international companies large and small and her PhD work at Imperial College London are the foundation of her research themes: business experimentation for sustainability, sustainable business models, and urgency of implementing sustainability-oriented innovation. Ilka is closely involved with business practice through research projects, and in speaker and advisory roles. Email ilka.weissbrod@leuphana.de



Stefan Schaltegger is Professor of Corporate Sustainability Management as well as Founder and Head of the Centre for Sustainability Management (CSM) and the MBA Sustainability Management at Leuphana University Lüneburg, Germany. His research fields in the area of corporate sustainability management cover sustainable entrepreneurship, stakeholder theory, sustainability accounting and management. He has received various research awards and an honorary doctorate and is associate editor or member of the editorial board of international management and sustainability journals. Email stefan.schaltegger@leuphana.de

# Entrepreneurship-driven organizational transformation for sustainability: a sensemaking lens

Charlott Hübel

Centre for Sustainability Management, Leuphana Universität Lüneburg, Lueneburg, Germany

### Abstract

**Purpose** – This paper adopts a sensemaking lens to explore the process of entrepreneurship-driven organizational transformation for sustainability.

**Design/methodology/approach** – Analysis is based on an in-depth case study of a large European meat company. Sensemaking by top and middle managers is analyzed over the period of 18 months.

**Findings** – The findings show how, over time, bidirectional sensemaking, that is, sensemaking for and of sustainability-oriented entrepreneurship, directed and accelerated organizational transformation for sustainability. The case company transformed with regard to organizational strategy, structure, operations and identity. The process revealed temporally different involvement of top and middle managers in sensemaking.

**Originality/value** – This paper offers unique insights into fast and emergent sustainability-oriented change in an established organization within a highly topical context. The results highlight how continuous and increasingly shared sensemaking can help top and middle managers navigate organizational change for sustainability in dynamic environments over time.

Keywords Sustainability, Transformation, Corporate entrepreneurship, Organizational change, Sensemaking

Paper type Research paper

## Introduction

To address pressing sustainability challenges and simultaneously remain competitive, established organizations are required to transform toward sustainability in a timely manner (Millar *et al.*, 2012; De Matos and Clegg, 2013). Transformative organizational change can be facilitated and accelerated through sustainability-oriented entrepreneurship, that is, innovative activities that help organizations exploit entrepreneurial opportunities while meeting environmental and social objectives (Miles *et al.*, 2009; Provasnek *et al.*, 2016; Schaltegger and Wagner, 2011). Such innovative activities may include sustainable product, process or business model innovations (Klewitz and Hansen, 2014; Phan *et al.*, 2009) as well as the creation of joint ventures (Espinosa and Suanes, 2011) and spin-offs (Parhankangas and Arenius, 2003). Entrepreneurship-driven changes amount to organizational transformation for sustainability, when sustainability is integrated in essential domains of organizational activity, including strategy, structure, work practices, power distribution, culture and identity (Sroufe, 2017; Müller and Pfleger, 2014; Romanelli and Tushman, 1994).

Pursuing entrepreneurial activities (Hill and Levenhagen, 1995; Bettiol *et al.*, 2012; Dutta and Thornhill, 2014) and embedding sustainability into the organization (Van der Heijden *et al.*, 2012; Hahn *et al.*, 2014) involve high complexities and uncertainties. To manage these processes, there is a need for organizational decision-makers to *make sense* of them (Weick *et al.*, 2005; Lüscher and Lewis, 2008; Maitlis and Christianson, 2014). On the one hand, they need to recognize and conceptualize entrepreneurial opportunities for being able to pursue them (Cornelissen and Clark, 2010, p. 539; Mitchell *et al.*, 2011; Kaffka *et al.*, 2020); on the other hand, they need to interpret activities to be able to deduce appropriate organizational changes (Hill and



Journal of Organizational Change Management © Emerald Publishing Limited 0953-4814 DOI 10.1108/JOCM-03-2021-0067

Funding: This research was funded by the Robert Bosch Stiftung (Award number: 12.5.F082.0021.0).

Entrepreneurshipdriven transformation

Received 5 March 2021 Revised 1 October 2021 Accepted 16 December 2021 JOCM

Levenhagen, 1995; Bettiol *et al.*, 2012; Dutta and Thornhill, 2014). Due to the dynamism of the entrepreneurial context (Johnson and Bock, 2017; Hill and Levenhagen, 1995) and the status quo challenging nature of sustainability measures (Miles *et al.*, 2009; Hahn *et al.*, 2014), the need for sensemaking might reoccur along an organization's sustainability-oriented entrepreneurial journey. This can have crucial implications for organizational transformation over time, rendering a process perspective important (Langley *et al.*, 2013). This research therefore explores *how managerial sensemaking directs entrepreneurship-driven organizational transformation for sustainability over time*.

The adequacy of the research focus results from two main gaps within the literature. First, there is no research at the intersection of sensemaking, entrepreneurship-driven organizational change and sustainability. While previous studies have analyzed sensemaking for sustainability-oriented organizational change (Van der Heijden *et al.*, 2012) and sensemaking for entrepreneurship-driven organizational change (Corley and Gioia, 2004), to date no study brings these insights together. Second, there is no research that investigates how sensemaking in the context of entrepreneurship-driven organizational change unfolds over the course of time with potentially changing entrepreneurial activities. In this regard, few studies have distinguished different directions of sensemaking involved in entrepreneurial endeavors and organizational change (Brown *et al.*, 2015; Smith and Cao, 2007) but have not yet considered how these types interact over time. Similarly, research has pointed to different roles of top and middle managers in sensemaking for entrepreneurship-driven change (Gioia and Chittipeddi, 1991; Balogun and Johnson, 2004; Lüscher and Lewis, 2008) but lacks consideration of potentially divergent involvement throughout the organization's entrepreneurial journey.

The empirical work constitutes an 18-month case study of a large European meat company that has engaged in entrepreneurial activities in the alternative protein field. Analysis of top and middle manager sensemaking reveals that bidirectional sensemaking, that is, sensemaking *for* and sensemaking *of* sustainability-oriented entrepreneurship, led to increasingly radical entrepreneurial activities and increasingly transformative organizational changes. The findings contribute to the sensemaking literature by (1) specifying the temporal progression of bidirectional sensemaking in entrepreneurial established organizations and (2) outlining the diverging involvement of top and middle managers in sensemaking and their implications over time. The findings contribute to the organizational change literature by showing (3) how continuous bidirectional sensemaking can navigate entrepreneurship-driven transformation and (4) how an accumulation of meaning can result in comprehensible but moderate organizational transformation for sustainability.

The introduction of sensemaking as analytical lens is followed by an outline of the research context and the methods adopted for data collection and data analysis. Next, the findings are presented and thereupon discussed by highlighting key contributions. The concluding remarks include managerial implications, limitations and research recommendations.

## A sensemaking lens

Sensemaking has been established as a vital lens for studying organizational change (Weick, 1979; Van de Ven and Poole, 2005). This study analyzes sensemaking at the intersection of entrepreneurship, organizational transformation and sustainability, that is, sensemaking regarding sustainability-oriented entrepreneurship and its implications for organizational transformation. Here, analysis distinguishes between (1) sensemaking *for* sustainability-oriented entrepreneurship, which allows for the pursuit of entrepreneurial activities, and (2) sensemaking *of* sustainability-oriented entrepreneurship, which helps initiate respective organizational changes. Top managers and middle managers adopt crucial roles in sensemaking in both directions (Gioia and Chittipeddi, 1991; Balogun and Johnson, 2004; Lüscher and Lewis, 2008).

## Sensemaking for sustainability-oriented entrepreneurship

Research on entrepreneurial cognition proposes that entrepreneurship "starts with imagining the opportunity for novel ventures" in the sense that individual entrepreneurs construct meaning for themselves and others to justify the new path (Cornelissen and Clark, 2010, p. 539; Mitchell *et al.*, 2011; Kaffka *et al.*, 2020). It is argued that such sensemaking *for* entrepreneurship is particularly guided by entrepreneurs' cognitive frames, which help them to construct meaning of changes in the external and organizational environment (Bettiol *et al.*, 2012; Dutta and Thornhill, 2014). For the context of sustainability, Seidel *et al.* (2013) mention how sensemaking *for* sustainability-oriented entrepreneurship can be triggered by society's increasing environmental awareness and concomitant increased demand for environmentally friendly products.

Previous research has suggested that top managers adopt a crucial role in sensemaking *for* entrepreneurship, as their belief systems substantially guide sensemaking and entrepreneurial action (Smith and Cao, 2007). However, for top managers in established organizations, as opposed to independent entrepreneurs, sensemaking *for* entrepreneurship can come with challenges. As "embedded agents," they make sense of new meanings always in relation to existing ones (Garud and Karnøe, 2001; Garud *et al.*, 2007). Hence, becoming entrepreneurial often implies questioning the current interpretive scheme and developing and applying a new one (Hill and Levenhagen, 1995). As sustainability measures can be particularly status quo challenging for conventional organizations (Hahn *et al.*, 2014), it is suggested that a corporate sustainability strategy with clear goals (Arnold and Hockerts, 2011; Kennedy *et al.*, 2017) and sustainability principles that are fully embraced by managers (Miles *et al.*, 2009) can facilitate the pursuit of sustainability-oriented entrepreneurship.

### Sensemaking of sustainability-oriented entrepreneurship

The implementation of entrepreneurial activities often confronts organizational members with a new organizational status quo that needs to be interpreted and rationalized, that is, made sense of (Hill and Levenhagen, 1995; Bettiol *et al.*, 2012; Dutta and Thornhill, 2014). Such sensemaking *of* entrepreneurial activities involves asking the basic question of "What is going on?" (Lüscher and Lewis, 2008; Weick *et al.*, 2005; Gioia and Chittipeddi, 1991). The need for sensemaking is particularly high in the context of sustainability innovations, as they usually aim at replacement instead of enhancement (Schaltegger and Wagner, 2011; Magnusson *et al.*, 2003) and thereby call into question long-standing practices and cognitive frames (Hahn *et al.*, 2014). A study by Van der Heijden *et al.* (2012) identified sensemaking for organizational change toward sustainability as an emergent, gradual process, suggesting that such change develops slowly over time. Research has left open the question how this sensemaking-directed change would develop in the dynamic and disruptive context of sustainability-oriented entrepreneurship.

The literature distinguishes different roles for top and middle managers in sensemaking *of* entrepreneurial activities. In their study on identity change during a corporate spin-off, Corley and Gioia (2004) found that top managers' sensemaking helped them to move from initial ambiguity toward a renewed clarity about the new venture. The literature gives different indications of which organizational changes could support this "renewed clarity": While innovation management research suggests a structural separation between conventional and innovative units (Gilbert, 2006; De Visser *et al.*, 2010), research on hybrid organizations stresses the potential of "selectively coupling" different business logics (Pache and Santos, 2013). Besides making sense of arising complexity themselves, there is a need for top managers to promote the sensemaking of other organizational members about new developments. Here, research on sense-giving emphasizes the importance of metaphors and narratives as well as process facilitators, including organizational routines and practices (Maitlis and Lawrence, 2007; Gioia and Chittipeddi, 1991; Hill and Levenhagen, 1995).

Entrepreneurshipdriven transformation

Middle managers are important recipients of sense-giving, as they interpret and implement organizational change (Lüscher and Lewis, 2008; Balogun and Johnson, 2004). The sense-giving literature describes how sense-giving can impact change recipients, particularly pointing to negative or undesired outcomes. On the one hand, recipients could "normalize" the change by making it fit to existing logics, thereby avoiding sensemaking altogether (Ashforth and Kreiner, 2002). On the other hand, they could engage in negative sensemaking, meaning that they discover inconsistencies between their professional identity and the proposed change, and thereby experience temporal identity instability (Conroy and O'Leary-Kelly, 2014) and resist change (Nag et al., 2007). Previous organizational change research has outlined that managers' resistance of or support for sustainability-oriented organizational change is strongly dependent on their lived experiences, interpersonal interactions as well as personal beliefs and perceptions (Cherrier et al., 2012; Harris and Crane, 2002; Stoughton and Ludema, 2012: Stokes and Harris, 2012). Khan (2018) thus suggests that, to shape sensemaking positively in sustainability-related initiatives, top managers should focus on continuous informal communication practices with employees. In this regard, Maitlis (2005) argues that if both leaders and employees engage in sense-giving, they can jointly create a unitary account with consistent actions.

### Linking the two directions of sensemaking

Combining sensemaking *for* and *of* entrepreneurship is a matter of temporality. While sensemaking *for* is future-oriented, sensemaking *of* occurs after an event has passed. In this regard, Brown *et al.* (2015), in their conceptual paper on sensemaking in organizational studies, have differentiated prospective and retrospective sensemaking. The authors encourage research to further unpack prospective sensemaking and to "foreground the practices of strategy and processes of change" (p. 272). Smith and Cao (2007), who conceptualized an entrepreneurial perspective on the firm-environment relationship, give a first indication of linking the two directions of sensemaking, resulting in entrepreneurial action, and how entrepreneurial action, in turn, can lead to ambiguity, learning and potential changes in top managers' belief systems. However, the study does not explore respective progressions over time. In a more recent paper, Zhang (2016) found how managers' perceptions can act as mediator between external developments and organizational change, thereby suggesting a co-evolutionary perspective of organizational change over time.

To date, no study has combined the two directions of sensemaking, sensemaking *for* and sensemaking *of*, to analyze entrepreneurship-driven organizational transformation, not least in the context of sustainability. By combining the two introduced sensemaking lenses to analyze organizational change for sustainability in the entrepreneurial context and by distinguishing the involvement of top and middle manager sensemaking over time, this study can offer novel insights into the role of sensemaking in introducing and navigating sustainability-oriented change in established organizations.

### Methods

This paper studies the case of the entrepreneurship-driven transformation for sustainability of a large, family-run European meat product producer, "BIGCO" (Yin, 2017). Analyzing the "temporal progressions" of top and middle managers' sensemaking *for* and *of* entrepreneurial activities over a period of 18 months, this study can be placed in the field of process studies (Langley *et al.*, 2013, p. 1). However, as the investigation is also concerned with the extent of resulting organizational transformation, that is, the outcome of sensemaking, it includes certain elements of a consequential or outcome analysis (Van de Ven and Huber, 1990). The following sections introduce the case context and detailed data collection and analysis methods.

### Case context

Conventional meat producers have been facing challenges related to environmental impacts, animal welfare, working conditions and consumer health (Steinfeld *et al.*, 2006). Additionally, there has been a growing consumer trend toward vegetarianism, with up to one-third of the population in some Western European countries choosing to consume limited amounts of meat (Statista, 2020). At the same time, technological advances have made it possible to re-create the taste and texture of animal-based products using plant proteins, mycoproteins and animal cells (Van der Weele *et al.*, 2019). Since 2015, start-ups have entered the market offering highly innovative, environmentally and ethically sound meat substitutes (Shapiro, 2018). As these products have the potential to fundamentally disrupt the established meat industry, some meat producers like BIGCO have begun to seize entrepreneurial opportunities in the alternative protein field, thereby orienting toward sustainability-related value propositions. BIGCO has formed alliances with start-ups for cell-based, insect-based, plant-based and 3-D-printed protein solutions; has engaged in collaborative plant-based product innovation; and has co-founded an accelerator platform and a joint venture for the distribution of sustainable protein products. The entrepreneurial activities are part of the company's strategy to secure its competitiveness and long-term survival.

### Data collection

Data were collected for 18 months, from August 2018 until February 2020 (Figure 1), starting nine months after BIGCO's first entrepreneurial activities and two months after the activities had been formalized through a separate board and business unit. To reconstruct BIGCO's whole entrepreneurial journey, interviewees were asked to recall the pre-entrepreneurial and first entrepreneurial phase, and this recall was used as retrospective data (Pettigrew, 1990).

As shown in Table 1, data collection comprised semi-structured interviews (Wengraf, 2001) and ethnographic interviews (Munz, 2017) with top and middle managers, participant observation (DeWalt and DeWalt, 2002) and documents (Prior, 2003), with each data source fulfilling a different purpose during the analysis. Interviewed top managers include the CEO and chairman, the board lead for alternative proteins (former head of M&A) and the CEO of key BIGCO subsidiaries. Interviewed middle managers include managers from sales, marketing, project management, product development, communication, sustainability management and the director for alternative proteins.

### Data analysis

Data analysis was based on qualitative techniques (Miles and Huberman, 1984; Morse, 1994) and included three main steps. In step one, a coding process conducted with the qualitative data analysis software program MAXQDA generated first-order descriptive, process and emotion codes (Saldaña, 2015). This first analysis phase was an ongoing process that started



Figure 1. BIGCO's entrepreneurial activities and study observation period

Entrepreneurshipdriven transformation

#### ★ Alliance with alternative protein start-up

JOCM	Data types	Sources	Use in the analysis
	Interviews	12 semi-structured interviews with top and middle managers 20 ethnographic interviews with top and middle managers	<ul> <li>Reconstruct the entrepreneurial process</li> <li>Investigate sensemaking processes</li> <li>Investigate organizational transformation</li> <li>Supplement formal interviews</li> <li>Gain deeper insights into personal reflection (i.e. sensemaking)</li> </ul>
	Balanced participatory observation	11 events (press events, meetings, industry events)	<ul><li>Examine shared sensemaking</li><li>Investigate organizational transformation</li></ul>
Table 1.         Data types, sources         and use in the analysis	Documents	61 publicly available documents 8 internal documents	• Examine external communication of entrepreneurial activities and organizational transformation

after the first formal interviews and that was based on an iterative approach, which required going back and forth between the data, (i.e. interviews, observation and documents) and the literature (i.e. insights from sensemaking, corporate entrepreneurship and organizational transformation research). The resulting 59 codes were subsumed under four overarching code categories:

- (1) *Initial conditions*: It includes codes related to the company characteristics *before* the initiation of entrepreneurial activities (informed by Bettiol *et al.*, 2012; Lounsbury *et al.*, 2019; Miles *et al.*, 2009).
- (2) Sensemaking: It includes codes related to ambiguity and rationalization efforts as expressed by top and middle managers (informed by Weick, 1995; Lüscher and Lewis, 2008; Gioia and Chittipeddi, 1991).
- (3) *Entrepreneurial activities*: It includes codes related to product, process and business model innovation activities, collaborations with start-ups and new venture formation (informed by Phan *et al.*, 2009; Morris *et al.*, 2011; Miles *et al.*, 2009; Klewitz and Hansen, 2014).
- (4) Organizational transformation: It includes codes related to changes to strategy, structure, work practices, power distribution and identity (informed by Romanelli and Tushman, 1994; Orlikowski, 1996). The exclusive focus on top and middle managers and limited observation period did not allow for the identification of culture changes. However, the category includes changes to managers' personal perceptions, which give first indications of culture changes for sustainability (informed by Stoughton and Ludema, 2012; Stokes and Harris, 2012).

In step two, the timeline of BIGCO's entrepreneurial activities was reconstructed. In line with McMullen and Dimov's (2013) notion of the "entrepreneurial journey," which allows for subdiving activities into phases as part of a dynamic process, BIGCO's entrepreneurial journey could be divided into a pre-entrepreneurial phase and three ensuing, partly overlapping entrepreneurial phases (Table 2).

Codes and respective text sections for the four categories were ordered chronologically along the entrepreneurial phases, with a clear demarcation of top and middle manager sensemaking. The findings will be presented accordingly. In step three of the data analysis, a presentation and group meeting with four key BIGCO managers in late February 2020 served to validate the study's findings. This measure did not result in any additional codes or categories.

	Pre- entrepreneurial phase	1st entrepreneurial phase	2nd entrepreneurial phase	3rd entrepreneurial phase	Entrepreneurship- driven transformation
Time period Entr. activities	Until 12/2017	12/2017–09/2018 Forming alliances with start-ups for cell-based	07/2018–09/2019 Forming further alliances with start-	02/2019–02/2020 Co-founding a global accelerator platform for	
		meat and insect-based meat analogues; plant- based meat analogues, initiating/intensifying own product innovation	ups for plant-based fish, egg and cheese products	protein solutions and a sustainability-focused joint venture for the distribution of protein products across Europe	Table 2. BIGCO's entrepreneurial phases

# Findings

The findings show that along three phases of increasingly radical entrepreneurial activities in the alternative protein field, BIGCO went through transformative changes to strategy, structure, power distribution, work practices, identity and personal perceptions. As shown in Figure 2, transformation was directed by managerial sensemaking *for* and *of* sustainability-oriented entrepreneurship. The temporal occurrence and intensity of sensemaking by individual top and middle managers differed in the process. The following sections detail managers' sensemaking along the three entrepreneurial phases.

# Pre-entrepreneurial phase

Being considered among employees as the company's main *"visionary"* (ethnographic interview 10/2018) and someone who is *"incredibly open for new paths"* (Interview 9), BIGCO's



Figure 2. Sensemaking directing BIGCO's entrepreneurshipdriven organizational transformation for sustainability CEO was the first to engage in sensemaking *for* sustainability-oriented entrepreneurship in the alternative protein field:

Veganism is a topic which develops over time. Why should not we do it? We are in the convenience area [...] Whether I produce animal protein or plant protein, this is pretty much the same. I have the know-how to process the raw material into end products. Why should not we have a look at the vegan field, when the niche is big enough? (Interview 1)

The CEO particularly "ogled" with the topic of cultivated meat, seeing it as an important trend that has to be dealt with (Interview 1). According to the CEO, exploring the alternative protein was in accordance with the general sustainability and innovation orientation of the company (Interview 1). Thus, the CEO instructed the then head of M&A to search for partners in the emerging cultivated meat and plant-based industry. The head of M&A, however, was "surprised, why we should deal with [the cultivated meat] topic at all" and perceived the plant-based field as "an obscure corner in a market niche with hitherto low potential to dissolve the boundaries towards flexitarianism" (Interview 10). Following discussions with BIGCO's CEO and extensive research, however, the former head of M&A grew increasingly interested in the topic and started to wonder, too, whether "we have to adapt somehow to remain sustainable" (Interview 10).

### First entrepreneurial phase

In late 2017 and early 2018, the former head of M&A kick-started three strategic alliances between BIGCO and start-ups for cultivated meat, insect-based meat analog and plant-based meat analog. Middle managers from sales and marketing, despite becoming involved in the operational aspects of the partnerships, largely reacted to the new developments with indifference (Interviews 5; 8; 12):

Things you do not know, you naturally always consider with a big question mark [...] Well, perhaps you google it once, but then for a few weeks it's gone again. [The alliance activities] were known, but it did not really make us [sales managers] think that it would go anywhere. (Interview 5)

In contrast, BIGCO's top management engaged in extensive reflections and identified enormous potential of the alternative protein field. Particularly, the CEO and former head of M&A had realized that the start-ups' products were perfectly compatible with BIGCO's existing expertise, infrastructure and customer base and provided an opportunity for additional business growth (Interviews 1; 10). These realizations were specifically driven by decisive experiences in meetings between the top management team and the start-ups:

For me, trying the product [cultivated meat] for the first time was one of the greatest experiences. In this moment, we left the PowerPoint world, tasted the product and realized: This exists. This is no gibberish of scientists, it is reality [...] This was a pivotal moment for me (Interview 10).

Making sense *of* the first entrepreneurial activities made the top management team feel the need to formulate a company vision that would cover the new activities in the alternative protein field (ethnographic interview 5/2019; board discussion 2/2020): "*Actually, our aim is to become provider of high quality proteins. Not necessarily limited to meat*" (Interview 1). To account for activities in both areas, meat and alternative proteins, and to align identity with strategy, BIGCO's top management introduced the new strategic claim "growth through diversity" and created a separate business unit for alternative proteins (Interviews 1; 2; 11). Seeing the need to give sense of the new strategy and attaching more importance to it (Interview 11), a former sales employee was appointed alliance coordinator and the former head of M&A was appointed board lead for alternative proteins in July 2018 (Interviews 1;10). In terms of work practices, the CEO considered different options:

JOCM

In textbook fashion, you would need to create a separate company [...]But I do not know whether Entrepreneurshipthis is the right way, as we definitely have people who are able to think dichotomously. (Interview 1)

Thus, he decided to have relevant employees in sales, marketing and product development work on meat and alternative proteins topics simultaneously. The approach should emphasize the integral character of BIGCO's new identity and ensure authentic internal transformation (ethnographic interview 05/2019; Interview 10; board discussion 02/2020).

### Second entrepreneurial phase

BIGCO's new identity and strategy triggered further sensemaking for entrepreneurship in the top management team, particularly among the board lead for alternative proteins and CEO:

The initial thought was: Who are we? Meat. And what do we want to include? Meat substitutes.[...] But then we said: Well, if we define ourselves as protein provider, there are still certain areas, fish and eggs, which we have not covered yet. (Interview 10)

Accordingly, the board lead for alternative proteins initiated further alliances with start-ups for plant-based fish, egg and cheese products, resulting in a diversified alliance portfolio *"covering all areas"* relevant for flexitarians (Interview 10). At the same time, however, he weighed the risk of becoming too dependent on alliance partners and therefore considered investing more energy on applying the acquired product know-how internally (ethnographic interviews 08/2019; 09/2019). Consequently, in-house plant-based product innovation was increased. Here, for the CEO, it was crucial to focus on plant-based instead of vegetarian products, as otherwise "topics such as laying hen production and killing of male chicks catch up with you again" (Interview 1).

The new entrepreneurial activities triggered further sensemaking. Due to the increased impact on in-house activities and expressed doubts about the direction of change by some employees, the CEO felt the need for an increase in employee communication:

I continuously explain to employees - not all, but most of those who need to know - why I do what I do. [I explain that] the vegan topic or alternative protein products stabilize the company and that there is nothing to be afraid of (Interview 1)

To support the CEO's efforts to "widen the thinking of employees" (Interview 2) and achieve their acceptance, the top management team started to establish key figures in the company in August 2019. They appointed a new Director for Alternative Proteins – a former slaughterhouse manager who had joined the company two years earlier (Interviews 9; 10). Besides these power distribution changes, the second entrepreneurial phase also triggered crucial changes in personal perceptions. The data assert that increased interactions with the alternative protein community made BIGCO's middle managers become increasingly interested in alternative proteins (Interviews 4; 5; 7; 8; 10) and prompted few managers from project management and product development to start considering global sustainability issues (Interviews 3; 4). BIGCO's project manager and trend scouter, who had joined BIGCO three years before, recounts how he became interested in global food security:

I never really thought about, how I can provide food so that everyone on this planet has enough. These are perspective that I only adopted because of such people [start-up owners] and such companies [alternative protein start-ups]. (Interview 4)

Sustainability-related reflections among individual middle managers were further encouraged by the new director for alternative proteins, who had a passion for innovations and sustainability (Interview 9; ethnographic interview 10/2019). Particularly the board lead for alternative proteins, due to repeated and close interactions with the start-up

driven transformation partners, began to adopt new views on food security and resource efficiency and became increasingly enthusiastic about driving organizational change toward alternative proteins (ethnographic interviews 11/2018: 02/2019: 05/2019: Interview 10). However, some sales managers, who did not fully grasp the immediacy of the induced changes, still saw a shift in organizational identity toward protein provider, something that will only become relevant later: "Maybe in the future, as a vision for 2050" (Interview 8).

### Third entrepreneurial bhase

The newly gained enthusiasm for the alternative protein field prompted BIGCO's board lead for alternative proteins to engage once again in sensemaking for new entrepreneurial activities:

There is no vehicle worldwide, where you can invest, if you want to position yourself broadly in the alternative protein field. It does not exist [...] There is no platform yet, which combines industry know-how with new products. (Interview 10)

Since BIGCO's activities were too small to found a separate spin-off company (ethnographic interview 01/2019), the board lead drove the decision of BIGCO's top management team to opt for a "hybrid model" (Interview 10): co-founding a global accelerator platform for plant-based product innovations and a joint venture for the distribution of respective products across Europe.

The diversity of entrepreneurial activities triggered further sensemaking of these activities, which, for the first time, explicitly included middle managers (see Figure 3). To encourage shared sensemaking, the board lead for alternative proteins and the director of alternative proteins initiated regular team meetings with people from sales, production, marketing and product development (Interviews 9; 10). The questions posed by different managers and discussed during these exchanges concerned potential future partnerships (ethnographic interview 10/2019), the roles of the accelerator platform and the new venture (Interview 11) and implications of the new activities for branding (Interview 12), product development (Interview 4) and structural integration (Interview 9).

According to the board lead for alternative proteins, finding answers to these questions "is by far the greatest challenge that we face right now, because it is getting emotional." (Interview 10). Above all, the top management team's plans to turn BIGCO into a "food company with two



IOCM

Figure 3.

over time

divisions" (Interview 10) with an alternative protein brand on equal footing with BIGCO's Entrepreneurshiphitherto ubiquitous meat brand was met with resistance by some marketing and sales mangers (ethnographic interviews 10/2019: 11/2019a: 11/2019b: Interview 9). The marketing manager, for instance, rejected the idea altogether: "The [meat] brand already radiates trusts and I do not have to try building a new brand with a modest use of funds." (Interview 12). Some sales mangers shared this skepticism and thought of the transformation as a change from "[meat brand], the slaughter company to [meat brand], the protein company." Concerning the speed and direction of organizational transformation, the director for alternative proteins observed two opposing camps:

Many [long-term employees] react with fear, you can see it. And others, especially the young, enthusiastic and involved employees want to move things forward and find the topic really exciting. (Interview 9)

For some long-term employees, no longer regarding the meat brand as a synonym for BIGCO "requires an insane change in thinking" (Interview 9). The difficulty of getting long-term employees on board, including some managers from marketing and sales, made the top management team realize that cognitive integration would be more difficult to achieve than initially thought (ethnographic interview 11/2019; board discussion 02/2020). They thus considered creating new positions in the alternative protein unit to take pressure off current employees still working in both areas (board discussion 02/2020). As of February 2020, the data indicate that tensions have decreased and that BIGCO's overarching identity with separate sub-identities finds broader acceptance (board discussion 02/2020).

### Summary of findings

Along three entrepreneurial phases, managerial sensemaking directed BIGCO's transformation in essential domains of organizational activity. Two findings are of particular importance. First, continuous sensemaking for and of sustainability-oriented entrepreneurship adopted a bidirectional nature, resulting in increasingly radical entrepreneurial activities and increasingly impactful organizational change over time. While sensemaking was initially limited to the top management team, particularly to the CEO and the board lead for alternative proteins, it became a shared process among top and middle managers in late phases of BIGCO's entrepreneurial journey. The late explicit involvement of middle managers in sensemaking revealed resistance by some managers and led to temporal tensions regarding organizational change. Second, continuous sensemaking rendered transformation toward sustainability highly emergent over a short period. Notably, the fast pace of change did not imply radicality in terms of replacement and renewal. Rather, overarching claims aimed to achieve compatibility with business as usual and, thus, comprehensible evolution instead of disruption.

### **Discussion and contribution**

By adopting a sensemaking lens to examine the process of entrepreneurship-driven organizational transformation for sustainability, this process study contributes to the sensemaking and organizational change literature.

The findings contribute to the entrepreneurial and organizational sensemaking literature by shedding light on temporal progressions of managerial sensemaking in dynamic contexts. While previous research has begun to conceptualize different directions of sensemaking (Brown et al., 2015; Smith and Cao, 2007), this study identifies the dynamic interaction of prospective and retrospective sensemaking as a bidirectional, highly emergent process. It suggests that the continuous alternation between directions of sensemaking can help managers to navigate entrepreneurship-driven change for sustainability, with sensemaking by involved managers becoming increasingly concerned with sustainability issues over time. In addition, the analysis

driven transformation provides insights into managers' diverging involvement in the two directions of sensemaking. The findings generally confirm entrepreneurial sensemaking research by showing that sensemaking for entrepreneurship remained limited to "lead entrepreneurs" at top management level (Dutta and Thornhill, 2014; Smith and Cao, 2007). However, the board lead's sensemaking for entrepreneurship only occurred after the CEO had given sense to the board about new developments. This finding suggests that mutual sense-giving among the same hierarchical level can be fruitful, if not necessary, for initiating and advancing change. This study further contributes to the literature by detailing progressions of middle manager sensemaking of entrepreneurial activities and organizational change over time – from initial "normalization" (Ashforth and Kreiner, 2002) through "negative sensemaking" (Conroy and O'Leary-Kelly, 2014) to "renewed clarity" (Corley and Gioja, 2004). Here, the individual "sensemaking paths" of middle managers differed substantially: While relatively new and innovation-oriented employees welcomed change for sustainability and started to engage in respective reflections in the second entrepreneurial phase, long-term employees from sales and marketing engaged in sensemaking only once processes had been formalized and eventually voiced their resistance. This finding suggests that support for or resistance to sustainability-oriented change is not only dependent on experiences or personal beliefs (Cherrier et al., 2012; Harris and Crane, 2002; Stoughton and Ludema, 2012; Stokes and Harris, 2012), but also on managers' ability for implicit sensemaking, which, in turn, depends on the degree of cognitive attachment to established routines and professional identities. Here, an earlier integration of skeptical managers in explicit sensemaking might have allowed for a more seamless transition and a unitary account as mentioned by Maitlis (2005).

This study further contributes to research on organizational change for sustainability by identifying particularities of the entrepreneurial context. While Van der Heijden et al. (2012) found sensemaking for sustainability-oriented organizational change to be a long-term process, the case of BIGCO shows that alternating between sensemaking for and of sustainabilityoriented entrepreneurial activities can significantly speed up transformation. This finding is highly relevant considering the urgency of many of today's sustainability challenges and the mounting pressures for companies to transform (Millar *et al.*, 2012). However, in contrast to suggestions by innovation management research (Gilbert, 2006; De Visser et al., 2010), BIGCO managers decided against the structural and cognitive separation of the two business units and in contrast to conceptualizations by sustainability innovation research (Schaltegger and Wagner, 2011: Magnusson et al., 2003). BIGCO's entrepreneurial activities did not lead to the replacement of the established meat business. In fact, BIGCO largely became entrepreneurial and transformed *without* calling into question the current interpretive scheme or way of doing business (Hill and Levenhagen, 1995). Instead, BIGCO's new abstract core strategy and identity allowed for the coexistence of seemingly contradictory business foci. This accumulation of meaning was due to continued sensemaking of new meanings in relation to the established business (Garud and Karnøe, 2001; Garud *et al.*, 2007). This meaning-making was facilitated by the top management's understanding of BIGCO as an already sustainability-oriented company (Kennedy et al., 2017; Miles et al., 2009). Only with rising tensions in the last entrepreneurial phase did BIGCO's top management team decide to increase cognitive and operational separation of the two business units, aiming for a "selective (de-) coupling" of logics as suggested by Pache and Santos (2013). As this shows, continuous bidirectional and shared sensemaking of top and middle managers allowed for the successful management of organizational transformation. At the same time, however, the focus on sensemaking in relation to established meanings might have impeded more radical change for sustainability.

### Conclusion

IOCM

This study explored how managerial sensemaking directs processes of entrepreneurshipdriven organizational transformation for sustainability over time. The case results suggest that (1) bidirectional sensemaking for and of entrepreneurial activities helps managing and Entrepreneurshipaccelerating organizational transformation for sustainability, (2) a processual accumulation of meaning allows for overarching, integrative claims balancing the old and the new and (3) managers' diverging "sensemaking paths" can be aligned through explicit and shared sensemaking of top and middle managers concerning both structural and cognitive aspects.

driven transformation

### Limitations and future research

This study comes with certain limitations, indicating possible directions for future research. Limitations regard the generalizability of findings and the scope of analysis. Questions of generalizability arise, as this study was based on a unique case with an exceptional setting. Thus, further investigations of entrepreneurial organizations in industries confronted with different sustainability challenges are encouraged. Further research is needed to investigate long-term impacts on different organizational domains. For instance, longitudinal studies could provide a richer picture of organizational culture changes that go beyond individual-level perceptions. To gain even more comprehensive insights, future research could investigate multilevel interactions of sensemaking that include both intraorganizational dynamics and interactions between the organization and the institutional environment, thereby combining this study's analysis focus with the foci of Zhang (2016), Khan (2018) or He and Baruch (2009).

### Managerial implications

The study's results have important implications for management. First and foremost, they emphasize the necessity for an early formalization of shared sensemaking in sustainabilityoriented change that goes beyond individual sensemaking by few organizational change agents (as e.g. analyzed by Van der Heijden et al., 2012). By including relevant middle managers early on and considering their divergent experiences and beliefs, reflections about new sustainability requirements, disembedding from established routines and support for transformation can be facilitated. To encourage collective sensemaking, top managers could, for instance, initiate change workshops (Ala-Laurinaho et al., 2017), in the context of which participants can share future visions, capture rising tensions and jointly develop practices that correspond with the new organizational reality. Software-based concept mapping (Faily et al., 2012) and/or the introduction and use of strategic linguistic terms and phrases (Jalonen et al., 2018) could further facilitate collective sensemaking. The results also show that, due to continuous sensemaking in relation to established meanings, the outcome of transformation for sustainability was comprehensible but less radical. For instance, all of BIGCO's new positions were filled by internal employees. To advance more fundamental change for sustainability, an early cognitive, operational and structural separation of the two business units could be facilitated by the recruitment of external staff with specific expertise in sustainability innovations and their management.

### References

- Ala-Laurinaho, A., Kurki, A.L. and Abildgaard, J.S. (2017), "Supporting sensemaking to promote a systemic view of organizational change-contributions from activity theory", Journal of Change Management, Vol. 17 No. 4, pp. 367-387, doi: 10.1080/14697017.2017.1309566.
- Arnold, M.G. and Hockerts, K. (2011), "The greening Dutchman: Philips' process of green flagging to drive sustainable innovations", Business Strategy and the Environment, Vol. 20 No. 6, pp. 394-407, doi: 10.1002/bse.700.
- Ashforth, B.E. and Kreiner, G.E. (2002), "Normalizing emotion in organizations: making the extraordinary seem ordinary", Human Resource Management Review, Vol. 12 No. 2, pp. 215-235, doi: 10.1016/S1053-4822(02)00047-5.

- Balogun, J. and Johnson, G. (2004), "Organizational restructuring and middle manager sensemaking", Academy of Management Journal, Vol. 47 No. 4, pp. 523-549, doi: 10.2307/20159600.
- Bettiol, M., Di Maria, E. and Finotto, V. (2012), "Marketing in SMEs: the role of entrepreneurial sensemaking", *International Entrepreneurship and Management Journal*, Vol. 8 No. 2, pp. 223-248, doi: 10.1007/s11365-011-0174-3.
- Brown, A.D., Colville, I. and Pye, A. (2015), "Making sense of sensemaking in organization studies", Organization Studies, Vol. 36 No. 2, pp. 265-277, doi: 10.1177/0170840614559259.
- Cherrier, H., Russell, S.V. and Fielding, K. (2012), "Corporate environmentalism and top management identity negotiation", *Journal of Organizational Change Management*, Vol. 25 No. 4, pp. 518-534, doi: 10.1108/09534811211239209.
- Conroy, S.A. and O'Leary-Kelly, A.M. (2014), "Letting go and moving on: work-related identity loss and recovery", Academy of Management Review, Vol. 39, pp. 67-87, doi: 10.5465/amr.2011.0396.
- Corley, K.G. and Gioia, D.A. (2004), "Identity ambiguity and change in the wake of a corporate spinoff", Administrative Science Quarterly, Vol. 49 No. 2, pp. 173-208, doi: 10.2307/4131471.
- Cornelissen, J.P. and Clarke, J.S. (2010), "Imagining and rationalizing opportunities: inductive reasoning and the creation and justification of new ventures", *Academy of Management Review*, Vol. 35 No. 4, pp. 539-557.
- De Matos, J.A. and Clegg, S.R. (2013), "Sustainability and organizational change", Journal of Change Management, Vol. 13 No. 4, pp. 382-386, doi: 10.1080/14697017.2013.851912.
- De Visser, M., de Weerd-Nederhof, P., Faems, D., Song, M., Van Looy, B. and Visscher, K. (2010), "Structural ambidexterity in NPD processes: a firm-level assessment of the impact of differentiated structures on innovation performance", *Technovation*, Vol. 30 Nos 5-6, pp. 291-299, doi: 10.1016/j.technovation.2009.09.008.
- DeWalt, K.M. and DeWalt, B.R. (2002), Participant Observation: A Guide for Fieldworkers, AltaMira Press, Walnut Creek.
- Dutta, D.K. and Thornhill, S. (2014), "How do entrepreneurial growth intentions evolve? A sensemakingsensegiving perspective", *Journal of Small Business Strategy*, Vol. 24 No. 1, pp. 61-76.
- Espinosa, M.D.M.B. and Suanes, A.M. (2011), "Corporate entrepreneurship through joint venture", International Entrepreneurship and Management Journal, Vol. 7 No. 3, pp. 413-430, doi: 10.1007/ s11365-011-0203-2.
- Faily, S., Lyle, J., Paul, A., Atzeni, A., Blomme, D., Desruelle, H. and Bangalore, K. (2012), "Requirements sensemaking using concept maps", in Winckler, M., Forbrig, P. and Bernhaupt, R. (Eds), *Human-Centered Software Engineering, HCSE 2012, Lecture Notes in Computer Science*, Springer, Berlin and Heidelberg, Vol. 7623, doi: 10.1007/978-3-642-34347-6\_13.
- Garud, R. and Karnøe, P. (2001), "Path creation as a process of mindful deviation", in Garud, R. and Karnøe, P. (Eds), *Path Dependence and Creation*, Lawrence Erlbaum Associates, Mahwah, pp. 1-40.
- Garud, R., Hardy, C. and Maguire, S. (2007), "Institutional entrepreneurship as embedded agency: an introduction to the special issue", *Organization Studies*, Vol. 28 No. 7, pp. 958-969, doi: 10.1177/0170840607078958.
- Gilbert, C.G. (2006), "Change in the presence of residual fit: can competing frames coexist?", *Organization Science*, Vol. 17 No. 1, pp. 150-167, doi: 10.1287/orsc.1050.0160.
- Gioia, D.A. and Chittipeddi, K. (1991), "Sensemaking and sensegiving in strategic change initiation", Strategic Management Journal, Vol. 12 No. 6, pp. 433-448, doi: 10.1002/smj.4250120604.
- Hahn, T., Preuss, L., Pinkse, J. and Figge, F. (2014), "Cognitive frames in corporate sustainability: managerial sensemaking with paradoxical and business case frames", Academy of Management Review, Vol. 39 No. 4, pp. 463-487, doi: 10.5465/amr.2012.0341.
- Harris, L.C. and Crane, A. (2002), "The greening of organizational culture: management views on the depth, degree and diffusion of change", *Journal of Organizational Change Management*, Vol. 15 No. 3, pp. 214-234, doi: 10.1108/09534810210429273.

- He, H. and Baruch, Y. (2009), "Transforming organizational identity under institutional change", Entrepreneurship-Journal of Organizational Change Management, Vol. 22 No. 6, pp. 575-599, doi: 10.1108/ 09534810910997014.
- Hill, R.C. and Levenhagen, M. (1995), "Metaphors and mental models: sensemaking and sensegiving in innovative and entrepreneurial activities", Journal of Management, Vol. 21 No. 6, pp. 1057-1074, doi: 10.1016/0149-2063(95)90022-5.
- Jalonen, K., Schildt, H. and Vaara, E. (2018), "Strategic concepts as micro-level tools in strategic sensemaking", Strategic Management Journal, Vol. 39 No. 10, pp. 2794-2826, doi: 10.1002/smj.2924.
- Johnson, D. and Bock, A.J. (2017), "Coping with uncertainty: entrepreneurial sensemaking in regenerative medicine venturing", The Journal of Technology Transfer, Vol. 42 No. 1, pp. 33-58, doi: 10.1007/s10961-015-9465-0.
- Kaffka, G.A., Singaram, R. and Kraaijenbrink, J. (2020), "New venture creation and the development of entrepreneurial cognition", Academy of Management Proceedings, Vol. 2020 No. 1, p. 20704, doi: 10.5465/AMBPP.2020.37.
- Kennedy, S., Whiteman, G. and van den Ende, J. (2017), "Radical innovation for sustainability: the power of strategy and open innovation", Long Range Planning, Vol. 50 No. 6, pp. 712-725, doi: 10.1016/j.lrp.2016.05.004.
- Khan, A.S. (2018), "Institutions and sensemaking of change: institutional frame switching as sensemaking of microfinance in a Pakistani commercial bank", Journal of Organizational Change Management, Vol. 31 No. 3, pp. 532-555, doi: 10.1108/JOCM-01-2017-0001.
- Klewitz, J. and Hansen, E.G. (2014), "Sustainability-oriented innovation of SMEs: a systematic review", Journal of Cleaner Production, Vol. 65, pp. 57-75, doi: 10.1016/j.jclepro.2013.07.017.
- Langley, A.N.N., Smallman, C., Tsoukas, H. and Van de Ven, A.H. (2013), "Process studies of change in organization and management: unveiling temporality, activity, and flow", Academy of Management Journal, Vol. 56 No. 1, pp. 1-13.
- Lounsbury, M., Cornelissen, J., Granqvist, N. and Grodal, S. (2019), "Culture, innovation and entrepreneurship", Innovation, Vol. 21 No. 1, pp. 1-12, doi: 10.1080/14479338.2018.1537716.
- Lüscher, L.S. and Lewis, M.W. (2008), "Organizational change and managerial sensemaking: working through paradox", Academy of Management Journal, Vol. 51 No. 2, pp. 221-240, doi: 10.5465/ amj.2008.31767217.
- Magnusson, T., Lindström, G. and Berggren, C. (2003), "Architectural or modular innovation? Managing discontinuous product development in response to challenging environmental performance targets", International Journal of Innovation Management, Vol. 7 No. 1, pp. 1-26, doi: 10.1142/S1363919603000714.
- Maitlis, S. (2005), "The social processes of organizational sensemaking", Academy of Management *Journal*, Vol. 48 No. 1, pp. 21-49, doi: 10.5465/ami.2005.15993111.
- Maitlis, S. and Christianson, M. (2014), "Sensemaking in organizations: taking stock and moving forward", Academy of Management Annals, Vol. 8 No. 1, pp. 57-125, doi: 10.5465/19416520.2014. 873177.
- Maitlis, S. and Lawrence, T.B. (2007), "Triggers and enablers of sensegiving in organizations", Academy of Management Journal, Vol. 50 No. 1, pp. 57-84, doi: 10.5465/AMJ.2007.24160971.
- McMullen, J.S. and Dimov, D. (2013), "Time and the entrepreneurial journey: the problems and promise of studying entrepreneurship as a process", Journal of Management Studies, Vol. 50 No. 8, pp. 1481-1512. doi: 10.1111/joms.12049.
- Miles, M.B. and Huberman, A.M. (1984), Qualitative Data Analysis: A Sourcebook of New Methods, Sage, London.
- Miles, M.P., Munilla, L.S. and Darroch, J. (2009), "Sustainable corporate entrepreneurship", International Entrepreneurship and Management Journal, Vol. 5 No. 1, pp. 65-76, doi: 10.1007/ s11365-008-0074-3.

driven transformation

- Millar, C., Hind, P. and Magala, S. (2012), "Sustainability and the need for change: organisational change and transformational vision", *Journal of Organizational Change Management*, Vol. 25 No. 4, pp. 489-500, doi: 10.1108/09544780710828421.
- Mitchell, R.K., Randolph-Seng, B. and Mitchell, J.R. (2011), "Socially situated cognition: imagining new opportunities for entrepreneurship research", *Academy of Management Review*, Vol. 36 No. 4, pp. 774-776, doi: 10.5465/amr.2011.0001.
- Morris, M.H., Kuratko, D.F. and Covin, J.G. (2011), Corporate Entrepreneurship and Innovation, 3rd ed., Cengage Learning, Mason, OH.
- Morse, J.M. (1994), "Designing funded qualitative research", in Denzin, N.K. and Lincoln, Y.S. (Eds), Handbook of Qualitative Research, Sage, London, pp. 220-235.
- Müller, A.L. and Pfleger, R. (2014), "Business transformation towards sustainability", Business Research, Vol. 7 No. 2, pp. 313-350, doi: 10.1007/s40685-014-0011-y.
- Munz, E.A. (2017), "Ethnographic interview", in Allen, M. (Ed.), The Sage Encyclopedia of Communication Research Methods, Sage, Thousand Oaks.
- Nag, R., Corley, K.G. and Gioia, D.A. (2007), "The intersection of organizational identity, knowledge, and practice: attempting strategic change via knowledge grafting", *Academy of Management Journal*, Vol. 50, pp. 821-847, doi: 10.5465/amj.2007.26279173.
- Orlikowski, W.J. (1996), "Improvising organizational transformation over time: a situated change perspective", *Information Systems Research*, Vol. 7 No. 1, pp. 63-92, doi: 10.1287/isre.7.1.63.
- Pache, A.C. and Santos, F. (2013), "Inside the hybrid organization: selective coupling as a response to competing institutional logics", *Academy of Management Journal*, Vol. 56 No. 4, pp. 972-1001, doi: 10.5465/amj.2011.0405.
- Parhankangas, A. and Arenius, P. (2003), "From a corporate venture to an independent company: a base for a taxonomy for corporate spin-off firms", *Research Policy*, Vol. 32 No. 3, pp. 463-481, doi: 10.1016/S0048-7333(02)00018-5.
- Pettigrew, A.M. (1990), "Longitudinal field research on change: theory and practice", Organization Science, Vol. 1, pp. 267-292, doi: 10.1287/orsc.1.3.267.
- Phan, P.H., Wright, M., Ucbasaran, D. and Tan, W.L. (2009), "Corporate entrepreneurship: current research and future directions", *Journal of Business Venturing*, Vol. 24 No. 3, pp. 197-205, doi: 10.1016/j.jbusvent.2009.01.007.
- Prior, L.F. (2003), Using Documents in Social Research, Sage, London.
- Provasnek, A.K., Schmid, E., Geissler, B. and Steiner, G. (2016), "Sustainable corporate entrepreneurship: performance and strategies toward innovation", *Business Strategy and the Environment*, Vol. 26 No. 4, pp. 521-535, doi: 10.1002/bse.1934.
- Romanelli, E. and Tushman, M.L. (1994), "Organizational transformation as punctuated equilibrium: an empirical test", Academy of Management Journal, Vol. 37 No. 5, pp. 1141-1166, doi: 10.5465/ 256669.
- Saldaña, J. (2015), The Coding Manual for Qualitative Researchers, Sage, London.
- Schaltegger, S. and Wagner, M. (2011), "Sustainable entrepreneurship and sustainability innovation: categories and interactions", *Business Strategy and the Environment*, Vol. 20 No. 4, pp. 222-237, doi: 10.1002/bse.682.
- Seidel, S., Recker, J. and Vom Brocke, J. (2013), "Sensemaking and sustainable practicing: functional affordances of information systems in green transformations", *MIS Quarterly*, Vol. 37 No. 4, pp. 1275-1299, available at: http://www.jstor.org/stable/43825792.
- Shapiro, P. (2018), How Growing Meat without Animals Will Revolutionize Dinner and the World, Gallery Books, New York.
- Smith, K.G. and Cao, Q. (2007), "An entrepreneurial perspective on the firm-environment relationship", *Strategic Entrepreneurship Journal*, Vol. 1 Nos 3-4, pp. 329-344, doi: 10.1002/ sej.25.

JOCM

- Sroufe, R. (2017), "Integration and organizational change towards sustainability", *Journal of Cleaner Entrepreneurship Production*, Vol. 162, pp. 315-329, doi: 10.1016/j.jclepro.2017.05.180.
- Statista (2020), "Share of population who followed a flexitarian diet in Great Britain in 2019, by region", available at: https://www.statista.com/statistics/1062370/likelihood-of-following-flexitarian-diet-in-great-britain-by-gender-and-age/ (accessed 20 September 2020).
- Steinfeld, H., Gerber, P., Wassenaar, T.D., Castel, V., Rosales, M., Rosales, M. and de Haan, C. (2006), Livestock's Long Shadow: Environmental Issues and Options, Food and Agriculture Organization, Rome.
- Stokes, P. and Harris, P. (2012), "Micro-moments, choice and responsibility in sustainable organizational change and transformation: the Janus dialectic", *Journal of Organizational Change Management*, Vol. 25 No. 4, pp. 595-611, doi: 10.1108/09534811211239245.
- Stoughton, A.M. and Ludema, J. (2012), "The driving forces of sustainability", Journal of Organizational Change Management, Vol. 25 No. 4, pp. 501-517, doi: 10.1108/09534811211239191.
- Van de Ven, A.H. and Huber, G.P. (1990), "Longitudinal field research methods for studying processes of organizational change", *Organization Science*, Vol. 1 No. 3, pp. 213-219, doi: 10.1287/orsc.1. 3.213.
- Van de Ven, A.H. and Poole, M.S. (2005), "Alternative approaches for studying organizational change", Organization Studies, Vol. 26 No. 9, pp. 1377-1404, doi: 10.1177/0170840605056907.
- Van der Heijden, A., Cramer, J.M. and Driessen, P.P. (2012), "Change agent sensemaking for sustainability in a multinational subsidiary", *Journal of Organizational Change Management*, Vol. 25 No. 4, pp. 535-559, doi: 10.1108/09534811211239218.
- Van der Weele, C., Feindt, P., van der Goot, A.J., van Mierlo, B. and van Boekel, M. (2019), "Meat alternatives: an integrative comparison", *Trends in Food Science and Technology*, Vol. 88, pp. 505-512, doi: 10.1016/j.tifs.2019.04.018.
- Weick, K.E. (1979), The Social Psychology of Organizing, 2nd ed., McGraw-Hill, New York.
- Weick, K.E. (1995), Sensemaking in Organizations, Sage, Thousand Oaks, Vol. 3.
- Weick, K.E., Sutcliffe, K.M. and Obstfeld, D. (2005), "Organizing and the process of sensemaking", Organization Science, Vol. 16 No. 4, pp. 409-421, doi: 10.1287/orsc.1050.0133.
- Wengraf, T. (2001), Qualitative Research Interviewing, Sage, London.
- Yin, R.K. (2017), Case Study Research and Applications: Design and Methods, Sage, London.
- Zhang, X. (2016), "Co-evolution between institutional environments and organizational change", Journal of Organizational Change Management, Vol. 29 No. 3, pp. 381-403, doi: 10.1108/JOCM-12-2014-0218.

### **Corresponding author**

Charlott Hübel can be contacted at: charlott.huebel@leuphana.de

driven transformation

# Annex

- I. Author contributions to dissertation papers and publication status
- II. Presentations related to this dissertation
- III. Declaration
## I. Author contributions to dissertation papers and publication status

No.	Short title	Author contributions	Author status	Weighting factor	Publication status
1	Industry Actor Barriers	CH: Conceptualization, literature review, investigation, formal analysis, writing original draft StS: Supervision, conceptualization, paper development, review &	Co-author with pre- dominant contribution	1.0	Published in Sustainable Production and Consumption Single-blind peer review IF: 8.921
_		editing			VHB JQ3: n.r. (SJR: 1.361)
2	Innovation Manage- ment	<ul> <li>SDS: Conceptualization, literature review, formal analysis, writing original draft</li> <li>CH: Conceptualization, literature review, substantial paper development</li> </ul>	Co-author with equal contribution	1.0	To be submitted Targeted journal: <i>Research Policy</i>
		<b>StS:</b> Supervision, conceptualization, review & editing			
3	Alliance Learning	CH: Investigation, literature review, formal analysis, conceptualization, writing original draft IW: Conceptualization, formal analysis, review & editing StS: Supervision, conceptualization, review & editing	Co-author with pre- dominant contribution	1.0	Published in <i>Long</i> <i>Range Planning</i> Double-blind peer review IF: 7.825 VHB JQ3: B (SJR: 3.239)
4	Managerial Sense- making		Single author	1.0	Published in the Journal of Organizational Change Management Double-blind peer review IF: 2.07 VHB JQ3: n.r. (SJR: 0.505)
			SUM:	4.0	( ), (

(§16 of cumulative dissertation guideline)

*CH* = *Charlott Hübel; IW* = *Dr. Ilka Weissbrod: SDS*= *Jun.-Prof. Dr. Samanthi Dijkstra-Silva; StS* = *Prof. Dr. Dr. Stefan Schaltegger* 

Date	Conference / Colloquium	Type of contribution	Title of contribution
Dec. 14 2017	PoST Research Colloquium	PhD Outline Presentation	The Role of Entrepreneurship in Transforming the Meat Industry
March 12 2018	CSM Research Retreat	Paper Presentation	The Transformation of the Meat Industry – Barriers and Scopes of Action
June 12-14 2018	9 <sup>th</sup> International Sustainability Transition Conference	Speed Talk (Co-presenter)	Understanding "Processes of Sustainability Transformation" – an inter- and transdisciplinary project
Dec. 14 2018	CSM Research Retreat	Paper Presentation	Barriers to a Sustainability Transformation of the Meat Industry: An Analysis of Lock- ins from an Industry Actor Perspective
Feb. 6-8 2019	Leverage Points 2019 Conference	Speed Talk + Poster Presentation	The Role of Sustainability Pioneers in Transforming the Meat Industry
July 14 2019	CSM Research Retreat	Paper Presentation	Experience meets innovation - How alliances between a meat producing incumbent and alternative protein pioneers might advance the sustainability transformation of the meat industry
Oct. 24-25 2019	Gem. Herbsttagung der Wissenschaftlichen Kommissionen Nachhaltigkeitsmanage ment und Produktionswirtschaft	Paper Presentation	Experience meets innovation: How alliances between a meat producing incumbent and alternative protein pioneers can advance a sustainability transformation of the meat industry
Oct. 21-23 2019	CSM + IQD Resarch Colloquium "Sustainability Management"	Paper Presentation	Experience meets innovation: a case study on alliance learning between a mainstream company and sustainability startups
July 2-4 2020	36th EGOS Colloquium 2020	Paper Presentation	The impact of entrepreneurial activities on established organizations over time: An identity change perspective
July 06 2020	PoST-Colloquium	Paper Presentation	The role of entrepreneurial incumbents in transforming (the meat) industry: Developing first ideas for the framework paper
Oct. 09 2020	CSM Research Retreat	Paper Presentation	Entrepreneurship-driven organizational transformation for sustainability - a sensemaking perspective
Feb. 16-17 2021	PoST Final Event	PhD Overview Presentation (Co-presenter)	Companies in and for transformation: Industry Actors' Agency

## II. Presentations related to this dissertation

## **III. Declaration**

(§16 of cumulative dissertation guideline)

I avouch that all information given in this appendix is true in each instance and overall.

Charlott Hübel